Extremal Problems on Directed Hypergraphs and the Erdös-Gyárfás Ramsey Problem Variant for Graphs

BY

ALEXANDER CAMERON B.A., Hendrix College, Conway, AR, 2006 J.D., Duke University, Durham, NC, 2011 M.S., University of Illinois at Chicago, Chicago, IL, 2017

THESIS

Submitted as partial fulfillment of the requirements for the degree of Doctor of Philosophy in Mathematics in the Graduate College of the University of Illinois at Chicago, 2018

Chicago, Illinois

Defense Committee: György Turán, Chair and Advisor Dhruv Mubayi, Advisor Shmuel Friedland Lev Reyzin Robert Sloan, Computer Science

ACKNOWLEDGEMENTS

First, I want to thank Marisa Borreca for her love, support, and endless patience with me when my research was not going well. It's not at all clear to me how I could have accomplished any of this without her.

I want to thank Gyuri Turán for providing me with a new direction when I needed it the most and for giving up so much of his time to listen to my ideas (even the bad ones). His support made all of the difference.

I want to thank Dhruv Mubayi for sharing both his expertise and his contagious enthusiasm for working on difficult questions. His ability to cut directly to the core issue of a problem is something I will try to emulate for the rest of my career.

Thank you also to the rest of my defense committee, Lev Reyzin, Shmuel Friedland, and Robert Sloan, for their time and insight.

I want to thank my mom for always being supportive of my life choices. She paid for me to take college courses in math when I was thirteen, and she supported my decision to get this PhD after law school instead of being an attorney. I'm sure this wasn't an easy position to get behind, but she's the one who nurtured my love of math early on so, you know, it was kinda her fault anyway.

I want to thank my dad for teaching me to love new ideas. He is a Renaissance man, and I have followed his example of life-long learning (perhaps too much). Specifically, when I was twelve he decided to work through a few college textbooks to learn math. I borrowed his books when he was finished, and here I am today. Thank you to Duff Campbell, Byungchul Cha, and Robert Dunn for being great teachers and for their advice and recommendations when applying for grad school. Thank you to Dave Schonert, Charles Mullins, Ze'ev Barel, Corey Boby, Robert Eslinger, George Connor, and everyone else who taught me math along the way. And thank you to David Sutherland who advised me to pick a young subject like combinatorics to do my research in. He was right.

Thank you to MSCS department at UIC for accepting me into their program and for supporting me financially for so many years.

Thank you to everyone who took the time over the last several years to listen to me talk about problems or shared their ideas with me: Emily Heath, Sam Cole, Florian Pfender, Mike Ferrara (and the GRWC in general), Bernard Lidický, Abhishek Methuku, Sean English, Lizard Reiland, Ervin Győri, Dániel Gerbner, Balázs Keszegh, Miklós Simonovits, András Gyárfás, Péter Hajnal, Sayan Mukherjee, Caroline Terry, and many others.

Thank you to all of my fellow math grad students at UIC. I had fun playing cards and working next to you in various coffee shops around Chicago.

Finally, I must thank Brittany Myers for convincing me to remain in graduate school when I was about to walk away. I would not be getting this degree without her advice. It's as simple as that.

Alex

CONTRIBUTION OF AUTHORS

Chapters [] § as well as Chapter [1] reproduce, with some minor changes, a paper published by the Electronic Journal of Combinatorics [12] for which I was the sole author. I retain the copyright to the publication. An early version of this paper was originally published on arXiv in two parts [8, 9]. Chapter 9 contains, with some minor changes, a paper published to arXiv [10] for which I was the sole author. Chapter [11] provides definitions and background on the (p,q)-coloring problem. Chapters [12] and [13] reproduce, with some minor changes, joint work with Emily Heath that has been reviewed and accepted for publication in Combinatorics, Probability & Computing [14]. It has yet to appear, but an early version has been published to arXiv [13]. Chapter [14] reproduces, with some minor changes, part of a paper published to arXiv [11] for which I am the sole author. Chapter [15] is a brief conclusion to the chapters about (p,q)-coloring. Appendix [A] is an algorithm used in Chapters [13] and [14] and is contained in both of the relevant papers.

TABLE OF CONTENTS

ACKNOWLEDGEMENTS.	ii
CONTRIBUTION OF AUTHORS	iv
LIST OF FIGURES.	viii
SUMMARY.	xi
Chapter 1.Introduction to Extremal Problems and Directed Hypergraphs.1.The forbidden subgraph problem.2.Additional extremal concepts.3.Directed graphs and hypergraphs.4. $2 \rightarrow 1$ directed hypergraphs.	$ \begin{array}{c} 1 \\ 2 \\ 5 \\ 6 \\ 7 \end{array} $
Chapter 2. The 4-resolvent Graph R_4	12
Chapter 3.The 3-resolvent Graph R_3 1.The oriented version2.The standard version	17 18 21
Chapter 4.The Escher Graph E 1.The oriented version2.Two lower bound constructions for $ex(n, E)$ 3.Add and remove edges4.No copy of E is created and the number of edges can only increase5.Case 1: $ T' \ge 5$ 6.Case 2: $ T' \le 4$ and some $x \in T'$ gives an T_x with 2 independent edges7.Case 3: $ T' = 0, 1, 2$ 8.Case 4: $ T' = 3$ 9.Case 5: $ T' = 4$	26 26 27 29 31 32 32 33 34 35
Chapter 5. The Graph I_0 1. Gates2. Bounding $\sum_{x \in V(C)} T_x $ for any connected component C of the gate3. Upper bound on $ex_o(n, I_0)$ 4. Lower bound constructions	38 40 42 46 49
Chapter 6. The Graph I_1	54
Chapter 7. The Graph H1 1. Counting edges by possible tail pairs 2. No pair points to more than one vertex 3. Some pair of vertices is the tail set to multiple edges of H	59 61 62 68

TABLE OF CONTENTS (continued)

4. The oriented extremal number	72
5. Intersections of multiedge triples in the standard version	75
6. An H_1 -free graph with $n-2$ multiedge triples	79
7. Fewer than $n-2$ multiedge triples	81
Chapter 8. The Graph H_2	84
Chapter 9. Generalized Directed Hypergraphs	87
1. Basic definitions and results	87
1.1. Blowups and blowup density	91
1.2. Supersaturation and related results	94
<u>2. Jumps</u>	97
2.1. Jumps pass up the lattice	100
2.2. Jumps do not pass down the lattice	101
3. Continuity and approximation	104
Chapter 10. Additional Questions about Directed Hypergraphs	107
1. Extremal numbers for tournaments	107
2. GDHs with $r \to 1$ edges	108
3. Differences between oriented and standard extremal numbers	111
4. General structural results	112
5. GDH questions	113
Chapter 11 Introduction to the (n, q) -coloring problem	115
$\underbrace{(p,q)}_{(p,q)} \underbrace{(p,q)}_{(p,q)} \underbrace{(p,q)}_{(p,q)}$	110
Chapter 12. The Modified CFLS coloring.	119
1. Avoided configurations	120
2. Modified CFLS.	122
Chapter 13. A (5,5)-coloring construction.	125
1 The Modified Inner Product coloring	140
1. The Modified limer roduct coloring	$120 \\ 127$
Inervision 1.1. Motivation	120 127 128
1. The Moduled finite Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 .	120 127 128 129
1.The Moduled line Froduct coloring1.1.Motivation1.2.The coloring χ_1 1.3.Algebraic definitions and facts	127 127 128 129 130
1. The Modified finite Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts 1.4. Properties of χ_1 .	127 128 129 130 132
1. The Modified line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings.	120 127 128 129 130 132 135
1. The Modified finite Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations.	127 127 128 129 130 132 135 136
1. The Moduled line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations. 3. Splitting the coloring.	127 128 129 130 132 135 136 137
1. The Moduled line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations. 3. Splitting the coloring. 3.1. The third configuration.	127 127 128 129 130 132 135 136 137 139
1. The Moduled line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations. 3. Splitting the coloring. 3.1. The third configuration.	127 127 128 129 130 132 135 136 137 139 146
1.The Moduled line Froduct coloring.1.1.Motivation.1.2.The coloring χ_1 .1.3.Algebraic definitions and facts.1.4.Properties of χ_1 .2.Combining the colorings.2.1.The first two configurations.3.Splitting the coloring.3.1.The third configuration.1.The algebraic construction.	127 128 129 130 132 135 136 137 139 146 147
1. The Moduled line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations. 3. Splitting the coloring. 3.1. The third configuration. 1. The algebraic construction. 1. The algebraic constructions. 2. Combining the coloring construction.	127 128 129 130 132 135 136 137 139 146 147 148
1. The Moduled line Froduct coloring. 1.1. Motivation. 1.2. The coloring χ_1 . 1.3. Algebraic definitions and facts. 1.4. Properties of χ_1 . 2. Combining the colorings. 2.1. The first two configurations. 3. Splitting the coloring. 3.1. The third configuration 1. The algebraic construction. 1. The algebraic savided by CFLS.	123 127 128 129 130 132 135 136 137 139 146 147 148 149
1.The Moduled Inlet Froduct coloring.1.1.Motivation.1.2.The coloring χ_1 1.3.Algebraic definitions and facts.1.4.Properties of χ_1 2.Combining the colorings.2.1.The first two configurations.3.Splitting the coloring3.1.The third configuration1.The algebraic construction.1.The algebraic construction2.Combining the construction3.Configurations avoided by CFLS.3.1.General "color cycle" configurations.	123 127 128 129 130 132 135 136 137 139 146 147 148 149 149
1.Inter Froduct coloring1.1.Motivation1.2.The coloring χ_1 1.3.Algebraic definitions and facts1.4.Properties of χ_1 2.Combining the colorings2.1.The first two configurations3.Splitting the coloring3.1.The third configuration1.The algebraic construction1.The algebraic construction2.Combining the constructions3.Configurations avoided by CFLS3.1.General "color cycle" configurations3.2.Configurations containing a monochromatic P_3	127 128 129 130 132 135 136 137 139 146 147 148 149 149 150
1.Inter Froduct coloring.1.1.Motivation.1.2.The coloring χ_1 .1.3.Algebraic definitions and facts.1.4.Properties of χ_1 .2.Combining the colorings.2.1.The first two configurations.3.Splitting the coloring.3.1.The third configuration.Chapter 14.A (5, 6)-coloring construction.1.The algebraic construction.2.Combining the construction.3.Configurations avoided by CFLS.3.1.General "color cycle" configurations.3.2.Configurations containing a monochromatic P_3 .3.3.Configurations containing an alternating C_4 .	$ \begin{array}{r} 123 \\ 127 \\ 128 \\ 129 \\ 130 \\ 132 \\ 135 \\ 135 \\ 136 \\ 137 \\ 139 \\ 146 \\ 147 \\ 148 \\ 149 \\ 149 \\ 150 \\ 151 \\ \end{array} $

4. Configurations avoided by the algebraic coloring	. 154
Chapter 15. Additional questions about (p,q) -colorings	. 158
1. General (p, p) -coloring with $n^{1/(p-2)+o(1)}$ colors	. 158
2. A better bound for $f(n, 5, 7)$. 158
3. The hypergraph version	.158
Appendix A. Algorithm for reducing cases	. 160
Appendix. CITED LITERATURE	. 162
Appendix. VITA	. 165
Appendix. Copyrights to Published Papers	. 167

LIST OF FIGURES

1	A triangle-free graph with many edges.	3
2	The 4-resolvent graph R_4 .	12
3	The lower bound construction for a graph with no R_4	12
4	Link graph of a 4-resolvent.	13
5	Link graph for an R_4 -free graph argument (oriented case).	14
6	Link graph for an R_4 -free graph argument (standard case).	15
7	The 3-resolvent graph R_3	17
8	The unique R_3 -free extremal construction.	17
9	Forbidden link graph intersections for R_3 -free graphs	18
10	Link graph structure for R_3 -free graphs (oriented version).	19
11	Example link graph structure for R_3 -free graphs (standard version).	23
12	The Escher graph E .	26
13	An order on the vertices of a <i>E</i> -free graph.	27
14	An <i>E</i> -free extremal construction.	28
15	Another <i>E</i> -free extremal construction.	28
16	The graph I_0	38
17	Link graph for I_0	39
18	The structure of a connected component of the gate G	41
19	Gate structure for an extremal I_0 -free graph when $n \equiv 0 \mod 3$	46
20	Gate structures of an extremal I_0 -free graph when $n \equiv 1 \mod 3$	47

21 Gate structures of an extremal I_0 -free graph when $n \equiv 2 \mod 3$	48
22 C_3 plus an edge.	50
23 C_4 with 2 additional edges in opposite tail link graphs	51
24 C_3 plus a 2-path.	53
25 The graph I_1	54
26 T_i in the oriented extremal construction for even vertices.	56
27 T_i in the oriented extremal construction for odd vertices	56
28 T_v in the oriented extremal construction for odd vertices	57
29 The graph H_1	59
30 Directed link graph of H_1	60
31 Structure of an H_1 -free graph.	62
32 H_1 -free special case 1.	63
33 H_1 -free special case 2.	66
34 H_1 -free special case 3.	70
35 Intersections of multiedges that form an H_1	78
36 The unique extremal H_1 -free construction (standard case).	80
37 The graph H_2	84
38 Inductive construction of H_2 -free oriented graphs.	85
<u>39 The lattice of GDHs with three vertices.</u>	89
40 $\pi(F) = \frac{4}{27}$	101
41 Four configurations avoided by the CFLS coloring	121
42 A striped K_4	123
43 Three configurations not avoided by the modified CFLS coloring 1	127

$\Lambda\Lambda$	The geometric visualization can be misleading since we are working over a
44	The geometric visualization can be inisiading since we are working over a
	finite field, but monochromatic neighborhoods are contained in affine planes. 128
45	The intersection of two monochromatic neighborhoods usually lies on an
10	
	affine line
16	The intersection of two monochrometic neighborhoods
40	The intersection of two monochromatic neighborhoods
47	Four configurations that each contain a forbiddon "color cycle"
41	Tour configurations that each contain a forbidden color cycle.
40	
48	CFLS forbids these 16 configurations, each contain a monochromatic P_3 151
49	Configurations containing an alternating C_4
50	Five additional configurations avoided by $\varphi_1, \ldots, 153$
	<u></u>
51	Configurations aliminated by the modified algebraic coloring
DT.	Configurations eminimated by the modified algebraic coloring

SUMMARY

This thesis has two parts. The first part, about Turán-type problems for directed hypergraphs, consists of Chapters 1 through 10. The second part, about a variation of the Ramsey problem, consists of Chapters 11 through 15. All of the material is related under the umbrella of extremal combinatorics.

The first part of this thesis primarily examines the extremal number of edges for directed hypergraphs given certain forbidden subgraphs. These sorts of questions are a big area of research for graphs and hypergraphs. Here the question is applied to the $2 \rightarrow 1$ directed hypergraph. Chapter 1 defines all of the relevant concepts. Chapters 2 through 8 give the extremal numbers for every $2 \rightarrow 1$ directed hypergraph with exactly two edges. Chapter 9 generalizes the concept of a directed hypergraph to include many different relational structures and extends some classical extremal results to this larger class of models. Chapter 10 concludes the first part of the thesis with a few stray results and open questions.

The second part of this thesis is about the (p, q)-coloring problem. A (p, q)-coloring is an edge-coloring of the complete graph K_n for which any p vertices must span at least q distinct colors. The goal is to find the minimum number of colors necessary for which such a coloring exists. Chapter 11 defines the necessary concepts and provides background. Chapter 12 provides a construction that will be used in subsequent chapters. Chapter 13 details a (5, 5)-coloring. Chapter 14 details a (5, 6)-coloring. Finally, Chapter 15 briefly explores some additional ideas for continued research.

CHAPTER 1

Introduction to Extremal Problems and Directed Hypergraphs

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

We will begin this chapter with the basic definitions of uniform graphs and hypergraphs. For any set V and any positive integer r we let $\binom{V}{r}$ denote the set of r-subsets of V in what follows. Additionally, given a positive integer n, we let [n] denote the set $\{1, 2, ..., n\}$.

DEFINITION 1.1. A graph is an ordered pair of sets G = (V, E) where V, the set of vertices, is finite, and E, the set of edges, is some subset of $\binom{V}{2}$.

Technically, this defines a simple graph in that it has no loops (an edge from a vertex to itself) or multiedges (two or more edges between the same two vertices). This definition generalizes to larger r.

DEFINITION 1.2. For some integer $r \ge 2$, an r-uniform hypergraph is an ordered pair of sets H = (V, E) where V, the set of vertices, is finite, and E, the set of edges (or hyperedges), is some subset of $\binom{V}{r}$. Often, we let V(H) and E(H) denote the vertex and edge sets of H when these sets have not been given explicitly.

Graphs and hypergraphs are widely-studied combinatorial objects, and many questions have been asked (and sometimes answered) about them. Here, we are concerned primarily with extremal questions.

1. The forbidden subgraph problem

Turán-type extremal problems for uniform graphs and hypergraphs make up a large and well-known area of research in combinatorics. Of these problems, the forbidden subgraph problem is the most basic: "Given a family of forbidden r-uniform hypergraphs \mathcal{F} , what is the maximum number of edges an r-uniform hypergraph on n vertices can have without containing any member of \mathcal{F} as a (not necessarily induced) subgraph?" Such problems were named after Paul Turán due to his important early results and conjectures concerning forbidden complete r-graphs [42, 43, 44]. We formalize this idea in the definitions that follow.

DEFINITION 1.3. Given two r-uniform hypergraphs, H and G, we call a function $\phi: V(H) \rightarrow V(G)$ a homomorphism if it preserves the edges of H:

$$v_1v_2\cdots v_r \in E(H) \implies \phi(v_1)\phi(v_2)\cdots\phi(v_r) \in E(G).$$

We will write $\phi: H \to G$ to indicate that ϕ is a homomorphism.

DEFINITION 1.4. Given a family \mathcal{F} of r-uniform hypergraphs, we say that a hypergraph H is \mathcal{F} -free if no injective homomorphism $\phi : F \to H$ exists for any $F \in \mathcal{F}$. If $\mathcal{F} = \{F\}$ we will simply write that G is F-free.

DEFINITION 1.5. Given a family \mathcal{F} of r-uniform hypergraphs, let the nth extremal number $ex(n, \mathcal{F})$ denote the maximum number of edges that any \mathcal{F} -free hypergraph on n vertices can have.

The extremal numbers of families of forbidden hypergraphs indicate the threshold number of edges at which any hypergraph, no matter how unstructured, is forced to have some local substructure.

For example, if a graph on n vertices has more than $n^2/4$ edges, then it must contain a 3-clique, three vertices that are all pairwise adjacent, no matter its structure. The complete bipartite graph with nearly equal parts (see Figure 1) demonstrates that we



FIGURE 1. A triangle-free graph with many edges.

can have at least this many edges without a 3-clique. The fact that this is the best that we can do is called Mantel's Theorem, and we say that $n^2/4$ is the *n*th extremal number for the 3-clique K_3 , $ex(n, K_3) = n^2/4$. Turán's Theorem generalized this result for cliques of any size [42], [43], [44]. It states that the maximum number of edges that a graph can have before it is forced to contain a clique of k vertices is the same as the number of edges found in the complete balanced (k - 1)-partite graph. These kinds of questions are difficult to answer in general for hypergraphs and other combinatorial structures. Even for 3-uniform hypergraphs, the extremal number of a 4-clique is unknown.

Often, it is easier to discuss these notions in terms of edge density rather than number of edges.

DEFINITION 1.6. Let H be an r-uniform hypergraph with n vertices and e(H) edges. Then the edge density of H is

$$d_H = \frac{e(H)}{\binom{n}{r}}.$$

DEFINITION 1.7. Given a forbidden family of r-uniform hypergraphs \mathcal{F} , the limit of the maximum edge densities of \mathcal{F} -free hypergraphs as the number of vertices goes to infinity is known as the **Turán density** of the family,

$$\pi(\mathcal{F}) = \lim_{n \to \infty} \frac{ex(n, \mathcal{F})}{\binom{n}{r}}.$$

A simple averaging argument demonstrates that such a limit always exists. For graphs, it is well-known that the Turán density of any forbidden family is determined by the minimum chromatic number of the graphs in the family. This is the famous Erdős-Stone Theorem [23]. Loosely speaking, the chromatic number of a graph is the minimum number of parts that any partition of the vertices can have such that no edge is contained inside a part.

DEFINITION 1.8. Let G be a graph and k be a positive integer. A **proper vertex** coloring of G with k colors is an assignment $f : V(G) \rightarrow [k]$ such that $f(x) \neq f(y)$ for any $xy \in E(G)$. The minimum k for which a proper vertex coloring of G exists is known as the chromatic number of G, $\chi(G)$. When $\chi(G) = 2$, we say that G is bipartite.

For any forbidden family of graphs \mathcal{F} , Erdős and Stone 23 showed that $\pi(\mathcal{F}) = \frac{k-2}{k-1}$ where

$$k = \min_{F \in \mathcal{F}} \{\chi(F)\}.$$

In particular, a forbidden graph F has Turán density zero if and only if it is bipartite. This idea extends to hypergraphs as well.

DEFINITION 1.9. Let H be an r-uniform hypergraph. We call H degenerate if $\pi(H) = 0.$

DEFINITION 1.10. Let H be an r-uniform hypergraph. We say that H is r-partite if there exists a partition of the vertices of H into r parts such that every edge of Hcontains exactly one vertex from each part.

Erdős **[IS]** showed that an *r*-uniform hypergraph is degenerate if and only if it is *r*-partite. Moreover, $\pi(H) \geq \frac{r!}{r^r}$ for any *r*-uniform hypergraph *H* that is not *r*-partite. This can be seen by taking the sequence of *r*-partite hypergraphs with nearly equal parts and every possible edge. The hypergraphs in this sequence are all *H*-free and their edge densities tend towards $\frac{r!}{r^r}$ as the number of vertices increase.

5

However, there is no known generalization of the result relating chromatic number and Turán density to hypergraphs. Even for the complete 3-graph on 4 vertices, $K_4^{(3)}$, the Turán density is unknown.

2. Additional extremal concepts

The label "Turán-type problem" is applied to more than just the forbidden subgraph problem. A closely related result for hypergraphs known as **supersaturation** says any large hypergraph with an edge density slightly more than the Turán density of some forbidden hypergraph F will not only contain a copy of F, but many copies. That is, let F be a an r-uniform hypergraph and let $\epsilon > 0$. For sufficiently large $n \ge n_0(F, \epsilon)$, any r-uniform hypergraph H on n elements with density $d_H \ge \pi(F) + \epsilon$ will contain at least $c\binom{n}{r}$ copies of F for some constant $c = c(F, \epsilon)$. In fact, this supersaturation result is used to prove the characterization of degenerate hypergraphs mentioned above.

Another, closely related, extremal question for hypergraphs known as the "jumping constant conjecture" was proposed by Erdős [22, 23].

DEFINITION 1.11. A real number $\alpha \in [0,1)$ is called a **jump** for an integer $r \ge 2$ if there exists some positive constant c which depends only on α such that for any $\epsilon > 0$ and positive integer l there exists a positive integer N for which any r-uniform hypergraph on $n \ge N$ vertices which has edge density at least $\alpha + \epsilon$ contains a subgraph on l vertices with edge density at least $\alpha + \epsilon$.

Informally, a jump is an edge density α for which any very large hypergraph with a slightly larger edge density must contain an arbitrarily large subgraph with edge density at least $\alpha + c(\alpha)$. That is, the density "jumps" by some fixed length $c(\alpha)$ when the overall edge density increases beyond α . The overall edge structure must get "clumpy." It is well-known that when r = 2, every $\alpha \in [0, 1)$ is a jump [22, 23]. Moreover, every $\alpha \in [0, \frac{r!}{r^r})$ is a jump for $r \ge 3$ [19]. The jumping constant conjecture asserted that every $\alpha \in [0, 1)$ is a jump for any r.

In 1984, Frankl and Rödl disproved the conjecture when they found the first instance of a nonjump for each $r \ge 3$ [27]. Since then many infinite sequences of nonjumps have been found, but the smallest known nonjump to date is $\frac{5r!}{2r^r}$ for each $r \ge 3$ determined by Frankl, Peng, Rödl, and Talbot [26]. The only additional jumps that have been found are all $\alpha \in [0.2299, 0.2316), [0.2871, \frac{8}{27})$ for r = 3 found by Baber and Talbot [4], using Razborov's flag algebra method [40].

3. Directed graphs and hypergraphs

Extremal problems like these have also been considered for directed graphs and multigraphs (with bounded multiplicity) [5], [6] and for the more general directed multihypergraphs [7].

Brown and Harary 6 determined the extremal numbers for several types of specific directed graphs including all tournaments - that is, a digraph with one edge in some orientation between every pair of vertices. Brown, Erdős, and Simonovits 5 determined the general structure of extremal sequences for every forbidden family of digraphs analogous to the Turán graphs for simple graphs.

The model of directed hypergraphs studied in $\boxed{7}$ have *r*-uniform edges such that the vertices of each edge are given a linear ordering. However, there are many other ways that one could conceivably define a uniform directed hypergraph. The graph theoretic properties of a more general definition of a nonuniform directed hypergraph were studied by Gallo, Longo, Pallottino, and Nguyen $\boxed{28}$. They defined a directed hyperedge as some subset of vertices with a partition into head vertices and tail vertices.

Langlois, Mubayi, Sloan, and Gy. Turán [31], 32 studied extremal properties of certain small configurations in a directed hypergraph model. This model can be thought of as a 2 \rightarrow 1 directed hypergraph where each edge has three vertices, two of which are "tails" and the third is a "head." They determined the extremal number for one such subgraph with two edges, and found the extremal number of a second configuration with two edges up to asymptotic equivalence. We will discuss their results in more detail in the following section. In Chapters 2 through 8, we determine the exact extremal numbers for every 2 \rightarrow 1 directed hypergraph with exactly two edges.

The totally directed hypergraph model considered in [7] and the $r \rightarrow 1$ directed hypergraph model resulting from the study of Horn clauses both lead to the natural question of all possible ways to define a directed hypergraph. The definition in this paper of the class of general directed hypergraph models attempts to unify all of the possible "natural" ways one could define a directed hypergraph so that certain extremal questions can be answered about all of them at once. Adding to the motivation of considering more general structures is the recent interest in Razborov's flag algebra method which applies to all relational theories and not just undirected hypergraphs. The fact that the *d*-simplex model studied by Leader as well as many other somewhat geometric models come out of the class defined in Chapter 9 was a very interesting accident.

4. $2 \rightarrow 1$ directed hypergraphs

The combinatorial structure treated in Chapters 2-8 is the $2 \rightarrow 1$ directed hypergraph defined as follows.

DEFINITION 1.12. $A \ 2 \rightarrow 1$ directed hypergraph is a pair H = (V, E) where V is a finite set of vertices and the set of edges E is some subset of the set of all pointed 3-subsets of V. That is, each edge is three distinct elements of V with one marked as special. This special vertex can be thought of as the head vertex of the edge while the other two make up the tail set of the edge. If H is such that every 3-subset of V contains at most one edge of E, then we call H oriented. For a given H we will typically write its vertex and edge sets as V(H) and E(H). We will write an edge as $ab \rightarrow c$ when the underlying 3-set is $\{a, b, c\}$ and the head vertex is c.

For simplicity we will usually refer to $2 \rightarrow 1$ directed hypergraphs as graphs or sometimes as $(2 \rightarrow 1)$ -graphs when needed to avoid confusion. This structure comes up as a particular instance of the model used to represent definite Horn formulas in the study of propositional logic and knowledge representation [1, 41]. Some combinatorial properties of this model were recently studied by Langlois, Mubayi, Sloan, and Turán [32, 31].

Before we can discuss their results we will need the following definitions which extend the concepts defined earlier in the chapter for graphs and hypergraphs to $2 \rightarrow 1$ directed hypergraphs.

DEFINITION 1.13. Given two graphs H and G, we call a function $\phi : V(H) \rightarrow V(G)$ a homomorphism if it preserves the edges of H:

$$ab \to c \in E(H) \implies \phi(a)\phi(b) \to \phi(c) \in E(G).$$

We will write $\phi: H \to G$ to indicate that ϕ is a homomorphism.

DEFINITION 1.14. Given a family \mathcal{F} of graphs, we say that a graph G is \mathcal{F} -free if no injective homomorphism $\phi: F \to G$ exists for any $F \in \mathcal{F}$. If $\mathcal{F} = \{F\}$ we will write that G is F-free.

DEFINITION 1.15. Given a family \mathcal{F} of graphs, let the nth extremal number ex (n, \mathcal{F}) denote the maximum number of edges that any \mathcal{F} -free graph on n vertices can have. Similarly, let the nth oriented extremal number $ex_o(n, \mathcal{F})$ be the maximum number of edges that any \mathcal{F} -free oriented graph on n vertices can have. Sometimes we will call the extremal number the standard extremal number or refer to the problem of determining the extremal number as the standard version of the problem to distinguish these concepts from their oriented counterparts. As before, if $\mathcal{F} = \{F\}$, then we will write ex(n, F) or $ex_o(n, F)$ for simplicity.

In [32, 31], the authors studied the extremal numbers for two small $(2 \rightarrow 1)$ -graphs. They refer to these two graphs as the 4-resolvent and the 3-resolvent configurations after their relevance in propositional logic. Here, we will denote these graphs as R_4 and R_3 respectively and define them formally as

$$V(R_4) = \{a, b, c, d, e\}$$
 and $E(R_4) = \{ab \to c, cd \to e\}$

and

$$V(R_3) = \{a, b, c, d\}$$
 and $E(R_3) = \{ab \to c, bc \to d\}.$

In [31], the authors determined $ex(n, R_4)$ exactly for sufficiently large n, and in [32] they determined the sequence $ex(n, R_3)$ up to asymptotic equivalence. In these papers, the authors discuss a third graph with two edges which they call an Escher configuration because it calls to mind the famous M.C. Escher piece in which two hands draw each other. This graph is on four vertices $\{a, b, c, d\}$ and has edge set $\{ab \rightarrow c, cd \rightarrow b\}$. In this paper, we will denote this graph by E. These three graphs turn out to be the only three graphs with exactly two edges and more than three vertices for which the extremal numbers are cubic in n. They are also the only three with two edges on more than three vertices that do not satisfy the following definition.

DEFINITION 1.16. A graph H is degenerate if its vertices can be partitioned into three sets, $V(H) = T_1 \cup T_2 \cup K$ such that every edge of E(H) is of the form $t_1t_2 \rightarrow k$ for some $t_1 \in T_1$, $t_2 \in T_2$, and $k \in K$.

An immediate consequence of Theorem 9.4 shown in Chapter 9 is that the extremal numbers for a graph H are cubic in n if and only if H is not degenerate.

In this specific model of directed hypergraphs, there are nine different graphs with exactly two edges. Of these, four are not degenerate. One of these is the graph on three vertices with exactly two edges, $V = \{a, b, c\}$ and $E = \{ab \rightarrow c, ac \rightarrow b\}$. It is trivial to see that both the standard and oriented extremal numbers for this graph are $\binom{n}{3}$. The other three nondegenerate graphs are R_4 , R_3 , and E. We will determine both the standard and oriented extremal numbers for each of these graphs in Sections 2, 3, and 4 respectively.

Of the five degenerate graphs with exactly two edges, one has extremal numbers that are trivial to find. This is the graph with two independent edges, $V = \{a, b, c, d, e, f\}$ and $E = \{ab \rightarrow c, de \rightarrow f\}$. The extremal number for this graph comes directly from the known extremal number of the undirected 3-graph that consists of two independent edges - that is, the maximum number of edges in a 3-graph with edge intersection sizes never equal to zero. That extremal number is $\binom{n-1}{2}$ for sufficiently large n. Therefore, the oriented extremal number for two independent $2 \rightarrow 1$ edges is also $\binom{n-1}{2}$ and the standard extremal number is $3\binom{n-1}{2}$.

We will call the other four degenerate graphs with two edges I_0 , I_1 , H_1 , and H_2 and define them as follows:

- $V(I_0) = \{a, b, c, d, x\}$ and $E(I_0) = \{ab \to x, cd \to x\}$
- $V(I_1) = \{a, b, c, d\}$ and $E(I_1) = \{ab \to c, ad \to c\}$
- $V(H_1) = \{a, b, c, d, x\}$ and $E(H_1) = \{ax \to b, cx \to d\}$
- $V(H_2) = \{a, b, c, d\}$ and $E(H_2) = \{ab \to c, ab \to d\}$

Here, the subscripts indicate the number of tail vertices common to both edges. The I graphs also share a head vertex while the H graphs do not. We will determine the oriented and extremal numbers for each of these graphs in Chapters 5-8.

The proofs that follow rely heavily on the concept of a link graph. For undirected r-graphs, the link graph of a vertex is the (r-1)-graph induced on the remaining vertices such that each (r-1)-set is an (r-1)-edge if and only if that set together with the specified vertex makes an r-edge in the original r-graph [29]. In the directed

hypergraph model here, there are a few ways that we could define the link graph of a vertex. We will need the following three definitions.

DEFINITION 1.17. Let $x \in V(H)$ for some graph H. The tail link graph of $x T_x$ is the simple undirected 2-graph on the other n-1 vertices of V(H) with edge set defined by all pairs of vertices that exist as tails pointing to x in some edge of H. That is, $V(T_x) = V(H) \setminus \{x\}$ and

$$E(T_x) = \{ yz : yz \to x \in H \}.$$

The size of this set, $|T_x|$ will be called the tail degree of x. The degree of a particular vertex y in the tail link graph of x will be denoted $d_x(y)$.

Similarly, let D_x be the directed link graph of x on the remaining n-1 vertices of V(H). That is, let $V(D_x) = V(H) \setminus \{x\}$ and

$$E(D_x) = \{ y \to z : xy \to z \in E(H) \}.$$

Finally, let L_x denote the total link graph of x on the remaining n-1 vertices. That is, $V(L_x) = V(H) \setminus \{x\}$ and

$$E(L_x) = E(T_x) \cup E(D_x).$$

So L_x is a partially directed 2-graph.

The following notation will also be used when we want to count edges by tail sets.

DEFINITION 1.18. For any pair of vertices $x, y \in V(H)$ for some graph H let t(x, y) denote the number of edges with tail set $\{x, y\}$. That is

$$t(x,y) = |\{v: xy \to v \in E(H)\}|.$$

CHAPTER 2

The 4-resolvent Graph R_4

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

In [32], the authors gave a simple construction for an R_4 -free graph. Partition the vertices into sets T and K and take all possible edges with tail sets in T and head vertex in K. When there are n vertices, this construction gives $\binom{t}{2}(n-t)$ edges where t = |T|. This is optimized when $t = \lceil \frac{2n}{3} \rceil$. In [31], the authors showed that this number of edges is maximum for R_4 -free graphs for sufficiently large n and that the construction is the unique extremal R_4 -free graph.

We now give an alternate shorter proof that $\lfloor \frac{n}{3} \rfloor \begin{pmatrix} \lceil \frac{2n}{3} \rceil \\ 2 \end{pmatrix}$ is an upper bound on the extremal number for R_4 for sufficiently large n in both the standard and oriented versions of the problem. The proof also establishes the uniqueness of the construction.



FIGURE 2. The 4-resolvent graph R_4 .



FIGURE 3. The lower bound construction for a graph with no R_4 .



FIGURE 4. *H* contains a copy of R_4 if and only if the link graph of some vertex *v* contains a directed edge and an undirected edge that do not intersect.

THEOREM 2.1. For all $n \ge 29$,

$$ex_o(n, R_4) = \left\lfloor \frac{n}{3} \right\rfloor \begin{pmatrix} \left\lceil \frac{2n}{3} \right\rceil \\ 2 \end{pmatrix}$$

and for all $n \geq 56$,

$$ex(n, R_4) = \left\lfloor \frac{n}{3} \right\rfloor \begin{pmatrix} \left\lfloor \frac{2n}{3} \right\rfloor \\ 2 \end{pmatrix}.$$

Moreover, in each case there is one unique extremal construction up to isomorphism when $n \equiv 0, 1 \mod 3$ and exactly two when $n \equiv 2 \mod 3$.

PROOF. In either the standard or the oriented model, let H be an R_4 -free graph on n vertices. Partition V(H) into sets $T \cup K \cup B$ where T is the set of vertices that appear in tail sets of edges but never appear as the head of any edge, K is the set of vertices that do not belong to any tail set, and B is the set of vertices that appear as both heads and tails.

If B is empty, then H is a subgraph of some R_4 -free graph with the same structure as the lower bound construction. Therefore, H is either isomorphic to this construction or has strictly fewer edges. So assume that there exists some $v \in B$. The link graph L_v must contain at least one undirected edge and at least one directed edge. If any undirected edge is independent from any directed edge in L_v , then v would be the intersection vertex for an R_4 in H. Therefore, every directed edge in L_v is incident to every undirected edge.



FIGURE 5. A simple graph on n-1 vertices with red and blue edges such that each red edge is incident to each blue edge and there is at least one blue edge, xy, and at least one red edge, yz, can have no edge contained in the remaining n-4 vertices. Moreover, only red edges can go from x to the remaining vertices and only blue edges can go from zto the remaining vertices.

We want to show that if $v \in B$, then $|E(L_v)| = O(n)$. Determining an upper bound on the number of edges in L_v is equivalent to determining an upper bound on the number of red and blue edges on n-1 vertices such that each red edge is incident to each blue edge and there is at least one edge of each color.

If we are working in the oriented model where multiple edges on the same triple are not allowed then no pair of vertices in L_v can hold more than one edge. If we are working in the standard model, then two vertices in this graph may have up to three edges between them, say two red and one blue.

First, we consider the oriented version. In this case we have at least one edge of each color and they must be incident. So let xy be blue and let yz be red. Then all other edges must be incident to x, y, or z. Moreover, any edge from x to the remaining n - 4 vertices must be red since it is independent from yz and any edge from z to the remaining n - 4 must be blue. Therefore, there are at most 2(n - 4) edges from $\{x, y, z\}$ to the remaining n - 4 vertices.



FIGURE 6. When two vertices are allowed to have up to two red edges and one blue edge, then an adjacent red and blue edge pair is either incident in one or two vertices.

In the standard case our initial two red and blue edges may either be incident as before with xy blue and yz red or they might be incident in two vertices so that xyholds both a red and a blue edge. If none of the first type of incidence exists, then there can be at most 3 edges, all on xy.

So assume that the first type of incidence exists - xy is a blue edge and yz is a red edge. As before, all other edges must be incident to these three vertices such that any edge from x to the remaining n - 4 vertices must be red, and any edge from z to these vertices must be blue. Edges from y may be either color.

However, note that if any vertex of the n - 4 has a red edge from x, then none of the other vertices can have a blue edge from y or z. Similarly, any vertex with a blue edge from z means that no other vertices can have red edges from x or y. Therefore, if x has more than one red neighbor among the n - 4 vertices, then there are at most 4(n - 4) edges between $\{x, y, z\}$ and the n - 4 remaining vertices (since red edges have multiplicity up to 2). If z has more than one blue neighbor, then there are at most 2(n - 4) edges between $\{x, y, z\}$ and the n - 4 remaining vertices. Otherwise, x and z each have at most one neighbor among the n - 4 vertices, and the best we can do is 3(n - 4) edges, all from y. Therefore, there are at most 4(n - 4) additional edges. In either the standard or oriented versions of the problem, edges that do not contain vertices of B must have their tails in T and their heads in K. So there are at most

$$\left\lfloor \frac{n-b}{3} \right\rfloor \begin{pmatrix} \left\lceil \frac{2(n-b)}{3} \right\rceil \\ 2 \end{pmatrix}$$

edges that do not intersect B where b = |B|. Hence,

$$|E(H)| < \left\lfloor \frac{n-b}{3} \right\rfloor \left(\begin{bmatrix} \frac{2(n-b)}{3} \\ 2 \end{bmatrix} \right) + cnb$$

where c = 2 in the oriented case and c = 5 in the standard case.

This expression is maximum on $b \in [0, n]$ only at the endpoint b = 0 for all $n \ge 29$ when c = 2 and for all $n \ge 56$ when c = 4.

Therefore, we can never do better than the lower bound construction. Moreover, since B must be empty to reach this bound, then the construction is unique when $n \equiv 0, 1 \mod 3$. When $n \equiv 2 \mod 3$, then

$$\left\lfloor \frac{n}{3} \right\rfloor \begin{pmatrix} \left\lceil \frac{2n}{3} \right\rceil \\ 2 \end{pmatrix} = \left\lceil \frac{n}{3} \right\rceil \begin{pmatrix} \left\lfloor \frac{2n}{3} \right\rfloor \\ 2 \end{pmatrix}$$

so there are exactly two non-isomorphic extremal constructions in that case.

CHAPTER 3

The 3-resolvent Graph R_3

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

In [32], the authors gave a simple construction for an R_3 -free graph. Partition the vertices into sets A and B and take all possible edges with a tail set in A and head vertex in B plus all possible edges with a tail set in B and head vertex in A. When there are n vertices, this construction gives $(n-a)\binom{a}{2} + a\binom{n-a}{2}$ edges where a = |A|. This is optimized when $a = \lceil \frac{n}{2} \rceil$. The authors showed that this number of edges is asymptotically equivalent to the extremal numbers for R_3 .

We show that in both the standard and the oriented versions of this problem that this construction is in fact the best that we can do. We will start with the oriented case since it is less technical.



FIGURE 7. The 3-resolvent graph R_3 .



FIGURE 8. The unique R_3 -free extremal construction.



FIGURE 9. Forbidden intersection types in L_x for any vertex x in an R_3 -free graph.

1. The oriented version

THEOREM 3.1. For all n,

$$ex_o(n, R_3) = \left\lfloor \frac{n}{2} \right\rfloor \left\lceil \frac{n}{2} \right\rceil \frac{n-2}{2}.$$

Moreover, there is one unique extremal R_3 -free construction up to isomorphism for each n.

PROOF. Let H be an R_3 -free oriented graph on n vertices. Consider the total link graph, L_x , for some $x \in V(H)$. If

$$yz, z \to t \in E(L_x)$$

or if

$$y \to z, z \to t \in E(L_x),$$

then H is not R_3 -free (See Figure 9).

Let $U_x \subseteq V(L_x)$ be the set of vertices that appear as the tail vertex of some directed edge in L_x . Then no edges of L_x can be contained entirely inside U_x - it is an independent set with respect to both directed and undirected edges. Moreover, all undirected edges of L_x must appear entirely within the complement, $C_x := V(L_x) \setminus U_x$. Hence, if we let $u_x = |U_x|$, then

$$2|E(H)| = \sum_{x \in V(H)} |D_x| \le \sum_{x \in V(H)} u_x(n-1-u_x).$$

Each term of this sum is maximized when $u_x \in \left\{ \lfloor \frac{n-1}{2} \rfloor, \lceil \frac{n-1}{2} \rceil \right\}$. Therefore, the result is immediate if n is even. The situation is slightly more complicated for odd n.



FIGURE 10. The structure of L_x for any x in an R_3 -free graph

In this case,

$$u_x(n-1-u_x) \le \left(\frac{n-1}{2}\right)^2$$

for each x. However, we need $u_x = \frac{n-1}{2}$ in order to attain this maximum value. This would mean that there are $\frac{n-1}{2}$ vertices in C_x , and so there are at most $\left(\frac{n-1}{2}\right)$ edges in T_x . Therefore, if $u_x = \frac{n-1}{2}$ for each $x \in V(H)$, then

$$|E(H)| = \sum_{x \in V(H)} |T_x| < \frac{(n-2)(n-1)(n+1)}{8} = \left\lfloor \frac{n}{2} \right\rfloor \left\lceil \frac{n}{2} \right\rceil \frac{n-2}{2}.$$

Hence, we must assume that there exist some vertices for which $u_x \neq \frac{n-1}{2}$. For each x let $i_x \in \{0, \dots, \frac{n-1}{2}\}$ be the integer such that

$$u_x(n-1-u_x) = \left(\frac{n-1}{2} - i_x\right) \left(\frac{n-1}{2} + i_x\right).$$

Then,

$$|E(H)| \le \frac{1}{2} \sum_{x \in V(H)} \left(\frac{n-1}{2} - i_x \right) \left(\frac{n-1}{2} + i_x \right) = \frac{n(n-1)^2}{8} - \frac{1}{2} \sum_{j=0}^{\frac{n-1}{2}} k_j j^2$$

where k_j is the number of vertices $x \in V(H)$ for which $i_x = j$.

Since the construction gives $\frac{(n-2)(n-1)(n+1)}{8}$ for odd n, then we are only interested in beating this. So set

$$\frac{(n-2)(n-1)(n+1)}{8} \le \frac{n(n-1)^2}{8} - \frac{1}{2}\sum_{j=0}^{\frac{n-1}{2}}k_j j^2$$

This gives

(1)
$$\sum_{j=0}^{\frac{n-1}{2}} k_j j^2 \le \frac{n-1}{2}.$$

Since we can also find |E(H)| by counting the number of undirected edges over the L_x , then we can upper bound the number of these by assuming $u_x = \frac{n-1}{2} - i_x$ for each x since this increases the size of C_x . This gives

$$|E(H)| \le \sum_{x \in V(H)} {\binom{\frac{n-1}{2} + i_x}{2}} = \frac{n^3 - 4n^2 + 3n}{8} + \frac{1}{2} \sum_{j=0}^{\frac{n-1}{2}} j(n+j-2)k_j.$$

We can also set this greater than or equal to the known lower bound:

$$\frac{(n-2)(n-1)(n+1)}{8} \le \frac{n^3 - 4n^2 + 3n}{8} + \frac{1}{2}\sum_{j=0}^{\frac{n-1}{2}} j(n+j-2)k_j$$

to get

(2)
$$\frac{(n-1)^2}{2} \le \sum_{j=0}^{\frac{n-1}{2}} k_j j^2 + (n-2) \sum_{j=0}^{\frac{n-1}{2}} k_j j.$$

Subtracting (1) from (2) gives

$$\frac{(n-1)(n-2)}{2} \le (n-2)\sum_{j=0}^{\frac{n-1}{2}} k_j j.$$

Therefore,

$$\sum_{j=0}^{\frac{n-1}{2}} k_j j^2 \le \frac{n-1}{2} \le \sum_{j=0}^{\frac{n-1}{2}} k_j j,$$

and so

$$0 \le \sum_{j=0}^{\frac{n-1}{2}} k_j (j-j^2).$$

Since $j - j^2 < 0$ for any $j \ge 2$ and $j - j^2 = 0$ when j = 0, 1, then $k_j = 0$ for all $j \ge 2$.

Moreover, once all these are set to zero we get that

$$k_1 \le \frac{n-1}{2} \le k_1.$$

Therefore, $k_1 = \frac{n-1}{2}$ and so $k_0 = \frac{n+1}{2}$ since $\sum k_j = n$. This gives the desired upper bound.

Now we can show that the lower bound construction is the unique extremal example up to isomorphism. Let H be an extremal example on n vertices, and define a relation, \sim , on the vertices such that $x \sim y$ if and only if either x = y or $y \in U_x$. This defines an equivalence relation on V(H). Reflexivity and symmetry are both immediate. For transitivity note that the proof of the upper bound requires that every possible directed edge be taken from U_x to C_x for each $x \in V(H)$. Therefore, if we assume towards a contradiction that $y \in U_x$ and $z \in U_y$ but $z \notin U_x$, then $z \in C_x$. So $xy \to z \in E(H)$ which means $z \in C_y$, a contradiction.

When *n* is even there must be exactly two equivalence classes each of size $\frac{n}{2}$. Similarly, when *n* is odd there must be two equivalence classes of sizes $\frac{n-1}{2}$ and $\frac{n+1}{2}$. Therefore, the lower bound construction must be unique.

2. The standard version

THEOREM 3.2. For all $n \ge 6$,

$$ex(n, R_3) = \left\lfloor \frac{n}{2} \right\rfloor \left\lceil \frac{n}{2} \right\rceil \frac{n-2}{2}.$$

Moreover, there is one unique extremal R_3 -free construction up to isomorphism for each n.

PROOF. Let H be an R_3 -free graph on n vertices. Let $x \in V(H)$, and call any pair of vertices in L_x a multiedge if they contain more than one edge. Let $V(L_x) = U_x \cup C_x \cup M_x$ where M_x is the set of vertices that are incident to multiedges (that is, the minimal subset of vertices that contains all multiedges) and U_x and C_x are defined on the rest of the vertices as in Theorem 3.1. The goal is to show that if M_x is nonempty for any vertex x, then H has strictly fewer than the number of edges in the unique oriented construction given in Theorem 3.1. Therefore, that construction must be the unique extremal R_3 example for the standard problem as well.

There are three possibilities for multiedges in M_x : two oppositely directed edges, one directed edge and one undirected edge, and one undirected edge with two oppositely directed edges. If $y, z \in M_x$ have two directed edges between them, then neither ynor z is incident to any other edge in L_x since any incidence would create one of the two forbidden edge incidences of L_x as discussed in the previous theorem.

If y and z have only one directed edge (assume it is $y \to z$) and one undirected edge between them, then y cannot be incident to any more edges for the same reason as before, but z can be incident to undirected edges as well as directed edges with z at the head. This means that z may be the vertex of intersection of a star of these types of multiedges within M_x . Between any two such stars, the vertices of intersection may have an undirected edge between them, but no directed edges.

Therefore, the structure of the internal directed edges of M_x looks like Figure 11 with only the vertices of intersection of the single directed edge stars able to accept more edges from the rest of L_x . Directed edges from the rest of the graph to M_x must originate in U_x . Therefore, if M_x consists of d double directed edge pairs of vertices and k single directed stars with the *i*th star containing s_i vertices, then the total number of directed edges incident to vertices of M_x is at most

$$2d + \sum_{i=1}^{k} (s_i - 1 + u)$$



FIGURE 11. Example structure of M_x with 3 single directed edge stars and 4 double directed pairs.

where u is the number of vertices in U_x .

If we assume that M_x is nonempty, then $|M_x| = m \ge 2$. The number of directed edges incident to or inside of M_x is at most m + k(u - 1). Therefore, for $u \ge 2$, the number of directed edges incident to vertices of M_x is maximized when the number of single directed edge stars is maximized. This is $\lfloor \frac{m}{2} \rfloor$ stars. Therefore, there are at most

$$\frac{m}{2}(u+1)$$

directed edges incident to vertices of M_x . Thus, if $|C_x| = c$, then L_x can have at most $uc + \frac{m}{2}(u+1)$ directed edges. And since $u \ge 2$, then

$$uc + \frac{m}{2}(u+1) < u(c+m).$$

So L_x has strictly less directed edges than a complete bipartite graph on the same number of vertices would. In Theorem 3.1 every L_x needed to be a complete bipartite graph in terms of the directed edges in order for the maximum number of edges to be obtained, and only in the case of odd n could some of these bipartitions be less than equal or almost equal. In those cases the parts could only have $\frac{n-1}{2} - 1$ and $\frac{n-1}{2} + 1$ vertices. Therefore, the only way that u(c+m) could have more than this is if u = c + m and so $u = \frac{n-1}{2}$.

We assume that $m \ge 2$ and $u \ge 2$, but if both are equal to 2, then c = u - m = 0and n = 4, a contradiction since n is odd. Therefore, one of them must be strictly greater. So

$$uc + \frac{m}{2}(u+1) < (u-1)(u+1) = \left(\frac{n-1}{2} - 1\right)\left(\frac{n-1}{2} + 1\right).$$

This leaves only the cases where u = 0 and u = 1 which are both trivial.

So every link graph of H that contains a multiedge has strictly fewer than $(\frac{n-1}{2})^2 - 1$ directed edges. This is enough to prove that an extremal R_3 -free graph on an even number of vertices must be oriented. However, if there are an odd number of vertices it is possible that there could be enough directed link graphs with the maximum $(\frac{n-1}{2})^2$ directed edges to make up the deficit for the directed link graphs with strictly less than $(\frac{(n-3)(n+1)}{4})$ due to multiedges.

In this case there would need to be at least $\frac{n+3}{2}$ vertices with directed link graphs that are complete bipartite graphs with parts of size $\frac{n-1}{2}$ each. Let S be the set of these vertices. For any $x, y \in S$ define the relation $x \sim y$ if and only if $y \in U_x$. As in the proof of Theorem 3.1, this turns out to be an equivalence relation. By the definition of S one equivalence class can hold at most $\frac{n+1}{2}$ vertices. So there must be two nonempty classes. Let these classes be A and B.

Given some $x, y \in A$, suppose there is some $z \notin S$ such that $z \in U_x$ and $z \notin U_y$. Then it follows that $z \in C_y$ and therefore there is an edge $xy \to z$ and an edge $xz \to w$ for some $w \in C_x$. Together these make a copy of R_3 , a contradiction. Therefore, any zthat is in U_x for some $x \in A$ is in U_y for all $y \in A$.

Let C be the set of vertices that are in every U_x for $x \in A$ but not in A itself. Since A is nonempty, there is at least one vertex $x \in A$, and by definition $|U_x| = \frac{n-1}{2}$. Therefore, $|A| + |C| = \frac{n+1}{2}$. Similarly, let D be the set of vertices that are in every U_x for $x \in B$ but are not in B itself. By the same reasoning we get that $|B| + |D| = \frac{n+1}{2}$. Hence, |A| + |B| + |C| + |D| = n + 1. However, note that the sets A, B, C, and D are disjoint. So $|A| + |B| + |C| + |D| \le n$, a contradiction. This is enough to show the result.
CHAPTER 4

The Escher Graph E

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

In this chapter, we will prove the following result on the maximum number of edges of an E-free graph.

THEOREM 4.1. For all n,

$$ex(n,E) = \binom{n}{3} + 2$$

and there are exactly two extremal construction up to isomorphism for each $n \ge 4$.

But first we will prove the easier oriented version of the problem. This result will be needed to prove Theorem 4.1.

1. The oriented version

THEOREM 4.2. For all n,

$$ex_o(n, E) = \binom{n}{3}$$

and there is exactly one extremal construction up to isomorphism.



FIGURE 12. The Escher graph E.



FIGURE 13. An "almost" linear ordering on the vertices of an E-free directed hypergraph.

PROOF. The upper bound here is trivial so we need only come up with an *E*-free construction that uses $\binom{n}{3}$ edges. Let *H* be the directed hypergraph defined on vertex set V(H) = [n] and edge set,

$$E(H) = \{ab \to c : a < b < c\}.$$

That is take some linear ordering on the n vertices and for each triple direct the edge to the largest vertex. Then every triple has an edge and H contains no copy of E.

Now we will show that this construction is unique. Let H be an E-free graph on n vertices and $\binom{n}{3}$ edges. Define a relation on the vertices, \prec , where $x \prec y$ if and only if there exists an edge in E(H) with x in the tail and y as the head vertex. Then \prec is a partial ordering of the vertices that is almost linear in that every pair of vertices are comparable except for the two smallest elements (see Figure 13).

We now shift our attention to the standard version of the problem where a triple of vertices can have more than one edge. Here, both of the lower bound constructions are similar to the unique extremal construction in the oriented version.

2. Two lower bound constructions for ex(n, E)

The first construction is the same as the extremal construction in the oriented case but with two additional edges placed on the "smallest" triple. That is, let $H_1 = ([n], E_1)$ where

$$E_1 = \{ab \to c : a < b < c\} \cup \{13 \to 2, 23 \to 1\}.$$

See Figure 14.



FIGURE 14. The first extremal construction, H_1 , for an *E*-free directed hypergraph on *n* vertices.



FIGURE 15. The second extremal construction, H_2 , for an *E*-free graph on *n* vertices.

Moreover, it is important to note that if an *E*-free graph with $\binom{n}{3} + 2$ edges has at least one edge on every vertex triple, then it must be isomorphic to H_1 . This is because we can remove two edges to get an *E*-free subgraph where each triple has exactly one edge. Therefore, this must be the unique extremal construction established in Theorem 4.2. The only way to add two edges to this construction and avoid creating an Escher graph is to add the additional edges to the smallest triple under the ordering.

The second construction is also based on the oriented extremal construction. Let $H_2 = ([n], E_2)$ where

$$E_2 = (E_1 \setminus \{23 \to 4, 23 \to 1\}) \cup \{14 \to 2, 14 \to 3\}.$$

See Figure 15.

For the rest of this section we will show that any *E*-free graph is either isomorphic to one of these two constructions or has fewer than $\binom{n}{3} + 2$ edges. Roughly speaking, the

strategy is to take any E-free graph and show that we can add and remove edges to it so that we preserve E-freeness, remove most multiple edges from triples that had more than one, and never decrease the overall number of edges.

3. Add and remove edges

Let H be an E-free graph and represent its vertices as the disjoint union of three sets:

$$V(H) = D \cup R \cup T$$

where D (for 'Done') is the set of all vertices that have complete graphs on three or more vertices as tail link graphs, R (for 'Ready to change') is the set of vertices not in D that have at least three edges in their tail link graphs, and T is the set of all other vertices (those with 'Two or fewer edges in their tail link graphs').

The plan is now to remove and add edges in order make a new graph H' which is also E-free, has at least as many edges as H, and whose vertices make a disjoint union,

$$V(H') = D' \cup T'$$

where D' and T' are defined exactly the same as D and T except in terms of the vertices of H'.

That is, for each vertex $x \in R$, we will add all possible edges to complete T_x . This moves x from R to D. The edges removed will be all those that pointed from x to a vertex that points to x. This will destroy triples with more than one edge as we go. The following observation will ensure that this procedure only ever moves vertices from R to D, from R to T, from R to R, and from T to T. Since each step moves one vertex from R to D and ends when R is empty, then the procedure is finite. Here is the observation: LEMMA 4.1. Let H be an E-free graph, and let $x, y \in V(H)$. If $d_x(y), d_y(x) > 0$, then $d_x(y) = d_y(x) = 1$. In other words, for any two vertices, x and y, if $d_y(x) \ge 2$, then $d_x(y) = 0$.

PROOF. Suppose not. Let $d_x(y), d_y(x) > 0$ and suppose $d_x(y) \ge 2$. Then there exist two distinct vertices, a and b such that

$$ay \to x, by \to x \in E(H).$$

There also exists a vertex c such that $xc \to y \in E(H)$. Since c must be distinct from either a or b if not both, then this yields an Escher graph.

Now, let us make the procedure slightly more formal: While there exist vertices in R, pick one, $x \in R$, and for each pair $a, b \in V(T_x)$, add the edge $ab \to x$ to E(H) if it is not already an edge. Then, for each $a \in V(T_x)$, remove all edges of E(H) of the form $xs \to a$ for any third vertex s.

Since there were at least three edges in T_x , then the added edges will move x from R to D. The removed edges, if any, will only affect vertices in R or in T since if xs is removed from T_a , then this implies that $a \in T_x$ and that $x \in T_a$ and so both had degree one in the other's tail link graph. Hence, $a \notin D$. Moreover, an affected vertex in R will either stay in R or move to T while an affected vertex in T will stay in T since it is only losing edges from its tail link graph.

At the end of this process D' will contain no triple of vertices with more than one edge. Therefore, the only such triples of vertices of H' will be entirely in T' or will consist of vertices from both T' and D'. We will show later that there cannot be too many of these triples. First, we need to show that after each step of this procedure, no Escher graph is created and at least as many edges are added to the graph as removed.

4. No copy of E is created and the number of edges can only increase

Fix a particular vertex $x \in R$ to move to D. Add and remove all of the designated edges. Suppose that we have created an Escher graph. Since the only added edges point to x, then the configuration must be of the form, $ab \to x, xc \to a$ for some distinct vertices, a, b, and c. Therefore, $a \in V(T_x)$ and so $xc \to a$ would have been removed in the process.

Now we will show that at least as many edges have been added to H as removed by induction on the number of independent edges in T_x . Start by assuming there are 0 independent edges in T_x and assume that there are k vertices in T_x that have degree one. Then at most k edges will be removed. If k = 0, then no edges are removed and there is a strict increase in the number of edges.

If k = 1, then let y_1 be the vertex with degree one and let y_2 be the vertex it is incident to. Since $d_x(y_2) \neq 1$ and $d_x(y_2) \geq 1$, then $d_x(y_2) \geq 2$. So there exists a third vertex, y_3 , and similarly, $d_x(y_3) \geq 2$ but y_3 is not adjacent to y_1 . Hence, there exists a fourth vertex, y_4 . So at most one edge is removed and at least two edges are added, $y_1y_3 \rightarrow x$ and $y_1y_4 \rightarrow x$. Therefore, there is a strict increase in the number of edges. If k = 2, then the fact that T_x has at least three edges means that there must be at least two additional vertices in T_x . Hence, at most two edges are removed but at least three are added. If $k \geq 3$, then at most k are removed but $\binom{k}{2}$ are added which nets

$$\binom{k}{2} - k = \frac{k(k-3)}{2} \ge 0$$

edges added.

Now, for the induction step, assume that T_x has m > 0 independent edges and that the process on a T_x with m - 1 independent edges adds just as many edges as it removes. Let yz be an independent edge in T_x and let A be the set of vertices of T_x that are not y or z. Since T_x has at least three edges, then A contains at least three vertices. Therefore, the number of added edges is at least 6 between A and $\{y, z\}$. 6. CASE 2: $|T'| \leq 4$ AND SOME $x \in T'$ GIVES AN T_x WITH 2 INDEPENDENT EDGES 32 The number of edges removed from T_y and T_z together is at most 2. By assumption, the number of edges removed from the other tail link graphs of vertices in A is offset by the number of edges added inside A. Therefore, there is a strict increase in the number of edges.

To summarize, we have shown that H' is an *E*-free graph such that

$$|E(H)| \le |E(H')|$$

and

$$V(H') = D' \cup T'$$

such that any triple of vertices of H' with more than one edge must intersect the set T'. We will now consider what is happening in T' by cases.

5. Case 1: $|T'| \ge 5$

Let $T' = \{x_1, x_2, \dots, x_t\}$ for $t \ge 5$. For each x_i remove all edges of H' that have x_i as a head. By the definition of T' this will remove at most 2t edges from H'.

Next, add all edges to T' that follow the index ordering. That is, for each triple $\{x_i, x_j, x_k\}$ add the edge that points to the largest index, $x_i x_j \to x_k$ where i < j < k. This will add $\binom{t}{3}$ edges. The new graph has

$$\binom{t}{3} - 2t \ge 0$$

more edges than H'. Moreover, it is E-free and oriented. Therefore, $|E(H)| < \binom{n}{3}$.

6. Case 2: $|T'| \leq 4$ and some $x \in T'$ gives an T_x with 2 independent edges

Assume that some $x \in T'$ has a tail link graph T_x such that $ab, cd \in E(T_x)$ for four distinct vertices, $\{a, b, c, d\}$. If

$$d_a(x) = d_b(x) = d_c(x) = d_d(x) = 1,$$

then $a, b, c, d, x \in T'$, a contradiction of the assumption that $|T'| \leq 4$.

Therefore, we can add the edges

$$ac \to x, ad \to x, bc \to x, bd \to x$$

and remove any edges that point to a vertex from $\{a, b, c, d\}$ with x in the tail set. Because x has zero degree in at least one of those tail link graphs, then we have removed at most three edges and added four, a strict increase. We have also not created any triples of vertices with more than one edge or any Escher graphs.

We may now assume that $|T'| \leq 4$ and that the tail link graphs of vertices in T' are never two independent edges.

7. Case 3: |T'| = 0, 1, 2

First, note that if H' has a triple with more than one edge $\{x, y, z\}$ then at least two of its vertices must be in T' as a consequence of Lemma 4.1. Therefore, if |T'| = 0, 1, then H' is oriented and so

$$|E(H)| \le |E(H')| \le \binom{n}{3}.$$

Moreover, if $T' = \{x, y\}$ and H' is not oriented, then any vertex triple with more than one edge must have two edges of the form,

$$zx \to y, zy \to x$$

for some third vertex z. If there exist two such vertices $z_1 \neq z_2$ that satisfy this, then there would be an Escher graph. Hence, there is at most one vertex triple with more than one edge and it would have at most two edges. Therefore,

$$|E(H)| \le |E(H')| \le \binom{n}{3} + 1.$$

First, suppose that there exists a triple $\{x, y, z\}$ with all three possible edges. Then $T' = \{x, y, z\}$. Since any triple with multiple edges must intersect T' in at least two vertices, then any additional such triple would make an Escher graph with one of the edges in T'. Therefore, H' has exactly one triple of vertices with all three edges on it and no others. So

$$|E(H) \le |E(H')| \le \binom{n}{3} + 2.$$

Moreover, to attain this number of edges, no triple of vertices can be empty of edges. In this case, H' must be isomorphic to the first construction H_1 .

Next, assume that no triple of vertices has all three edges and let $T' = \{x, y, z\}$. Therefore, H' needs at least two triples of vertices that each hold two edges or else

$$|E(H)| \le |E(H')| \le \binom{n}{3} + 1$$

automatically. Suppose one of the multiedges is $\{x, y, z\}$ itself. Then without loss of generality let the edges be $xy \to z$ and $xz \to y$. The second triple with two edges must have its third vertex in D'. Call this vertex v. The vertex x cannot be in this second triple of vertices without creating an Escher graph. So the edges must be $vy \to z$ and $vz \to y$. But this also creates an Escher graph.

Therefore, neither of the two triples that hold two edges are contained entirely within T'. So without loss of generality they must be $vx \to y, vy \to x$ and $wy \to z, wz \to y$. If $v \neq w$, then $vx, wz \in T_y$, a contradiction to our assumption that T' contains no vertices with tail link graphs that are two independent edges. Hence, v = w.

Since $v \in D$, then T_v has at least three vertices. Moreover, since v is in the tail link graphs of each vertex of T', then none of these vertices can be in T_v . Remove all edges pointing to the vertices of T'. This is at most 6 edges. Add all possible edges with v as the head and a tail set among the set $V(T_v) \cup \{x, y, z\}$. This adds at least 12 new edges. The new graph is oriented and E-free. Therefore, $|E(H)| < {n \choose 3}$.

First, assume that there is some triple $\{x, y, z\}$ that contains all three possible edges. As before, there are no additional triples with more than one edge. So

$$|E(H)| \le |E(H')| \le \binom{n}{3} + 2$$

The first construction H_1 is the unique extremal construction under this condition since all triples must be used at least once.

So assume that all triples with more than one edge have two edges each. Then we must have at least two. Assume that one of them is contained within $T' = \{a, b, c, d\}$. Without loss of generality let it be $ab \rightarrow c, ac \rightarrow b$. Since the second such triple intersects T' in at least two vertices, then it must intersect $\{a, b, c\}$ in at least one vertex.

If it intersects $\{a, b, c\}$ in two vertices, then without loss of generality (to avoid a copy of E) the second triple must be of the form $ab \to x, ax \to b$. Hence, $x \in T'$ so x = d. But now there is no edge possible on $\{b, c, d\}$. Therefore, there must be a third such triple for H' to have $\binom{n}{3} + 2$ edges. This triple must be $ac \to d, ad \to c$. And the only way to actually make it to the maximum number of edges now must be to have an edge on every other triple.

Every triple of the form $\{b, c, s\}$ for $s \in D$ must have the edge $bc \to s$ since the other two options would create an Escher graph. Similarly, $bd \to s$ and $cd \to s$ are the only options for triples of the form $\{b, d, s\}$ and $\{c, d, s\}$ respectively. Next, any triple of the form $\{a, b, s\}$ must hold the edge $ab \to s$ since the other two edges create Escher graphs. Similarly, every triple of the forms $\{a, c, s\}$ and $\{a, d, s\}$ must hold the edges $ac \to s$ and $ad \to s$ respectively.

Since each triple contained in D holds exactly one edge, then the induced subgraph on D must be isomorphic to the oriented extremal example of an E-free graph on n - 4

vertices. Therefore, the entire graph H' must be isomorphic to the second extremal construction H_2 in order to attain $\binom{n}{3} + 2$ edges.

So assume that the second triple with two edges intersects $\{a, b, c\}$ in only one vertex. Then these edges must be $xa \to d, xd \to a$. This can be the only additional triple with two edges. So to make it to $\binom{n}{3} + 2$ edges we need each triple to have an edge. However, the edge for $\{a, b, d\}$ is forced to be $ad \to b$ and the edge for $\{b, c, d\}$ is forced to be $bc \to d$. This makes an Escher graph. So

$$|E(H)| \le |E(H')| \le \binom{n}{3} + 1.$$

Now assume that no vertex triple with multiple edges is contained entirely within T', but assume that there are at least two such triples in H'. The only way that two triples could have distinct vertices in D' is if they were of the forms (without loss of generality), $xa \to b, xb \to a$, and $yc \to d, yd \to c$. Otherwise, the pairs of the two triples that are in T' would intersect resulting in either a copy of E (if both triples use the same pair) or a vertex in T' with two independent edges as a tail link graph. So there must be exactly two such triples. Therefore, all other triples of vertices must contain exactly one edge in order to reach $\binom{n}{3} + 2$ edges overall. To avoid the forbidden subgraph this edge must be $ab \to c$ for the triple $\{a, b, c\}$ and $cd \to a$ for the triple $\{a, c, d\}$. But this is an Escher graph. Hence, not all triples may be used

$$|E(H)| \le |E(H')| \le \binom{n}{3} + 1$$

and so

Therefore, we may now assume for each multiedge triple that the vertex from D' is always x. First, assume that there are only two such triples. As before, if we assume that the only two such triples are $xa \to b, xb \to a$ and $xc \to d, xd \to c$, then there can be not be an edge on both $\{a, b, c\}$ and $\{a, c, d\}$. Hence, there would be a suboptimal number of edges overall. On the other hand, if the only two such triples are adjacent in T', then they are, without loss of generality, $xa \to b, xb \to a$ and $xb \to c, xc \to a$. In this case, no edge can go on the triple $\{a, b, c\}$ at all and so there are at most $\binom{n}{3} + 1$ edges overall.

Therefore, we must assume there are at least three such triples that meet at x. If these three triples make a triangle in T', then they are $xa \to b, xb \to a, xb \to c, xc \to b$, and $xc \to a, xa \to c$. Again, there can be no edges on the triple $\{a, b, c\}$. Hence, every other triple must hold an edge to attain $\binom{n}{3} + 2$ edges overall.

On the triple $\{a, b, d\}$ this edge must be $ab \to d$ to avoid making a copy of E. Similarly, we must have the edges $ac \to d$ and $bc \to d$. But this means that $d \notin T'$, a contradiction.

On the other hand, if there are three triples of vertices with more than one edge on each that do not make a triangle in T' or if there are four or more such triples, then x is in the tail link graphs for each vertex in T'. Hence, none of these vertices may be in the tail link graph, T_x . However, $x \in D'$ so its tail link graph has at least three vertices. Remove all edges pointing to vertices of T' (at most 8). Add all edges pointing to x with tail sets in T' (6 new edges) and between T' and $V(T_x)$ (at least 12 new edges). So this adds at least ten edges to H' to create H''. H'' is oriented so

$$|E(H)| < |E(H'')| \le \binom{n}{3}.$$

This exhausts all of the cases and establishes that

$$\operatorname{ex}(n, E) = \binom{n}{3} + 2$$

with exactly two extremal examples up to isomorphism.

CHAPTER 5

The Graph I_0

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

Let I_0 denote the forbidden graph where two edges intersect in exactly one vertex such that this vertex is the head of both edges. That is $V(I_0) = \{a, b, c, d, x\}$ and $E(I_0) = \{ab \rightarrow x, cd \rightarrow x\}$ (see Figure 16). In this chapter, we will prove the following result on the oriented extremal numbers of I_0 .

THEOREM 5.1. For all $n \ge 9$,

$$ex_o(n, I_0) = \begin{cases} n(n-3) + \frac{n}{3} & n \equiv 0 \mod 3\\ n(n-3) + \frac{n-4}{3} & n \equiv 1 \mod 3\\ n(n-3) + \frac{n-5}{3} & n \equiv 2 \mod 3 \end{cases}$$

with exactly one extremal example up to isomorphism when 3|n, exactly 18 nonisomorphic extremal constructions when

$$n \equiv 1 \mod 3$$

and exactly 32 constructions when

FIGURE 16. The graph I_0 .



FIGURE 17. $ab, cd \in E(T_x)$ if and only if $ab \to x, cd \to x \in H$.

The proof for this is rather long. However, the standard version of the problem is much simpler so we will begin there.

THEOREM 5.2. For each $n \ge 5$,

$$ex(n, I_0) = n(n-2)$$

and for each $n \ge 6$, there are exactly $(n-1)^n$ different labeled I_0 -free graphs that attain this maximum number of edges.

PROOF. Let H be I_0 -free on $n \ge 5$ vertices. For any $x \in V(H)$, the tail link graph T_x cannot contain two independent edges (see Figure 17). Therefore, by the Erdős-Ko-Rado Theorem [21] the edge structure of T_x is either a triangle or a star with k edges all intersecting in a common vertex for some $0 \le k \le n - 2$. So each vertex $x \in V(H)$ is at the head of at most n - 2 edges. Hence,

$$|E(H)| = \sum_{x \in V(H)} |E(T_x)| \le n(n-2).$$

On the other hand, many different extremal constructions exist that give n(n-2)edges on n vertices without the forbidden intersection. Let

$$f:[n]\to [n]$$

be any function such that $f(x) \neq x$ for any $x \in [n]$. Define H_f as the graph with vertex set $V(H_f) = [n]$ and edge set

$$E(H_f) = \bigcup_{x \in [n]} \left\{ f(x)y \to x : y \in [n] \setminus \left\{ x, f(x) \right\} \right\}.$$

Certainly each vertex x is at the head of n-2 edges and each of its tail sets contains f(x) which prevents the forbidden subgraph. So $|E(H_f)| = n(n-2)$, and H_f is I_0 -free for any such function f.

Moreover, there are $(n-1)^n$ different functions f that will make such a construction on [n]. So this gives us $(n-1)^n$ labeled extremal I_0 -free graphs. Conversely, since any I_0 -free graph with the maximum number of edges must have n-2 edges in T_x for each vertex x, then all tail link graphs must be (n-2)-stars for all $n \ge 6$. Therefore, these constructions give all possible extremal examples.

The oriented version of this problem is less straight forward, but determining $ex_o(n, I_0)$ also begins with the observation that every tail link graph of an I_0 -free graph will either be a triangle, a star, or empty. Broadly speaking, as n gets large, it would make more sense for most, if not all, tail link graphs to be stars in order to fit as many edges into an I_0 -free graph. This motivates the following auxiliary structure.

1. Gates

Let H be some I_0 -free graph. For each $x \in V(H)$ for which T_x is a star (with at least one edge), let g(x) denote the common vertex for the edges of T_x . We will refer to this vertex as the *gatekeeper* of x (in that it is the gatekeeper that any other vertex must pair with in order to "access" x). In the case where T_x contains only a single edge we may choose either of its vertices to serve as the gatekeeper. In this way, we have constructed a partial function, $g: V(H) \rightarrow V(H)$.



FIGURE 18. The structure of a connected component of the gate G.

Next, construct a directed 2-graph G on the vertex set V(H) based on this partial function:

$$y \to x \in E(G) \iff y = g(x).$$

We will call this digraph the *gate* of H (or more properly, G is the gate of H under g since g is not necessarily unique).

The edge structure of any gate G is not difficult to determine. Since g is a partial function, then each vertex has in-degree at most one in G. Therefore, the structure of any connected component of G can be described as a directed cycle on k vertices, C_k , for $1 \leq k$ (where k = 1 implies a single vertex) unioned with k disjoint directed trees, each with its root vertex on this cycle (see Figure 18). We will refer to this kind of general structure as a k-cycle with branches.

Let

$$\mathcal{C} = \bigcup_{k=1}^n \mathcal{C}_k$$

be the set of maximal connected components of a gate of H where, for each k, C_k is the set of maximal connected components that are k-cycles with branches. Note that

$$|E(H)| = \sum_{x \in V(H)} |T_x| = \sum_{C \in \mathcal{C}} \left(\sum_{x \in V(C)} |T_x| \right) = \sum_{k=1}^n \left(\sum_{C \in \mathcal{C}_k} \left(\sum_{x \in V(C)} |T_x| \right) \right).$$

$$\sum_{x \in V(C)} |T_x|$$

as a function of the number of vertices, |V(C)|, for any $C \in \mathcal{C}_k$.

2. Bounding $\sum_{x \in V(C)} |T_x|$ for any connected component C of the gate

Loosely speaking, each gatekeeper edge of a connected component C represents at most n-2 edges of H. We will arrive at an upper bound on the sum $\sum_{x \in V(C)} |T_x|$ by adding this maximum for each edge of C, and then subtracting the number of triples of vertices that such a count has included more than once. This will happen for any triple of vertices which contain two or three gatekeeper edges. We make this observation formal in the following definition and lemma.

DEFINITION 5.1. Let G be some gate and let C be a maximal connected component of G. Let P(C) be the set of $2 \rightarrow 1$ possible edges defined by

$$P(C) = \bigcup_{a \to b \in E(C)} \{av \to b : v \in V(H) \setminus \{a, b\}\}$$

LEMMA 5.2. Let G be a gate, and let C be a maximal connected component of G. If a set of three distinct vertices $\{x, y, z\} \subseteq V(C)$ are spanned by two gatekeeper edges of G, then P(C) contains at least two edges on these three vertices.

PROOF. Without loss of generality, the two spanning edges on $\{x, y, z\}$ are either of the form

$$x \to y \to z \text{ or } x \leftarrow y \to z.$$

In the former case, P(C) contains the edges $xz \to y$ and $yx \to z$. In the latter case, P(C) contains the edges $yz \to x$ and $yx \to z$.

Now comes the main counting lemma.

2. BOUNDING $\sum_{x \in V(C)} |T_x|$ FOR ANY CONNECTED COMPONENT C OF THE GATE 43 LEMMA 5.3. Let H be an I_0 -free graph on $n \ge 8$ vertices. Let G be a gate of H. Let C be a maximal connected component of G with m vertices. Then

- $\sum_{x \in V(C)} |T_x| \le m(n-3)$ if $C \in \mathcal{C}_k$ for any $k \ne 3$ with equality possible only if $C = C_k$ for some $k \ge 4$,
- $\sum_{x \in V(C)} |T_x| \le m(n-3) + 1$ if $C = C_3$, and
- $\sum_{x \in V(C)} |T_x| \le m(n-3)$ for all other $C \in \mathcal{C}_3$ with equality possible only if C is a 3-cycle with exactly one nonempty directed path coming off of it.

PROOF. For convenience let

$$S = \sum_{x \in V(C)} |T_x|.$$

Note that for each $x \in V(C)$ with in-degree one, $ab \in T_x$ implies that $ab \to x \in P(C)$. Hence, if $C \notin C_1$, then every edge counted in the sum S is in P(C). Moreover, |P(C)| = m(n-2).

If $C \in C_k$ for $k \ge 4$, then by Lemma 5.2 each intersection of gatekeeper edges of C yields two edges on the same triple of vertices in P(C). Conversely, since C contains no C_3 , then each distinct triple of vertices contains at most two gatekeeper edges. Therefore, each triple contains at most two edges of P(C). Hence,

$$S \le m(n-2) - \sum_{x \in V(C)} \binom{d_G(x)}{2}$$

where $d_G(x)$ denotes the total number of vertices incident to x in the gate.

Since C has m edges, then $\sum_{x \in V(C)} d_G(x) = 2m$. So

$$S \le m(n-2) - \sum_{x \in V(C)} {d_G(x) \choose 2} \le m(n-3)$$

by Jensen's Inequality. Moreover, equality happens if and only if $d_G(x) = d_G(y)$ for all $x, y \in V(C)$. Therefore, this inequality is strict for all $C \in \mathcal{C}_k$ unless $C = C_k$. 2. BOUNDING $\sum_{x \in V(C)} |T_x|$ FOR ANY CONNECTED COMPONENT *C* OF THE GATE 44 Similarly, if $C \in \mathcal{C}_2$, then P(C) contains at least $\sum_{x \in V(C)} {\binom{d_G(x)}{2}}$ multiedges for the same reason as before. But here there are an additional n-2 edges counted for each triple containing the C_2 . Also,

$$\sum_{x \in V(C)} d_G(x) = 2(m-1).$$

Hence,

$$S \le m(n-2) - (n-2) - \sum_{x \in V(C)} {\binom{d_G(x)}{2}} \le (m-1)(n-2) - m{\binom{\frac{2(m-1)}{m}}{2}}$$

by Jensen's Inequality. This is strictly less than m(n-3).

In the acyclic case, Lemma 5.2 implies that the sum of all $|T_x|$ for each $x \in V(C)$ other than the root vertex is less than or equal to

$$(m-1)(n-2) - \sum_{x \in V(C)} {d_G(x) \choose 2}.$$

The root vertex itself is the head vertex of at most 3 edges in H so Jensen's Inequality gives

$$S \le (m-1)(n-2) - m\binom{\frac{2(m-1)}{m}}{2} + 3 < m(n-3)$$

for all $n \geq 8$.

Finally, if $C \in C_3$, then each intersection of gatekeeper edges of C yields two edges on the same triple of vertices in P(C). However, exactly one triple of vertices contains three gatekeeper edges and has three edges in P(C). But the rest have at most two since there is only one triangle in C. Therefore, $\sum_{x \in V(C)} {\binom{d_G(x)}{2}}$ counts each triple of vertices that contain more than one gatekeeper edge exactly once except for the triple that makes up the C_3 which it counts three times. Since we must subtract off two edges in P(C) on these three vertices to eliminate repeated triples, then we must 2. BOUNDING $\sum_{x \in V(C)} |T_x|$ FOR ANY CONNECTED COMPONENT *C* OF THE GATE 45 subtract $\sum_{x \in V(C)} {d_G(x) \choose 2} - 1$ from |P(C)|. Therefore,

$$S \le m(n-2) - \sum_{x \in V(C)} {d_G(x) \choose 2} + 1.$$

So by Jensen's Inequality,

 $S \le m(n-3) + 1$

with equality possible only if all of the degrees $d_G(x)$ are equal. This can only happen if $C = C_3$.

If we want to see for which $C \in C_3$ the second best bound of m(n-3) could be attained, then we need to set

$$\sum_{x \in V(C)} \binom{d_G(x)}{2} = m + 1.$$

Assume that the vertices are x_1, \ldots, x_m , and for each x_i let

$$d_i = d_G(x_i) - 2.$$

Then $\sum_{i=1}^{m} d_i = 0$ and a quick calculation shows that $\sum_{i=1}^{m} d_i^2 = 2$. Therefore, the only possibility is for some $d_i = 1$ and another to equal -1 and all the rest must be 0. This corresponds with one vertex degree equal to 3, another equal to 1, and all others equal to 2. The only way that this can happen in a C_3 with branches is to have exactly one branch, and that branch must be a directed path.

This shows that the best we can hope for in terms of the average number of edges per vertex over any connected component of the gate is $n - 3 + \frac{1}{3}$, and this could be attained only in the case where the component is a directed triangle with no branches. Otherwise, the average number of edges of a component is at most n - 3, and this is attainable only if the component is a directed triangle with a single directed path coming off of one of its vertices or a directed k-cycle with no branches for some $k \ge 4$.



FIGURE 19. Structure of the gate for an extremal I_0 -free graph when $n \equiv 0 \mod 3$.

This is enough for us to establish the upper bound for $ex_o(n, I_0)$ and to characterize the necessary structure of the gate for any graph attaining this upper bound.

3. Upper bound on $ex_o(n, I_0)$

Let H be an I_0 -free graph on $n \ge 9$ vertices. Let G be a gate of H. Let C be the set of maximal connected components of G and break C into three disjoint subsets based on the maximum average number of edges attainable for the components in each. That is, let

$$\mathcal{C} = \mathcal{D}_1 \cup \mathcal{D}_2 \cup \mathcal{D}_3$$

where \mathcal{D}_1 contains all components with maximum average number of edges per vertex strictly less than n-3: those components that are either acyclic, contain a C_2 , contain a C_3 with nonempty branches that are more than just a single path, or contain a C_k for $k \ge 4$ with some nonempty branch; \mathcal{D}_2 is the set of all components with maximum number of edges per vertex of n-3: those that contain a directed C_3 and exactly one directed path or those that are a directed k-cycles for any $k \ge 4$ and no branches; and \mathcal{D}_3 is the set of components with a maximum average greater than n-3: the directed triangles.

For each i let d_i be the total number of vertices contained in the components of \mathcal{D}_i . Then

$$|E(H)| \le d_3\left(n-3+\frac{1}{3}\right) + (n-d_3)(n-3)$$

with equality possible only if $d_1 = 0$. This is enough to prove the following.



FIGURE 20. The only possible structures of the gate of an extremal I_0 -free graph when $n \equiv 1 \mod 3$.

LEMMA 5.4. Let H be an I_0 -free graph on $n \ge 9$ vertices such that $n \equiv 0 \mod 3$, then

$$|E(H)| \le n(n-3) + \frac{n}{3}.$$

Moreover, the only way for H to attain this maximum number of edges is if the gate of H is a disjoint union of directed triangles.

The next two lemmas give the maximum number when $n \equiv 1, 2 \mod 3$. There is only slightly more to consider in these cases.

LEMMA 5.5. Let H be an I_0 -free graph on $n \ge 9$ vertices such that $n \equiv 1 \mod 3$, then

$$|E(H)| \le n(n-3) + \frac{n-4}{3}.$$

Moreover, the only way for H to attain this maximum number of edges is if the gate of H is a disjoint union of $\frac{n-4}{3}$ directed triangles together with either a directed C_4 or a 3-cycle with an extra edge.

PROOF. Since $n \equiv 1 \mod 3$, then $d_3 \leq n-1$. If $d_3 = n-1$, then the gate consists of $\frac{n-1}{3}$ disjoint directed triangles and one isolated vertex which means that

$$|E(H)| \le (n-1)\left(n-3+\frac{1}{3}\right)+3.$$



FIGURE 21. The only possible structures of the gate of an extremal I_0 -free graph when $n \equiv 2 \mod 3$.

If $d_3 \leq n-4$, then we can do better with

$$|E(H)| \le (n-4)\left(n-3+\frac{1}{3}\right) + 4(n-3)$$

only in the case of $\frac{n-4}{3}$ disjoint directed triangles and one component from \mathcal{D}_2 in the gate. Therefore,

$$|E(H)| \le n(n-3) + \frac{n-4}{3}.$$

LEMMA 5.6. Let H be an I_0 -free graph on $n \ge 11$ vertices such that $n \equiv 2 \mod 3$, then

$$|E(H)| \le n(n-3) + \frac{n-5}{3}.$$

Moreover, the only way for H to attain this maximum number of edges is if the gate of H is a disjoint union of $\frac{n-5}{3}$ directed triangles together with either a directed C_5 or a 3-cycle with a directed path of two edges.

PROOF. Since $n \equiv 2 \mod 3$, then $d_3 \leq n-2$ and equality implies that G consists of $\frac{n-2}{3}$ disjoint directed triangles and two additional vertices that are either both isolated, contain one edge, or are a C_2 giving 6, 3 + (n-2), or n-2 additional edges respectively. The best we can do when $d_3 = n - 2$ is therefore,

$$|E(H)| \le (n-2)\left(n-3+\frac{1}{3}\right) + (n+1).$$

Otherwise, $d_3 \leq n-5$ and the best we can do is

$$|E(H)| \le (n-5)\left(n-3+\frac{1}{3}\right) + 5(n-3).$$

This is better. Moreover, this will happen only when the five non-triangle vertices are in a component (or components) of G that give an average of n-3. So they must either make a C_5 or a directed triangle with one path.

4. Lower bound constructions

The structure of the gates necessary to attain the maximum number of edges for a I_0 -free graph determined in the previous section are also sufficient. Of these gates, none of them have acyclic components. Therefore, any graph that produces one of these gates has only vertices with stars for tail link graphs. This immediately implies that there is no I_0 in any graph that has such a gate.

Moreover, if H is a graph with a gate G that is one of these configurations, then

$$E(H) \subseteq \bigcup_{C \in \mathcal{C}} P(C)$$

where C is the set of maximal connected components of G. All that is left to do in order to construct an extremal example is to pick which edges of each P(C) to delete in order to eliminate triples of vertices with more than one edge.

LEMMA 5.7. Let H be an I_0 -free graph on $n \ge 9$ vertices such that $n \equiv 0 \mod 3$, then

$$|E(H)| \ge n(n-3) + \frac{n}{3}$$

and there is exactly one extremal construction up to isomorphism.



FIGURE 22. C_3 plus an edge.

PROOF. We know from Lemma 5.4 that the only way H can possibly attain

$$n(n-3) + \frac{n}{3}$$

edges is if its gate is the disjoint union of $\frac{n}{3}$ directed triangles. Therefore, each $P(C_3)$ contains exactly one vertex triple with all three possible edges. So two of these must be deleted for each component in order to arrive at an extremal construction. The three choices for this deletion on each component are all isomorphic to each other. Therefore, there is exactly one extremal construction up to isomorphism.

LEMMA 5.8. Let H be an I_0 -free graph on $n \ge 9$ vertices such that $n \equiv 1 \mod 3$, then

$$|E(H)| \ge n(n-3) + \frac{n-4}{3}$$

and there are exactly 18 extremal constructions up to isomorphism.

PROOF. We know from Lemma 5.5 that if H has $n(n-3) + \frac{n-4}{3}$ edges, then its gate is the disjoint union of $\frac{n-4}{3}$ directed triangles with either a directed C_4 or a C_3 plus an edge on the remaining 4 vertices. As in the previous proof, there is only one choice up to isomorphism for which edges to delete from each $P(C_3)$. However, this will not be true of the last component on the remaining four vertices.

First, let's consider the case where the last component is a C_3 plus one edge. Call the vertices $\{x, y, z, a\}$ where $x \to y \to z \to x$ is the C_3 and $x \to a$ is the additional edge. First, note that we have the following three mutually exclusive choices for edges with head vertices in this component:

(1)
$$xa \in T_y$$
 or $xy \in T_a$



FIGURE 23. C_4 with 2 additional edges in opposite tail link graphs.

- (2) $za \in T_x$ or $xz \in T_a$, and
- (3) $zx \in T_y, yz \in T_x$, or $xy \in T_z$.

This gives 12 choices, and each choice is unique up to isomorphism.

Next consider the case of C_4 . Each 3-subset of these four vertices holds two edges of P(C) - one that points along the direction of the two gatekeeper edges and one that points the middle vertex of the two gatekeeper edges. For each triple one of these edges must be deleted to arrive at a legal oriented construction.

Each tail link graph must have at least n - 4 edges, and combined they must contain four additional edges. Since each can have up to two more edges, then the distribution of these additional edges must be one of the following integer partitions of 4:

- 2, 2, 0, 0
- 2, 1, 1, 0
- 1, 1, 1, 1

There is only one choice up to isomorphism with a distribution of 2, 2, 0, 0. Each of the three ways to place 2, 1, 1, 0 around C_4 are possible but each distribution has only one way up to isomorphism. Finally, there are two ways up to isomorphism to put an extra edge into each tail link graph. So all together there are six nonisomorphic ways to distribute these extra edges to the C_4 tail link graphs.

LEMMA 5.9. Let H be an I_0 -free graph on $n \ge 9$ vertices such that $n \equiv 2 \mod 3$, then

$$|E(H)| \ge n(n-3) + \frac{n-5}{3}$$

and there are exactly 32 extremal constructions up to isomorphism.

PROOF. We can do the same kind of analysis when n = 3k + 2 as in the previous proof. We know from Lemma 5.6 that the gate of any extremal construction must be all directed triangles together with either a directed C_5 or a directed triangle with a directed path of length two coming off of it (see Figure 21).

First, consider the C_5 case. Let the vertices be $\{x_0, \ldots, x_4\}$. For each gatekeeper edge, $x_i \to x_{i+1}$, every edge of the form $x_i v \to x_{i+1}$ must be an edge in H for any vertex

$$v \neq x_i, x_{i+1}, x_{i-1}, x_{i+2}.$$

Each gatekeeper edge can represent up to two additional edges of H, but again, every intersection of gatekeeper edges requires a mutually exclusive choice. Ultimately, we can add 5 additional edges so the extra edges must be distributed in one of the following ways:

- 2, 2, 1, 0, 0
- 2, 1, 1, 1, 0
- 1, 1, 1, 1, 1

There are 2 ways to get the first distribution up to isomorphism, 4 ways to get the second, and 2 ways to get the third. Therefore, there are 8 extremal constructions with this gate up to isomorphism.

Now consider the case of a directed triangle with a directed two path coming off of it. If we label the vertices as $\{x, y, z, a, b\}$ (see Figure 24), the mutually exclusive choices are

(1) $ax \to y \text{ or } yx \to a$, (2) $az \to x \text{ or } zx \to a$, (3) $zx \to y, yz \to x$, or $xy \to z$, and (4) $xa \to b \text{ or } bx \to a$



FIGURE 24. C_3 plus a 2-path.

This gives 24 ways of reaching the maximum, and each way is unique up to isomorphism. Therefore, there are 32 total distinct extremal graphs up to isomorphism. ■

This establishes the main result of this chapter.

CHAPTER 6

The Graph I_1

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

Let I_1 denote the forbidden graph where two edges intersect in exactly two vertices such that one vertex is the head for both edges and the other is in the tail set of each edge. That is $V(I_1) = \{a, b, c, d\}$ and $E(I_1) = \{ab \rightarrow c, ad \rightarrow c\}$ (see Figure 40).

THEOREM 6.1. For all $n \ge 4$,

$$ex(n, I_1) = ex_o(n, I_1) = n \left\lfloor \frac{n-1}{2} \right\rfloor$$

and there are

$$\left(\frac{(n-1)!}{2^{\left\lfloor\frac{n-1}{2}\right\rfloor} \left\lfloor\frac{n-1}{2}\right\rfloor !}\right)^n$$

labeled graphs that attain this maximum in the standard case.

PROOF. Let H be an I_1 -free graph on n vertices. For any $x \in V(H)$, T_x is a simple undirected 2-graph on n-1 vertices such that no two edges are adjacent (this is true for either version of the problem). Therefore, the edges of T_x are a matching on at most n-1 vertices. So there are at most $\lfloor \frac{n-1}{2} \rfloor$ edges in T_x for every $x \in V(H)$.



FIGURE 25. The graph I_1 .

Thus,

$$|E(H)| = \sum_{x \in V(H)} |T_x| \le n \left\lfloor \frac{n-1}{2} \right\rfloor.$$

This shows the upper bound for both versions.

Now we want to find lower bound constructions. In the standard version of the problem there are many extremal constructions since for each vertex x, we may pick any maximum matching on the remaining n-1 vertices to serve as the edges of T_x . So

$$\operatorname{ex}(n, I_1) = n \left\lfloor \frac{n-1}{2} \right\rfloor.$$

Moreover, the number of labeled graphs that attain this maximum equals the number of ways to take a maximum matching to construct each tail link graph. For even k, the number of matchings on k vertices is

$$M_k = (k-1)M_{k-2}$$

since if we fix some vertex, then we can pick any of the remaining k-1 vertices to go with it and then take the number of matchings on the remaining n-2. Since $M_2 = 1$, then in general for even k,

$$M_k = \prod_{i=1}^{\frac{\kappa}{2}} (2i-1).$$

If k is odd, then we can first select the vertex left out of the matching to get

$$M_k = kM_{k-1} = k \cdot \prod_{i=1}^{\frac{k-1}{2}} (2i-1) = \prod_{i=1}^{\frac{k+1}{2}} (2i-1)$$

Therefore, the number of labeled extremal I_1 -free graphs on n vertices is

$$\left(\prod_{i=1}^{\lfloor \frac{n}{2} \rfloor} (2i-1)\right)^n = \left(\frac{(n-1)!}{2^{\lfloor \frac{n-1}{2} \rfloor} \lfloor \frac{n-1}{2} \rfloor!}\right)^n.$$

In the oriented version of the problem we need to be more careful with the construction. First, assume that n is even and define a graph H with vertex set $V(H) = \mathbb{Z}_n$



FIGURE 26. T_i in the oriented extremal construction for even n.



FIGURE 27. T_i in the oriented extremal construction on n+1 vertices for even n.

and edge set

$$E(H) = \bigcup_{i=0}^{n-1} \left\{ (i+2k)(i+2k+1) \to i : k = 1, 2, \dots, \frac{n-2}{2} \right\}.$$

This construction creates a maximum matching for each tail link graph (with i + 1 as the odd vertex out for each T_i). So H has the extremal number of edges and contains no I_1 . Therefore, all we need to show is that it has no triple with more than one edge. If H does contain such a triple, then there exist three integers in \mathbb{Z}_n that can be represented as both $\{a, a+2k, a+2k+1\}$ and $\{b, b+2i, b+2i+1\}$ with $a \neq b$. Without



FIGURE 28. T_v in the oriented extremal construction on n+1 vertices for even n.

loss of generality we can assume that b = 0. If a + 2k = 0, then a + 2k + 1 = 1, but 1 is not in any tail set that points to 0. Therefore, it must be the case that a + 2k + 1 = 0, but then a + 2k = n - 1. Therefore, the set is equal to $\{0, n - 1, n - 2\}$, and a = n - 2, but n - 1 does not point to n - 2, a contradiction. Therefore, H can have no such triple.

Now, we consider odd n + 1. Here, let $V(H) = \mathbb{Z}_n \cup \{v\}$ where v is a new vertex and use all of the edges from the even construction plus some new ones that all contain v. So $E(H) = E_{even} \cup E_{new} \cup E_v$ where

$$E_{even} = \bigcup_{i=0}^{n-1} \left\{ (i+2k)(i+2k+1) \to i : k = 1, 2, \dots, \frac{n-2}{2} \right\},\$$

and

$$E_{new} = \{v(i+1) \to i : i = 0, 1, \dots, n-1\}.$$

Certainly, the construction has so far avoided the forbidden subgraph and given each of the first n vertices the maximum number of tails. Now E_v can be constructed as any set of $\frac{n}{2}$ disjoint pairs of elements from \mathbb{Z}_n all pointing at v so that no pair consists of two sequential numbers mod n. So any maximum matching of the n elements that observes this condition will do. In particular, we can let

$$E_v = \left\{ (i)(n-i) \to v : i = 1, \dots, \frac{n}{2} - 1 \right\} \cup \left\{ (0) \left(\frac{n}{2}\right) \to v \right\}.$$

 So

$$\operatorname{ex}_o(n, I_1) = n \left\lfloor \frac{n-1}{2} \right\rfloor.$$

CHAPTER 7

The Graph H_1

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

Let H_1 denote the forbidden graph where two edges intersect in exactly one vertex such that it is in the tail set of each edge. That is $V(H_1) = \{a, b, c, d, x\}$ and $E(H_1) = \{ax \rightarrow b, cx \rightarrow d\}$ (see Figure 29). First we will show the following result for the oriented version of the problem.

THEOREM 7.1. For all $n \ge 6$,

$$ex_o(n, H_1) = \left\lfloor \frac{n}{2} \right\rfloor (n-2).$$

We will use this result to solve the standard version of the problem.

THEOREM 7.2. For all $n \ge 8$,

$$ex(n,H_1) = \binom{n+1}{2} - 3.$$

Moreover, there is one unique extremal construction up to isomorphism for each n.

First, note that the proof of Theorem 7.1 is straightforward when n is even. To get a lower bound construction we can take a maximum matching on the n vertices and



FIGURE 29. The graph H_1 .



FIGURE 30. *H* has a copy of H_1 with intersection vertex *x* if and only if the directed link graph D_x has a pair of disjoint directed edges.

use each pair of this matching as the tail set to point at all n-2 other vertices. That is, let H be the graph with vertex set,

$$V(H) = \{x_1, \dots, x_{\frac{n}{2}}, y_1, \dots, y_{\frac{n}{2}}\}$$

and edge set,

$$E(H) = \bigcup_{i=1}^{\frac{n}{2}} \left\{ x_i y_i \to z : z \in V(H) \setminus \{x_i, y_i\} \right\}.$$

To show that this is also an upper bound, let H be an H_1 -free oriented graph on n vertices. Then for any $x \in V(H)$, the directed link graph D_x cannot have two independent edges (see Figure 30). Therefore, D_x is either empty, a triangle, or a star with at most n-2 edges. Since $n \geq 5$, then $|D_x| \leq n-2$ for each x. So

$$|E(H)| = \frac{1}{2} \sum_{x \in V(H)} |D_x| \le \frac{1}{2}n(n-2).$$

Hence, we are finished for even n. However, this proof falls apart when n is odd. We will need a different strategy.

1. Counting edges by possible tail pairs

The basis of our strategy in getting an upper bound on $ex_o(n, H_1)$ is to count the edges of an H_1 -free graph H by its tail sets. That is,

$$|E(H)| = \sum_{\{x,y\} \in \binom{V(H)}{2}} t(x,y)$$

It is simple but important to note that if H is H_1 -free, then any two pairs of vertices that each points to two or more other vertices must necessarily be disjoint.

LEMMA 7.1. Let H be a H₁-free oriented graph. If $x_1, x_2, y_1, y_2 \in V(H)$ so that

$$t(x_1, y_1), t(x_2, y_2) \ge 2$$

and $\{x_1, y_1\} \neq \{x_2, y_2\}$, then $\{x_1, y_1\} \cap \{x_2, y_2\} = \emptyset$.

PROOF. Suppose, towards a contradiction, that $x_1 = x_2 = x$ but $y_1 \neq y_2$. Since $t(x, y_1) \geq 2$, then there exists some vertex z_1 distinct from x, y_1 , and y_2 such that

$$xy_1 \to z_1 \in E(H).$$

Similarly, since $t(x, y_2) \ge 2$, then there exists some vertex z_2 distinct from x, y_1 , and y_2 such that

$$xy_2 \to z_2 \in E(H)$$

If $z_1 \neq z_2$, then this gives a copy of H_1 .

So assume that they are the same vertex, $z_1 = z_2 = z$. Since $t(x, y_1) \ge 2$, then there is some second vertex that x and y_1 point to that is distinct from z. The only choice that would not create a copy of H_1 with the edge $xy_2 \to z$ is y_2 . Similarly, since $t(x, y_2) \ge 2$, then there is some second vertex that x and y_2 point to that is distinct from z. The only choice that would not create a copy of H_1 with the edge $xy_1 \to z$ is y_1 . So

$$xy_1 \to y_2, xy_2 \to y_1 \in E(H)$$


FIGURE 31. An H_1 -free graph on n vertices breaks down into k disjoint pairs that each point to at least two other vertices plus a remainder set R with n - 2k vertices that belong to no such pair.

which contradicts the fact that H is oriented.

Therefore, if we assume that H is H_1 -free on n vertices, then we can split its vertices up into k disjoint pairs such that each serves as a tail set to at least two edges of Hplus a set of n - 2k vertices that belong to no such pair. That is,

$$V(H) = \{x_1, y_1, \dots, x_k, y_k\} \cup R$$

so that $t(x_i, y_i) \ge 2$ for i = 1, ..., k and $t(w, v) \le 1$ for all other vertex pairs, $\{w, v\}$ (see Figure 31).

We now have two cases to consider. Either there are no such pairs (k = 0) or there is at least one $(k \ge 1)$.

2. No pair points to more than one vertex

Assume that k = 0. Then $t(x, y) \leq 1$ for every pair $\{x, y\} \in \binom{V(H)}{2}$. If $|D_x| \leq n-3$ for all $x \in V(H)$, then

$$|E(H)| = \frac{1}{2} \sum_{x \in V(H)} |D_x| \le \frac{1}{2}n(n-3) < \frac{1}{2}(n-1)(n-2)$$



FIGURE 32. The special case configuration discussed in Lemma 7.2. Here, vertex x joins with every other element to point to vertex y.

and we are done. Otherwise, there exists some vertex x that belongs to n-2 tail sets. Therefore, D_x is a star of directed edges with some common vertex of intersection y. Either t(x, y) = 0 or t(x, y) = 1.

If t(x, y) = 0, then all of the n-2 directed edges of D_x must point to y (see Figure 32). Such a configuration in H limits the number of edges to $\binom{n-1}{2}$ as proven in Lemma 7.2. On the other hand, if t(x, y) = 1, then $xy \to z \in E(H)$ for some vertex z, and $xv \to y$ for all other vertices $v \neq x, y, z$. Such a configuration in H will limit the number of edges to $\binom{n-1}{2}$ as proven in Lemma 7.3.

LEMMA 7.2. Let H be an oriented graph on $n \ge 6$ vertices such that $t(x, y) \le 1$ for each pair $\{x, y\} \in \binom{V(H)}{2}$. If H is H₁-free and contains vertices x and y such that $xv \to y \in E(H)$ for each $v \in V(H) \setminus \{x, y\}$, then

$$|E(H)| \le \binom{n-1}{2}.$$

See Figure 32.

PROOF. We want to show that there can be no more than $\binom{n-2}{2}$ additional edges in *H* other than the n-2 edges described in the statement of the lemma. This would give an upper bound on the total number of edges of

$$\binom{n-2}{2} + (n-2) = \binom{n-1}{2}.$$

First, note that every triple of the form $\{x, y, v\}$ already holds an edge. This implies that any additional edge cannot contain both x and y since H is oriented. On the other hand, if we were to add an edge, $vw \to u$, that excluded both x and y completely, then this new edge would create a copy of H_1 with the existing edge, $vx \to y$. Therefore, every additional edge must be on a triple of the form $\{v, w, x\}$ or $\{v, w, y\}$.

However, x is already in the maximum number of tails. So given any pair of non- $\{x, y\}$ vertices, $\{v, w\}$, the only possible additional edges are

$$vw \to x, vw \to y, yv \to w, \text{and } yw \to v.$$

The last three all appear on the triple, $\{v, w, y\}$, and are therefore mutually exclusive choices when it comes to adding them to the graph. The first two are also mutually exclusive choices since $t(v, w) \leq 1$.

So assume, towards a contradiction, that we could add $\binom{n-2}{2} + 1$ more edges to the existing configuration. Then some pair $\{v, w\}$ of non- $\{x, y\}$ vertices must be used twice. Without loss of generality, this means we must add the edges $vw \to x$ and $yv \to w$.

Now, let u be any of the remaining n - 4 vertices. The possible edge $uv \to y$ would create a copy of H_1 with $vw \to x$, and the possible edge $uv \to x$ would create a copy of H_1 with $vy \to w$. Therefore, the pair $\{v, u\}$ cannot be a tail set for any edge.

We can also view the potential additional edges as two different types: those that have a tail set of two non- $\{x, y\}$ vertices and those that have a tail set that includes y. There were originally at most $\binom{n-2}{2}$ of the first type that we are allowed to add in total, one edge for every distinct pair. However, v can now no longer be in a tail set with any of the other n - 4 vertices. So there are now at most $\binom{n-2}{2} - (n-4)$ edges of this first type left possible to add. Therefore, in order to add $\binom{n-2}{2} + 1$ edges over all, we will need at least n - 3 of them to be of the second type - those that have yin the tail set. Note that x must be an isolated vertex in the directed link graph D_y . Hence, there are at most n-3 tails containing y since otherwise the directed graph D_y would have n-2 edges among n-2 vertices. In this case, D_y would have two independent directed edges and so H would have a copy of H_1 with y as its intersection vertex. Moreover, D_y must be a star with a single vertex of intersection. Since $v \to w \in E(D_y)$, then this vertex of intersection must either be v or w.

Hence, in order to add $\binom{n-2}{2} + 1$ edges, we will need to have $\binom{n-2}{2} - (n-4)$ edges that have non- $\{x, y\}$ tail sets. Since the tail set, $\{v, w\}$, already points to x, then this implies that all such edges must also point to x. Otherwise, we would have some edge of the form $ab \to y$. If a = w or b = w, then this would create a copy of H_1 with $vw \to x$. If both elements are distinct from w, then we would still need to point the pair wa either to x or to y. Either choice would create a copy of H_1 .

Let u be one of the remaining vertices. Then u must be adjacent to a directed edge of D_y for there to be n-3 edges added with y in the tail set. If v is the vertex of intersection of D_y , then this edge must either be $u \to v$ or $v \to u$. Either yields a copy of H_1 . Similarly, if w is the vertex of intersection of D_y , then either $wy \to u \in E(H)$ or $uy \to w \in E(H)$. Again, either of these yields a copy of H_1 . Therefore, $\binom{n-2}{2} + 1$ edges cannot be added to the existing configuration.

LEMMA 7.3. Let H be an oriented graph on $n \ge 6$ vertices such that for each pair $x, y \in V(H), t(x, y) \le 1$. If H is H_1 -free and contains vertices x, y, and z such that $xy \to z \in E(H)$ and $xv \to y \in E(H)$ for each $v \in V(H) \setminus \{x, y, z\}$ (see Figure 33), then

$$|E(H)| \le \binom{n-1}{2}.$$

PROOF. Let $W = \{1, 2, ..., n - 3\}$ be the set of non- $\{x, y, z\}$ vertices. Any additional edge to this graph must have a tail set of the form $\{i, j\}, \{i, y\}, \{i, z\}$, or $\{y, z\}$ for $i, j \in W$. An ij tail can only point to x or to y and there are $\binom{n-3}{2}$ pairs like this possible. An iy tail cannot point to x because H is oriented. It cannot point



FIGURE 33. The special case configuration discussed in Lemma 7.3. Here, x joins with every vertex except z to point to y and then joins with y to point to z.

to j since that would create a copy of H_1 with $xy \to z$. Therefore, it could only point to z. An *iz* tail could not point to any j since this would create a copy of H_1 with the edge $ix \to y$. Therefore, it could only point to y or to x. And a yz tail could not point to x since H is oriented. Therefore, it could only point to some i.

Assume, towards a contradiction, that we can add

$$\binom{n-2}{2} + 1 = \binom{n-3}{2} + (n-3) + 1$$

edges to the existing configuration. Since we can add at most $\binom{n-3}{2}$ edges with tail sets made entirely of vertices from W, then we must have at least n-2 additional edges from the other possibilities.

For each $i \in W$ we could have

$$iy \to z, yz \to i, iz \to y, \text{and } iz \to x.$$

The first three of these are mutually exclusive choices since they are all on the same triple. Similarly, the last two are mutually exclusive choices since we are only allowing up to one edge per possible tail set.

Therefore, in order to add n-2 of these types of edges, two must use the same element of W. Given the mutually exclusive choices above this implies that there is some vertex $i \in W$ such that either $iz \to x, yi \to z \in E(H)$ or $iz \to x, yz \to i \in E(H)$. In the first case, ij is no longer a possible tail for any edge for all n - 4 remaining vertices $j \in W$. This is because $iz \to x$, $yi \to z$, and $ix \to y$ create a triangle in D_i . So any additional edge with i in the tail would give two independent edges in D_i and therefore a copy of H_1 .

Hence, we can get at most $\binom{n-3}{2} - (n-4)$ edges with tails in W. This means that we will need 2(n-3) edges from the other possible edges to make up the difference if we want to add

$$\binom{n-3}{2} + (n-3) + 1$$

more edges.

Since each of the n-3 vertices from W can be in up to two of these additional edges, then $iz \to x$ would need to be an edge for every $i \in W$ and that $\{y, z, i\}$ also needs to hold one edge for every $i \in W$.

If $yz \to i$ is used once, then we get a copy of H_1 with $jz \to x$ for some other $j \in W$. Therefore, for all $i \in W$ we must have the edges $iy \to z$ and $iz \to x$. However, any pair $i, j \in W$ can now point to nothing since the only possibilities for such a tail were x or y to begin with and both of these options create copies of H_1 . So in this case the most that we can add is

$$2(n-3) \le \binom{n-3}{2} + (n-3)$$

for all $n \ge 6$.

In the other case we have added $iz \to x$ and $yz \to i$ for some i. Which means that $yz \to j$ is not allowed for any $j \neq i$ from W. Also, $jz \to y$ would make a copy of H_1 with $iz \to x$ and $jz \to x$ would make a copy of H_1 with $yz \to i$. Therefore, for all $j \neq i$ we can only add the edge $jy \to z$.

In order to add $\binom{n-3}{2} + n - 2$ edges, we will need all of these as well as all possible edges with tails in W. However, since $iz \to x$, all of the edges with tails completely in W must also point to x. Otherwise, some pair ab would point to y. If a = i or

b = i, then this would make a copy of H_1 with $iz \to x$. If $i \neq a, b$, then consider where the pair ai points. It must either point to x or to y, but either of these would create a copy of H_1 .

So all pairs of W must point to x and for all $j \in W$ not equal to i we must have the edge $jy \to z$. But $jy \to z$ and $ij \to x$ create a copy of H_1 , a contradiction. Hence, it is not possible to add more than $\binom{n-3}{2} + (n-3)$ edges to the configuration. Since the configuration already has n-2 edges, then there can be no more than $\binom{n-1}{2}$ edges total.

Together these two lemmas take care of the cases where all pairs of vertices point to at most one vertex in H.

3. Some pair of vertices is the tail set to multiple edges of H

We return to our description of an H_1 -free oriented graph as being made up of $k \ge 1$ vertex pairs that each serve as tail sets to strictly more than one edge plus a set R of the remaining n - 2k vertices,

$$V(H) = \{x_1, y_1, \dots, x_k, y_k\} \cup R$$

(see Figure 31). For each pair $\{x_i, y_i\}$ we want to prove the following upper bound,

$$t(x_i, y_i) + \sum_{v \neq x_i, y_i} (t(x_i, v) + t(y_i, v)) \le n - 2.$$

That is, the total number of edges that include either x_i or y_i or both in the tail set is at most n-2.

Now,

$$|E(H)| = \sum_{\{x,y\} \in \binom{V(H)}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + \sum_{v \neq x_i,y_i} \left(t(x_i,v) + t(y_i,v) \right) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + t(y_i,v) \right) + \sum_{\{x,y\} \in \binom{R}{2}} t(x,y) \le \sum_{i=1}^{k} \left(t(x_i,y_i) + t(y_i,v) \right) + \sum_{i=1}^$$

Note that each pair of vertices in R act as a tail set at most once so

$$\sum_{\{x,y\}\in\binom{R}{2}} t(x,y) \le \binom{n-2k}{2}.$$

Therefore, proving the upper bound for each $\{x_i, y_i\}$ pair would imply that

$$|E(H)| \le k(n-2) + \binom{n-2k}{2}$$

Since

$$k(n-2) + \binom{n-2k}{2} = 2k^2 - (n+1)k + \binom{n}{2}$$

is a quadratic polynomial with positive leading coefficient in terms of k, then it is maximized at the endpoints. Here, that means at k = 1 and at $k = \lfloor \frac{n}{2} \rfloor$.

When n is odd, both of these values for k give the upper bound,

$$|E(H)| \le \binom{n-1}{2}.$$

Only when n is even can we do better and get

$$|E(H)| \le \frac{n(n-2)}{2}$$

in the case where $k = \frac{n}{2}$. In either case this give an upper bound of

$$|E(H)| \le \left\lfloor \frac{n}{2} \right\rfloor (n-2).$$

So we need only prove that, in general,

$$t(x_i, y_i) + \sum_{v \neq x_i, y_i} (t(x_i, v) + t(y_i, v)) \le n - 2.$$

This is straightforward to show if $t(x_i, y_i) \ge 3$. However, when $t(x_i, y_i) = 2$ there is a case where it fails to hold. This is taken care of in the following lemma.



FIGURE 34. An H_1 -free graph containing this configuration with have at most $\binom{n-1}{2}$ edges as shown in Lemma 7.4.

LEMMA 7.4. Let H be an oriented graph on $n \ge 6$ vertices. If H is H₁-free and contains vertices x, y, a, and b such that $\{x, y\}$ is the tail set to exactly 2 edges with

$$xy \to a, xy \to b, yb \to a \in E(H),$$

and for each $v \in V(H) \setminus \{x, y, a, b\}$, $xv \to y$ (see Figure 34), then

$$|E(H)| \le \binom{n-1}{2}.$$

PROOF. First consider which pairs of vertices could possibly be a tail set to an edge in this graph. Let $W = \{1, \ldots, n-4\}$ be the set of vertices other than $\{x, y, a, b\}$. Then $\{i, j\}$ can be a tail set to $ij \to x$ and $ij \to y$ for any pair $i, j \in W$. Since $xy \to a$, then xi can point to nothing other than y. Similarly, xa and xb could only possibly point to b and a respectively, but either would create a copy of H_1 with $xi \to y$ for any $i \in W$. Also, by assumption xy can point to nothing else. Hence, x is in no additional tail sets.

Since $yb \to a$ and $xy \to a$, then ya cannot point to b or to x. It can also not point to any $i \in W$ since this would create a copy of H_1 with $xy \to b$. So y can be in no additional tails. The pair ab can point to anything aside from y since H is oriented, and ai can point to x or y for any $i \in W$ but not to b or another element of W since either would create a copy of H_1 with $xi \to y$. Similarly, bi can point to y for each copy of H_1 with either $yb \to a$ or $xi \to y$.

Leaving aside the edges with tail sets completely in W for the moment, this means there are 4(n-4) + 1 possible edges remaining. There are n-4 each of types $ai \to x$, $ai \to y, bi \to y$, and $ab \to i$ plus one extra edge which is $ab \to x$.

Suppose we are able to use at least 2(n-4) + 1 of these edges. First, if one of them is $ab \to x$, then there could be none of the types $ai \to y$ or $bi \to y$. So all of the ones of type $ab \to i$ and $ai \to x$ would need to be used. But since $n \ge 6$, there are at least two vertices in W. So there would exist edges $ai \to x$ and $ab \to j$ with $i \neq j$, a copy of H_1 . Therefore, $ab \to x$ cannot be used if we want to get more than 2(n-4) of these edges.

Hence, we need to use at least three types of edges from the four possible types. Since any of the types $ai \to x$, $ai \to y$, and $bi \to y$ eliminate the possibility of using any edge $ab \to j$ where $j \neq i$, then we can use at most one of this last type of edge. But since $n \ge 6$, then $2(n-4) + 1 \ge 5$ which means one of the other types gets used at least twice. Regardless of which one it is, there can be nothing used from the $ab \to i$ types of edges.

Therefore, we must use 2(n-4) + 1 edges from only the first three types. So there must be a vertex *i* from *W* that belongs to three of these edges, say

$$ai \to y, bi \to y, and ai \to x.$$

But the edges $bi \to y$ and $ai \to x$ form an H_1 , a contradiction. Thus, at most 2(n-4) of these kinds of edges can be used over all.

Now let us look at the edges with tail sets contained in W. We have seen that each ij can point to x or to y, but nothing so far has kept the pair from pointing to both. However, if some pair does point to both, then no other tail could use either of these vertices since this would create a copy of H_1 . Therefore, if there are $1 \leq l$ such pairs, then there are at most $2l + \binom{n-4-2l}{2}$ edges with tails from W. If n = 6, then this gives exactly one such pair and only 7 edges overall. If $n \ge 7$, then $l \le \frac{n-4}{2}$ implies that

$$2l \le n-4 \le \binom{n-4}{2} - \binom{n-4-2l}{2}.$$

Hence,

$$2l + \binom{n-4-2l}{2} \le \binom{n-4}{2}$$

So there are at most $\binom{n-1}{2}$ edges in H.

4. The oriented extremal number

Now we can proceed with establishing the upper bound under the assumption that the configuration presented in Lemma 7.4 does not occur in our directed hypergraph. As we've seen, all that is necessary to show is that

$$t(x_i, y_i) + \sum_{v \neq x_i, y_i} (t(x_i, v) + t(y_i, v)) \le n - 2$$

for any pair of vertices $\{x_i, y_i\}$ that serves as the tail set to at least two edges.

So let $\{x, y\}$ be such a pair, and divide the rest of the vertices of H into two groups, those that are a head vertex to some edge with xy as the tail and those that are not. That is,

$$V(H) \setminus \{x, y\} = \{h_1, \dots, h_m\} \cup \{n_1, \dots, n_t\}$$

where for each i = 1, ..., m, there exists an edge, $xy \to h_i \in E(H)$ and for each $j = 1, ..., t, xy \to n_j \notin E(H)$ (note that t(x, y) = m and that m + t = n - 2).

Now, consider an edge that contains either x or y in the tail but not both. Then the other tail vertex is either some h_i or some n_j . In the case of n_j , this edge must either be of the form $xn_j \to y$ or $yn_j \to x$ to avoid a copy of H_1 with both $xy \to h_1$ and $xy \rightarrow h_2$. Moreover, since H is oriented, there can be at most one. Hence,

$$\sum_{j=1}^{t} \left(t(x, n_j) + t(y, n_j) \right) \le t.$$

Now consider a tail set that includes either x or y and some h_i . Without loss of generality, assume that xh_1 is the tail to some edge. Since $t(x, y) \ge 2$, there is some other vertex h_2 such that $xy \to h_2 \in E(H)$. In order to avoid a copy of H_1 with this edge, xh_1 must either point to y or to h_2 . However, $xh_1 \to y \notin E(H)$ since this would give the triple $\{x, y, h_1\}$ more than one edge.

Therefore, $xh_1 \to h_2$ is the only option. However, if $t(x, y) \ge 3$, then this will create a copy of H_1 with $xy \to h_3$. So xh_i and yh_i cannot be tails to any edge. So

$$\sum_{i=1}^{m} \left(t(x, h_i) + t(y, h_i) \right) = 0.$$

Therefore,

$$\begin{aligned} t(x,y) &+ \sum_{v \neq x,y} \left(t(x,v) + t(y,v) \right) \\ &= m + \sum_{j=1}^{t} \left(t(x,n_j) + t(y,n_j) \right) + \sum_{i=1}^{m} \left(t(x,h_i) + t(y,h_i) \right) \\ &\leq m + t \\ &= n - 2 \end{aligned}$$

when $t(x, y) \ge 3$.

The only other possibility is that t(x, y) = 2. So suppose this is the case and that the head vertices to xy are a and b. Without loss of generality, assume that $yb \rightarrow a \in E(H)$. Note that this precludes any edges of the form $yn_j \rightarrow x$. Similarly, if we added the edge $xa \rightarrow b$ or the edge $xb \rightarrow a$, then we could not add any edges of the form $xn_j \to y$ and so

$$\sum_{j=1}^{t} \left(t(x, n_j) + t(y, n_j) \right) = 0$$

Moreover, $ya \to b$ would lead to more than one edge on the triple $\{y, a, b\}$. So

$$\sum_{i=1}^{m} (t(x, h_i) + t(y, h_i)) = 2$$

and in total we would have,

$$t(x,y) + \sum_{v \neq x,y} (t(x,v) + t(y,v)) = 4 \le n - 2.$$

On the other hand, if xa and xb are not tails to any edge, then the only way we could get a sum of more than n-2 is if $xn_j \to y \in E(H)$ for all $j = 1, \ldots, n-4$. But this is exactly the configuration described in Lemma 7.4 which we have excluded.

Therefore,

$$t(x,y) + \sum_{v \neq x,y} (t(x,v) + t(y,v)) \le n - 2$$

for any such pair, and this is enough to establish that

$$\exp(n, H_1) \le \left\lfloor \frac{n}{2} \right\rfloor (n-2)$$

Conversely, we have already considered an extremal construction in the case where n is even, and this same construction will work when n is odd. That is, take a maximum matching of the vertices (leaving one out) and use each matched pair as the tail set for all n - 2 possible edges.

Another construction that works for odd n that is not extremal for even n is to designate one vertex as the only head vertex and then make all $\binom{n-1}{2}$ pairs of the rest of the vertices tail sets.

Therefore,

$$\operatorname{ex}_o(n, H_1) = \left\lfloor \frac{n}{2} \right\rfloor (n-2).$$

Also, note that the only way that any construction could have more than $\binom{n-1}{2}$ edges is if *n* is even *and* the vertices are partitioned into $\frac{n}{2}$ pairs such that each points to at least two other vertices. This fact comes directly from the requirement that $k = \frac{n}{2}$ in the optimization of

$$k(n-2) + \binom{n-2k}{2}$$

in order for the expression to be more than $\binom{n-1}{2}$.

5. Intersections of multiedge triples in the standard version

Now, let H be an H_1 -free graph on n vertices under the standard version of the problem so that any triple of vertices can now have up to all three possible directed edges. If we let t_H be the number of triples of vertices of H that hold at least one edge, and we let m_H be the number of triples that hold at least two, then we have the following simple observation:

$$|E(H)| \le t_H + 2m_H.$$

We start our path towards an upper bound on |E(H)| by finding an upper bound on the number of multiedge triples, m_H . We will need to prove some facts about the multiedge triples of H. First, any triple which holds two edges of H might as well hold three.

LEMMA 7.5. Let H be an H_1 -free graph such that some triple of vertices $\{x, y, z\}$ contains two edges. Define H' by V(H') = V(H) and

$$E(H') = E(H) \cup \{xy \to z, xz \to y, yz \to x\}.$$

Then H' is also H_1 -free.

PROOF. Suppose H' is not H_1 -free. Since H is H_1 -free and the two graphs differ by at most one edge, then they must differ by exactly one edge. Without loss of generality, say

$$\{xy \to z\} = E(H') \setminus E(H).$$

This edge must be responsible for creating the copy of H_1 in H'. So it must intersect another edge in exactly one vertex that is in the tail set of both.

Therefore, without loss of generality, there is an edge $xt \to s \in H$ where $\{s, t\} \cap \{y, z\} = \emptyset$. However, since $\{x, y, z\}$ already contained two edges of H, then $xz \to y \in H$. Since $xt \to s$ and $xz \to y$ make a copy of H_1 , then H cannot be H_1 -free, a contradiction.

Next, we want to show that no two multiedge triples can intersect in exactly one vertex.

LEMMA 7.6. Let H be a H_1 -free graph. If two vertex triples $\{x, y, z\}$ and $\{s, t, r\}$ each contain two or more edges of H, then

$$|\{x, y, z\} \cap \{s, t, r\}| \neq 1.$$

PROOF. Suppose

$$|\{x, y, z\} \cap \{s, t, r\}| = 1$$

By Lemma 7.5, since H is H_1 -free, the graph created from H by adding all three possible edges on the triples $\{x, y, z\}$ and $\{s, t, r\}$ is also H_1 -free. But if x = r and x, y, z, s, and t are all distinct, then this graph contains $xy \to z$ and $xs \to t$ which is a copy of H_1 , a contradiction.

Therefore, we can use an upper bound on the number of undirected 3-uniform hyperedges such that no two intersect in exactly one vertex as an upper bound on the number of multiedge triples. Moreover, the extremal examples are easy to describe which will be important for finding the upper bound for $ex(n, H_1)$ as well as for establishing the uniqueness of the lower bound construction.

LEMMA 7.7. Let H be a 3-uniform undirected hypergraph on n vertices such that no two edges intersect in exactly one vertex, then

$$|E(H)| \le \begin{cases} n & n \equiv 0 \mod 4 \\ n-1 & n \equiv 1 \mod 4 \\ n-2 & n \equiv 2, 3 \mod 4 \end{cases}$$

and H is the disjoint union of $K_4^{(3)}s$, $K_4^{(3)}s$ minus an edge (K_4^-) , and sets of edges that all share a common intersection of two vertices - a sunflower with a two vertex core.

PROOF. Two edges of H are either disjoint or they intersect in two vertices. So connected components of H that have 1 or 2 edges are both sunflowers. A third edge can be added to a two-edge sunflower by either using the two common vertices to overlap with both edges in two or by using one common vertex and the two petal vertices. So a connected component of H with 3 edges is either a sunflower or a K_4^- . The only way to connect a fourth edge to the three-edge sunflower is to make a fouredge sunflower, and this is true for a k-edge sunflower to a (k+1)-edge sunflower for all $k \geq 3$. The only way to add a fourth edge to the K_4^- is to make a $K_4^{(3)}$ and then no new edges may be connected to a $K_4^{(3)}$ without intersecting two of its edges in exactly one vertex each. Therefore, these are the only possible connected components of H. A sunflower with k edges uses k + 2 vertices, and a $K_4^{(3)}$ has four edges on 4 vertices. Therefore, if $n \equiv 0 \mod 4$ we can get at most n edges with a disjoint collection of $K_4^{(3)}$ s. Similarly, the best we can do when $n \equiv 1 \mod 4$ is n-1 edges with a disjoint collection of $K_4^{(3)}$ s plus one isolated vertex since any sunflower will automatically limit the number of edges to n-2. And if $n \equiv 2 \mod 4$ or $n \equiv 3 \mod 4$, then n-2 is the best that we can do.



FIGURE 35. An edge that intersects a $K_4^{(3)}$ block of multiedge triples in one or two tail vertices will create a copy of H_1 .

In general, the only way to actually have an H_1 -free graph with n multiedge triples is if the multiedge triples form an undirected 3-uniform hypergraph of $\frac{n}{4}$ disjoint $K_4^{(3)}$ blocks when $n \equiv 0 \mod 4$.

In this case there can be no additional directed edges in H since such an edge would either intersect one of these $K_4^{(3)}$ s in one tail vertex which would create a copy of H_1 since this means it intersects three of the multiedge triples in exactly one tail vertex (we may assume that each multiedge has all three edges per Lemma 7.5) or it would intersect one of the $K_4^{(3)}$ s in two tail vertices which means that it intersects two of the multiedge triples in exactly one tail vertex (see Figure 35).

So in this case, the number of total edges would be bound by

$$3n < \binom{n+1}{2} - 3$$

for all $n \geq 7$.

Next, the only way to have n - 1 multiedge triples is to either have $\frac{n-1}{4}$ disjoint $K_4^{(3)}$ blocks when $n \equiv 1 \mod 4$ or to have $\frac{n}{4} - 1$ disjoint $K_4^{(3)}$ blocks with one K_4^- when $n \equiv 0 \mod 4$. In the first case any additional edge must have at least one and perhaps two of its tail vertices in a single $K_4^{(3)}$ block of multiedge triples which we have already seen will create a copy of H_1 . So there are at most

$$3(n-1) < 3n < \binom{n+1}{2} - 3$$

total edges in this case.

In the second case, any additional edge that has no tail vertices in a $K_4^{(3)}$ block must have both tail vertices in the K_4^- . If the head to such an edge were outside of the K_4^- , then the edge must intersect one of the three multiedge triples of the block in exactly one tail vertex since there are two triples that it intersects in one tail vertex each, one of which must be a multiedge triple. On the other hand, it could have its head vertex inside the K_4^- . In this case, the additional edge must lie on the triple without multiple edges. This is the only edge that can be added. So there are at most

$$3(n-1) + 1 < 3n < \binom{n+1}{2} - 3$$

total edges in this case.

6. An H_1 -free graph with n-2 multiedge triples

Now, the only ways to have exactly n-2 multiedge triples is either to have $\frac{n}{4}-2$ of the $K_4^{(3)}$ blocks plus two K_4^- blocks of multiedge triples when $n \equiv 0 \mod 4$ or to have k of the $K_4^{(3)}$ blocks of multiedge triples plus a sunflower with n-4k-2 petals. The first case is suboptimal for the same reasons already considered. So let us consider the second case.

First, assume that k = 0 and that we have n - 2 multiedge triples that make a sunflower (see Figure 36). How many edges can we add? This structure already has all possible edges with 2 vertices in the core (or so we may assume by Lemma 7.5). On the other hand, if an additional edge has no vertices in the core, then it would intersect two multiedge triples in one tail vertex each which would create a copy of H_1 .

Therefore, any additional edge must include exactly one vertex from the core. If this vertex is in the tail set to the additional edge and the sunflower has at least three petals, then the additional edge intersects in exactly one tail vertex of the multiedge triples of the sunflower, a contradiction. Since we assume that $n \ge 6$, then the



FIGURE 36. The unique extremal construction for an H_1 -free graph has $\binom{n-2}{2} + 3(n-2)$ edges.

sunflower has at least three petals. Hence, any additional edge must intersect the core in only its head vertex.

If any two additional edges have different core vertices as the head, then either the tail sets of these edges must be exactly the same or completely disjoint to avoid a copy of H_1 . Hence, pairs of petal vertices that point to both core vertices must be independent of all other tail sets. And all other petal vertices fall into disjoint sets as to whether they are in additional edges that point to the first core vertex or the second. The number of additional edges will be maximized if every pair of petal vertices point to the same core vertex. Moreover, this will give a total of

$$3(n-2) + \binom{n-2}{2} = \binom{n+1}{2} - 3$$

edges.

We will soon see that this is the best that we can do and that this construction, where the multiedge triples make a sunflower with n-2 petals with $\binom{n-2}{2}$ additional edges pointing from pairs of petal vertices to a single core vertex, is unique up to isomorphism.

First we will need to see that k = 0 is the number of $K_4^{(3)}$ multiedge triple blocks that optimizes the total number of edges. So suppose there are k such blocks and that the other n - 4k vertices are in a sunflower. Then from prior considerations we know that any additional edge must have both tail vertices in this sunflower. If one of these tail vertices coincides with a petal vertex of the sunflower, then there will be a copy of H_1 . Therefore, the tail vertices must coincide with the core and the only possibility for such an edge is to point out to a vertex in one of the k blocks.

Therefore, there are at most

$$3(4k) + 3(n - 4k - 2) + \binom{n - 4k - 2}{2} + 4k$$

edges in such a construction. Since this expression is quadratic in k with positive leading coefficient, then it must maximize at the endpoints, k = 0 or $k = \frac{n}{4}$, and we already know that $k = \frac{n}{4}$ is suboptimal. Therefore, if there are exactly n - 2multiedge triples, then they must form a sunflower with a two-vertex core and from there the only way to maximize the total number of edges is to add every possible edge with tail set among the petal vertices all pointing to the same head vertex in the core.

7. Fewer than n-2 multiedge triples

Now suppose that H has fewer than n-2 multiedge triples. If $t_H \leq \binom{n-1}{2}$, then

$$|E(H)| \le t_H + 2m_H < \binom{n-1}{2} + 2(n-2) = \binom{n+1}{2} - 3.$$

So we must assume that $t_H > \binom{n-1}{2}$. Also, if $m_H = 0$, then we know that

$$|E(H)| \le \exp_o(n, H_1) = \left\lfloor \frac{n}{2} \right\rfloor (n-2) < \binom{n+1}{2} - 3.$$

So assume that there is at least one multiedge triple, $\{x, y, z\}$. This triple has at least two edges. Assume without loss of generality that they are $xy \to z$ and $xz \to y$.

Let H' be an oriented graph arrived at by deleting edges from multiedge triples of H until each triple has at most one edge and every triple that had at least one edge in H still has at least one in H'. In other words, H' is any subgraph of H such that $t_{H'} = t_H$ and $m_{H'} = 0$. Without loss of generality, assume that

$$xy \to z \in E(H').$$

Since $t_{H'} > \binom{n-1}{2}$, then *n* must be even. Moreover, there is a matching on the vertices so that every matched pair $\{a, b\}$ points to at least two other vertices. That is, $t(a, b) \ge 2$.

Now consider the directed link graphs of the vertices. As stated before, these are either triangles or stars with a common vertex. However, if two or more of these link digraphs have three or fewer edges each (for instance, if they are triangles), then there are fewer edges than we are assuming since

$$|E(H')| = \frac{1}{2} \sum_{x \in V(H')} |D_x| \le \frac{1}{2} \left(6 + (n-3)(n-2) \right) < \binom{n-1}{2}$$

for all $n \ge 8$. We will show that it must be the case that here at least two directed link graphs are restricted to at most three directed edges each, contradicting our assumptions about the number of edges in H.

First, note that $x \to z \in D_y$ and $y \to z \in D_x$. To avoid a contradiction, at least one of these two directed link graphs must have four or more edges. Without loss of generality, assume that it is D_y . Therefore, D_y is a star and not a triangle. So the additional three directed edges in D_y must either all be incident to z or to x.

If these directed edges are all incident to z, then y and z must be partners under the matching which means that x has another partner x' distinct from y and z. Since $t(x, x') \ge 2$ in H', then x' must point to two vertices in D_x . Since D_x already has $y \to z$ and no two edges may be independent in any directed link graph, then x' must point to y and to z, forming a triangle.

Next, consider $D_{x'}$. We know that

$$x \to y, x \to z \in D_{x'}$$

If there is an additional edge in $D_{x'}$ that does not complete this triangle then it is either of the form $x \to t$ or $t \to x$. If $x \to t \in D_{x'}$ then $x' \to t, y \to z \in D_x$, a contradiction. If $t \to x \in D_{x'}$, then $x' \to x \in D_t$. But since t has its own matched vertex, then there exists a distinct t' such that

$$t' \to x, t' \to x' \in D_{t'}$$

So either $|D_{x'}| \leq 3$ or $|D_{t'}| \leq 3$. Either way, this gives us two directed link graphs that have at most three edges each. So $t_{H'} < \binom{n-1}{2}$.

Therefore, we must assume that the three additional edges in D_y are incident to xand that y and x are partners under the matching. So z has some other partner under the matching z' distinct from x and y. Now, delete the edge $xy \to z$ from H' and add $xz \to y$ to get a new directed hypergraph H''. It follows that H'' has no multiedge triples and is H_1 -free since we still have a subgraph of H.

In adding $xz \to y$ we have added $x \to y$ to D_z . Since z' must point to two vertices in D_z , then this addition means that D_z is a triangle under H''. Hence, $|D_z| = 2$ under H'.

Now, the same argument as above applies to $D_{z'}$. The only way for $|D_{z'}| > 3$ would mean either $z \to a \in D_{z'}$ or $a \to z \in D_{z'}$ for some a distinct from x, y, z, and z'. The first case would mean that two independent directed edges, $z' \to a$ and $x \to y$ are in D_z , a contradiction. The second case would mean that $z' \to z \in D_a$. Since a has its own partner under the matching that must point to two vertices in D_a , then in this case, D_a is a triangle.

Therefore, $t_H > \binom{n-1}{2}$ and $m_H \ge 1$ cannot both be true in any H_1 -free graph. This is enough to complete the result,

$$\operatorname{ex}(n, H_1) = \binom{n+1}{2} - 3$$

This also exhausts the remaining cases in order to demonstrate that the extremal construction is unique.

CHAPTER 8

The Graph H_2

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

Let H_2 denote the forbidden graph where two edges intersect in exactly two vertices such that the set of intersection is the tail set to each edge. That is $V(H_2) = \{a, b, c, d\}$ and $E(H_2) = \{ab \rightarrow c, ab \rightarrow d\}$ (see Figure 37).

THEOREM 8.1. For all $n \geq 5$,

$$ex(n, H_2) = ex_o(n, H_2) = \binom{n}{2}$$

Moreover, there are $(n-2)^{\binom{n}{2}}$ different labeled H_2 -free graphs attaining this extremal number when in the standard version of the problem.

PROOF. Let H be H_2 -free. Regardless of which version of the problem we are considering, each pair of vertices acts as the tail set to at most one directed edge. Therefore,

$$\operatorname{ex}(n, H_2), \operatorname{ex}_o(n, H_2) \le \binom{n}{2}.$$



FIGURE 37. The graph H_2 .



FIGURE 38. Inductive construction of H_2 -free oriented graphs.

In the standard version of the problem any function, $f : {[n] \choose 2} \to [n]$, that sends each pair of vertices to a distinct third vertex, $f(\{a, b\}) \notin \{a, b\}$, has an associated H_2 -free construction H_f with ${n \choose 2}$ edges. That is, for any such function, f, let $V(H_f) = [n]$ and

$$E(H_f) = \left\{ a, b \to f(\{a, b\}) : \{a, b\} \in \binom{[n]}{2} \right\}.$$

Since each pair of vertices acts as the tail set to exactly one directed edge, then H_f is H_2 -free and has $\binom{n}{2}$ edges. So

$$\operatorname{ex}(n, H_2) = \binom{n}{2}.$$

Moreover, there are $(n-2)^{\binom{n}{2}}$ distinct functions from $\binom{[n]}{2}$ to [n] such that no pair is mapped to one of its members. Therefore, there are $(n-2)^{\binom{n}{2}}$ labeled graphs that are H_2 -free with $\binom{n}{2}$ edges.

In the oriented version of the problem lower bound constructions can be defined inductively on n.

First, let n = 5 and define G_5 as the oriented graph with vertex set

$$V(G_5) = \{0, 1, 2, 3, 4\}$$

and the following edges: $0, 1 \rightarrow 2; 1, 3 \rightarrow 0; 0, 4 \rightarrow 1; 0, 2 \rightarrow 3; 2, 4 \rightarrow 0; 0, 3 \rightarrow 4; 2, 3 \rightarrow 1; 1, 2 \rightarrow 4; 1, 4 \rightarrow 3; and 3, 4 \rightarrow 2.$

Each pair of vertices of G_5 are in exactly one tail set, and each triple of vertices appear together in exactly one edge. Therefore, this construction is H_2 -free with $\binom{5}{2}$ edges.

Now, let $n \ge 5$, and define G_{n+1} by $V(G_{n+1}) = \{0, 1, \ldots, n\}$ and

$$E(G_{n+1}) = E(G_n) \cup \{ni \to (i+1) : i = 0, \dots, n-1\}$$

where addition is taken modulo n.

Then G_{n+1} has *n* more edges than G_n . So $|E(G_{n+1})| = \binom{n+1}{2}$.

Any two new edges intersect in at most two vertices. Similarly, any new edge and any old edge also intersect in at most two vertices. Hence, at most one edge appears on a given triple of vertices. So G_{n+1} is oriented.

Moreover, all tail sets for the new edges are distinct from each other and from any tail sets for the edges of G_n . So G_{n+1} is H_2 -free. Therefore,

$$\operatorname{ex}_o(n, H_2) = \binom{n}{2}.$$

CHAPTER 9

Generalized Directed Hypergraphs

In this chapter we will generalize the standard version of a $2 \rightarrow 1$ directed hypergraph to a class of combinatorial structures that can all be considered to be uniform directed hypergraphs. This chapter is organized as follows. In Section [], we define the class of generalized directed hypergraphs (GDH) and extend the concepts of Turán density, blowups, and supersaturation to this setting. In Section [2], we define the idea of a jump for a given model of directed hypergraphs and prove several results about these jumps and how the jumps from one instance of the class relate to jumps in another. In Section [3], we adapt a couple of results proved in [7] for totally directed hypergraphs with multiplicity to any GDH.

1. Basic definitions and results

The following definition for a generalized directed hypergraph is intended to include most uniform models that could reasonably be called uniform directed hypergraphs. This includes models where the edges are r-sets each under some partition into k parts of fixed sizes r_1, \ldots, r_k with some linear ordering on the k parts. The definition only includes structures where an r-set could include multiple edges up to the number of possible orientations allowed. That is, we do not consider the "oriented" versions of the models where only one edge is allowed per r-set. The definition is given in terms of logic and model theory for convenience only. No deep results from those subfields are used. The use of this notation also makes further generalizations like nonuniform directed hypergraphs or oriented directed hypergraphs easy. DEFINITION 9.1. Let $\mathcal{L} = \{E\}$, a language with one r-ary relation symbol E. Let T be an \mathcal{L} -theory that consists of a single sentence of the form

$$\forall x_1 \cdots x_r E(x_1, \dots, x_r) \implies \bigwedge_{i \neq j} x_i \neq x_j \land \bigwedge_{\pi \in J_T} E(x_{\pi(1)}, \dots, x_{\pi(r)})$$

for some subgroup of the group of permutations on r elements, $J_T \subseteq S_r$. Call such a theory a generalized directed hypergraph theory and any finite model of T is a generalized directed hypergraph (GDH).

Note that this definition includes graphs, hypergraphs, and $r \to 1$ directed hypergraphs. For example, the theory for a $2 \to 1$ directed hypergraph is

$$T = \{ \forall xyz E(x, y, z) \implies x \neq y \land x \neq z \land y \neq z \land E(y, x, z) \}.$$

It is easy to see that when r = 2 we have only two GDH theories. The theory associated with the group S_2 is the theory of graphs, and the theory associated with the trivial group is the theory of directed graphs.

When r = 3 there are six subgroups of S_3 . Three of these are all isomorphic to \mathbb{Z}_2 with each generated by a permutation that swaps two elements. The corresponding GDH theory for any of these can be thought of as having pointed 3-sets for edges or as being $(2 \rightarrow 1)$ -graphs. Of the other subgroups, S_3 itself gives the theory of undirected 3-uniform hypergraphs, the trivial group gives totally directed 3-edges, and the subgroup generated by a three-cycle isomorphic to \mathbb{Z}_3 yields a GDH theory where the edges can be thought of as 3-sets that have some kind of cyclic orientation - either clockwise or counter-clockwise. Figure 39 summarizes the models of GDHs when r = 3. Note that in general, S_r always corresponds to the normal undirected rgraph model and the trivial group always corresponds to totally directed hypergraphs.

A fun thought experiment is to consider the kinds of edges that arise when r = 4. Many of them are geometric in nature. For instance, the alternating group A_4 gives a theory where edges can be thought of tetrahedrons (at least in an abstract sense).



FIGURE 39. The subgroup lattice of S_3 and the corresponding lattice of directed hypergraphs.

In fact, Leader and Tan [33] study the "oriented" versions of the models that come from the alternating groups for any $r \geq 3$.

In this chapter, when the theory is not specified we are simply discussing GDHs that are all models of the same fixed theory. When discussing multiple theories we will often refer to T-graphs to mean models of a GDH theory T. Throughout the chapter, J_T will always stand for the subgroup $J_T \subseteq S_r$ that determines the GDH theory Tand m_T will always be the order of this subgroup, $m_T = |J_T|$. Also, V_G and E_G will be used to denote the underlying set of elements of a model G and its relation set respectively.

The following basic propositions are given without proof. The first is a simple consequence that we are working in a relational language, and the second results from the fact that J_T is a group.

PROPOSITION 9.2. For any GDH theory T and any nonnegative integer n, there exists a GDH $G \models T$ on n elements. Moreover, for any nonnegative integer k < n, the substructure of G induced on any k-subset of the elements of G is also a T-graph.

PROPOSITION 9.3. Given a GDH G with r-ary relation set E_G , there exists an equivalence relation ~ on E_G defined by

$$(a_1,\ldots,a_r)\sim(b_1,\ldots,b_r)$$

if and only if for each $i, b_i = a_{\pi(i)}$ for some $\pi \in J_T$.

We can now use these propositions to extend the concepts of extremal graph theory to GDHs in a natural way.

DEFINITION 9.4. For any GDHG, an edge of G will always refer to an equivalence class of $[E_G]_{\sim}$.

DEFINITION 9.5. Given a GDH G on n elements, denote the number of edges of G by $e_T(G)$ and let the edge density of G be defined as

$$d_T(G) := \frac{e_T(G)}{\frac{r!}{m_T} \binom{n}{r}}.$$

Note that since

$$e_T(G) = \frac{|E_G|}{m_T},$$

then the density is

$$d_T(G) = \frac{(n-r)!|E_G|}{n!}$$

and could have been defined this way while mostly avoiding talk of edges as equivalence classes of E_G . However, the above definition makes the following extremal concepts reduce to their standard definitions in the undirected case.

DEFINITION 9.6. Given two GDHs G and H and a function $\psi: V_H \to V_G$, we say that ψ is a homomorphism if for all $(a_1, \ldots, a_r) \in E_H$, $(\psi(a_1), \ldots, \psi(a_r)) \in E_G$.

We say that G contains a copy of H if there exists some injective homomorphism, $\psi: V_H \to V_G$. Otherwise, we say that G is H-free. Similarly, we would say that a GDH G is \mathcal{F} -free for some family \mathcal{F} of GDHs if G is F-free for all $F \in \mathcal{F}$.

DEFINITION 9.7. Given a family of GDHs \mathcal{F} and a positive integer n, let the nth extremal number, $ex_T(n, \mathcal{F})$, be defined as the maximum number of edges over all \mathcal{F} -free GDHs on n elements,

$$ex_T(n, \mathcal{F}) := \max_{\mathcal{F}\text{-free } G_n} \{e_T(G_n)\}.$$

The Turán density of \mathcal{F} is defined as

$$\pi_T(\mathcal{F}) := \lim_{n \to \infty} \frac{ex_T(n, \mathcal{F})}{\frac{r!}{m_T} \binom{n}{r}}.$$

Our first main result is to show that these Turán densities exist for any GDH theory. The proof is the standard averaging argument used to show that these limiting densities exist for families of undirected hypergraphs [29].

THEOREM 9.1. For any GDH family \mathcal{F} the Turán density exists.

PROOF. Let G be an \mathcal{F} -free GDH on n elements with $ex_T(n, \mathcal{F})$ edges. For each $i = 1, \ldots, n$ let G^i be the subGDH of G induced by removing the *i*th vertex. Each edge of G appears in exactly n - r of these subGDHs. Therefore,

$$(n-r)e_T(G) = e_T(G^1) + \cdots + e_T(G^n).$$

Moreover, $e_T(G) = e_T(n, \mathcal{F})$ and each G^i is also \mathcal{F} -free so $e_T(G^i) \leq e_T(n-1, \mathcal{F})$. Therefore,

$$\operatorname{ex}_T(n,G) \le \frac{n}{n-r} \operatorname{ex}_T(n-1,\mathcal{F}).$$

 So

$$\frac{\operatorname{ex}_T(n,G)}{\frac{r!}{m_T}\binom{n}{r}} \leq \frac{n}{n-r} \frac{\operatorname{ex}_T(n-1,\mathcal{F})}{\frac{r!}{m_T}\binom{n}{r}} = \frac{\operatorname{ex}_T(n-1,\mathcal{F})}{\frac{r!}{m_T}\binom{n-1}{r}}.$$

Therefore, the sequence of these extremal densities is monotone decreasing as a function of n in the range [0, 1]. Hence, the limit exists.

1.1. Blowups and blowup density. We'll now extend the concept of the blowup of uniform hypergraphs to the more general setting of GDHs and define the corresponding notion of the blowup density. As with hypergraphs, the blowup of a GDH can be thought of as the replacement of each vertex with many copies and taking all of the resulting edges. Formally,

DEFINITION 9.8. Let G be a GDH with $V_G = \{x_1, \ldots, x_n\}$, and let $t = (t_1, \ldots, t_n)$ be a tuple of positive integers. Define the t-blowup of G to be the \mathcal{L} -structure G(t) where

$$V_{G(t)} = \{x_{11}, \dots, x_{1t_1}, \dots, x_{n1}, \dots, x_{nt_n}\}$$

and

$$(x_{i_1j_1},\ldots,x_{i_rj_r})\in E_{G(t)}\iff (x_{i_1},\ldots,x_{i_r})\in E_G.$$

PROPOSITION 9.9. Let G be a GDH on n vertices, and let $t = (t_1, \ldots, t_n)$ be a tuple of positive integers. Then the t-blowup of G is also a GDH.

PROOF. We need only show that the \mathcal{L} -structure G(t) models T. So let

$$(x_{i_1j_1},\ldots,x_{i_rj_r})\in E_{G(t)}.$$

Then $(x_{i_1}, \ldots, x_{i_r}) \in E_G$. Since $G \models T$ this implies that $i_a \neq i_b$ whenever $a \neq b$. Hence, the elements $x_{i_a j_a} \neq x_{i_b j_b}$ whenever $a \neq b$. It also implies that $(x_{i_{\pi(1)}}, \ldots, x_{i_{\pi(r)}}) \in E_G$ for any $\pi \in J_T$. Hence,

$$(x_{i_{\pi(1)}j_{\pi(1)}},\ldots,x_{i_{\pi(r)}j_{\pi(r)}}) \in E_{G(t)}$$

for any $\pi \in J_T$. Therefore, $G(t) \models T$.

Next, we consider the edge density of a given blowup by defining the edge polynomial for a GDH.

DEFINITION 9.10. Let G be a GDH on n vertices. For each r-set $R \in \binom{V_G}{r}$, let e_R be the number of edges of G in R. Then let the edge polynomial be

$$p_G(x) := \sum_{R \in \binom{V_G}{r}} e_R \prod_{i \in R} x_i.$$

This polynomial is a simple generalization of the standard edge polynomial for undirected hypergraphs. To see this more easily note that for a given GDH G, the edges of G are in bijection with the monomials the sum p_G were we to write the sum out with no coefficients greater than one.

From this we see that the edge density of the (t_1, \ldots, t_n) -blowup of G is

$$\frac{p_G(t_1,\dots,t_n)}{\frac{r!}{m_T}\binom{t}{r}} = m_T \frac{p_G(t_1,\dots,t_n)}{t(t-1)\cdots(t-r+1)}$$

where $t = \sum t_i$. Let t increase to infinity and for each t pick a vector (t_1, \ldots, t_n) that maximizes this edge density. Then this sequence of densities is asymptotically equivalent to the sequence of numbers

$$m_T p_G\left(\frac{t_1}{t},\ldots,\frac{t_n}{t}\right).$$

This motivates the following definition.

DEFINITION 9.11. Let G be a GDH on n vertices. Let

$$S^{n} = \left\{ (x_{1}, \dots, x_{n}) | x_{i} \ge 0 \land \sum_{i=1}^{n} x_{i} = 1 \right\},$$

the standard (n-1)-dimensional simplex. Define the blowup density of G as

$$b_T(G) = m_T \max_{x \in S^n} \{ p_G(x) \}.$$

Since any $x \in S^n$ is the limit of some sequence $\left\{ \left(\frac{t_1}{t}, \ldots, \frac{t_n}{t}\right) \right\}$ with positive t_i as $t \to \infty$, then the blowup density of a GDH G is the best limiting density of any sequence of blowups of G.

The remaining definition and basic result about blowups given in this subsection will be useful when extending results about jumps and nonjumps from undirected hypergraphs to GDHs generally in Section 3.

DEFINITION 9.12. Let T' and T be GDH theories such that $J_{T'} \subseteq J_T \subseteq S_r$. For a T-graph F and a T'-graph F' we say that F contains F' if $V_F = V_{F'}$ and every edge of F' is contained in some edge of F (where the edges are considered under their equivalence class definition as subsets of E_F and $E_{F'}$). We say that F is the minimum T-container of F' if F has no edges that do not contain edges of F'. PROPOSITION 9.13. Let T' and T be GDH theories such that $J_{T'} \subseteq J_T \subseteq S_r$. Let F' be a T'-graph and let F be the minimum T-container of F'. Then

$$\frac{m_{T'}}{m_T}b_T(F) \le b_{T'}(F') \le b_T(F)$$

with equality on the left if F' has exactly one edge contained in each edge of F and equality on the right if each edge of F contains all $\frac{m_T}{m_{T'}}$ possible edges of F'.

Moreover, if F' has exactly k edges contained in each edge of F, then

$$b_{T'}(F') = \frac{km_{T'}}{m_T} b_T(F).$$

PROOF. Let $|V_{F'}| = |V_F| = v$, then for any $x \in S^v$,

$$p_F(x) \le p_{F'}(x) \le \frac{m_T}{m_{T'}} p_F(x)$$

with equality on the left if F' has exactly one edge contained in each edge of F and equality on the right if each edge of F contains all $\frac{m_T}{m_{T'}}$ possible edges of F'. Hence,

$$\max_{x \in S^v} p_F(x) \le \max_{x \in S^v} p_{F'}(x) \le \max_{x \in S^v} \frac{m_T}{m_{T'}} p_F(x).$$

This implies that

$$\frac{m_{T'}}{m_T}b_T(F) \le b_{T'}(F') \le b_T(F).$$

In particular, if F' has exactly k edges contained in each edge of F, then for any $x \in S^v$,

$$p_{F'}(x) = k p_F(x)$$

which implies the result.

1.2. Supersaturation and related results. Supersaturation holds for GDHs as it does for undirected hypergraphs, and the proof of this result is the same as the one for hypergraphs found in [29] with only minor differences.

THEOREM 9.2 (Supersaturation). Let F be a GDH on k elements. Let $\epsilon > 0$. For sufficiently large $n \ge n_0(F, \epsilon)$, any GDH G on n elements with density $d(G) \ge \pi_T(F) + \epsilon$ will contain at least $c\binom{n}{k}$ copies of F for some constant $c = c(F, \epsilon)$.

PROOF. Fix some positive integer l so that

$$\operatorname{ex}_{T}(l,F) < \left(\pi_{T}(F) + \frac{\epsilon}{2}\right) \frac{r!}{m_{T}} \binom{l}{r}.$$

Let G be a GDH on n > l elements with edge density $d_T(G) \ge \pi(F) + \epsilon$. Then G must contain more than $\frac{\epsilon}{2} \binom{n}{l}$ *l*-sets with density at least $\pi_T(F) + \frac{\epsilon}{2}$. Otherwise, at most $\frac{\epsilon}{2} \binom{n}{l}$ *l*-sets contain more than $(\pi_T(F) + \frac{\epsilon}{2}) \binom{l}{r}$ edges. Therefore, we can count the number of edges in G by *l*-sets and get an upper bound of

$$\binom{n-r}{l-r}e_T(G) \le \frac{\epsilon}{2}\binom{n}{l}\binom{l}{r}\frac{r!}{m_T} + \left(1 - \frac{\epsilon}{2}\right)\binom{n}{l}\left(\pi_T(F) + \frac{\epsilon}{2}\right)\binom{l}{r}\frac{r!}{m_T}.$$

We can now replace $e_T(G)$ since

$$e_T(G) \ge (\pi_T(F) + \epsilon) \binom{n}{r} \frac{r!}{m_T}$$

This is enough to get the contradiction.

Since G contains more than $\frac{\epsilon}{2} {n \choose l} l$ -sets with density at least $\pi_T(F) + \frac{\epsilon}{2}$, then it contains a copy of F in each. A given copy of F appears in ${n-k \choose l-k} l$ -sets of G. Therefore, there are more than

$$\frac{\epsilon}{2} \binom{n}{l} \binom{n-k}{l-k}^{-1} = c \binom{n}{k}$$

distinct copies of F in G where

$$c = \frac{\epsilon}{2} \binom{l}{k}^{-1}.$$

Similarly, the following theorem is an extension from the same result for undirected hypergraphs, and the proof is an adaptation of the one found in [29].

THEOREM 9.3. Let F be a GDH on k vertices and let $t = (t_1, \ldots, t_k)$ be an k-tuple of positive integers. Then $\pi_T(F) = \pi_T(F(t))$.

PROOF. That $\pi_T(F) \leq \pi_T(F(t))$ is trivial since F(t) contains a copy of F so any F-free GDH is automatically F(t)-free.

Therefore, we only need to show that $\pi_T(F) \geq \pi_T(F(t))$. Suppose not, then for sufficiently large *n* there exists some F(t)-free GDH *G* on *n* elements with edge density strictly greater than $\pi_T(F)$. By supersaturation this implies that *G* contains $c\binom{n}{k}$ copies of *F*.

Define G^* to be the k-uniform hypergraph where $V_{G^*} = V_G$ and $\{a_1, \ldots, a_k\} \in E_{G^*}$ iff and only if $\{a_1, \ldots, a_k\}$ contains a copy of F in G. Since the edge density of G^* is c > 0, then for large enough n, G^* must contain an arbitrarily large complete k-partite subgraph.

For each edge F maps to the vertices in at least one out of k! total possible ways to make an injective homomorphism in G. Therefore, by Ramsey Theory, if we take the parts of this complete k-partite subgraph large enough and color the edges by the finite number of non-isomorphic ways that F could possibly map to the k vertices, we will get an arbitrarily large monochromatic k-partite subgraph where each part has t vertices. This must have been a copy of F(t) in G, a contradiction.

The fact that the Turán density of a blowup equals the Turán density of the original GDH leads to the following nice characterization of degenerate families of GDH - those families with Turán density zero.

THEOREM 9.4 (Characterization of Degenerate GDH). Let \mathcal{F} be some family of GDHs, then $\pi_T(\mathcal{F}) = 0$ if and only if some member $F \in \mathcal{F}$ is a subGDH of the t-blowup of a single edge for some vector, $t = (t_1, \ldots, t_r)$, of positive integers. Otherwise, $\pi(\mathcal{F}) \geq \frac{m_T}{r^r}$.

2. JUMPS

PROOF. Suppose that no member of \mathcal{F} is such a blowup. Then no member is contained in the (t, t, \ldots, t) -blowup of S. Let S(t) stand for this blowup, then the sequence of GDHs, $\{S(t)\}_{t=1}^{\infty}$, is an \mathcal{F} -free sequence. The density of any such S(t) is

$$d_T(S(t)) = \frac{t^r}{\frac{r!}{m_T} \binom{tr}{r}} = \frac{m_T t^r (tr - r)!}{(tr)!}.$$

These densities tend to $\frac{m_T}{r^r}$ as t increases. Therefore,

$$\pi_T(\mathcal{F}) \ge \frac{m_T}{r^r} > 0.$$

Conversely, suppose some $F \in \mathcal{F}$ is a (t_1, \ldots, t_r) -blowup of a single edge. By Theorem 9.3, $\pi_T(F) = \pi_T(S) = 0$ since $\exp(n, S) = 0$ for all n. Therefore, $\pi_T(\mathcal{F}) = 0$.

2. Jumps

Now we turn to the issue of finding jumps and nonjumps for GDH theories. The definition of a jump for undirected hypergraphs extends naturally to this setting as does the important connection between jumps and blowup densities.

DEFINITION 9.14. Let T be a GDH theory, then $\alpha \in [0,1)$ is a jump for T if there exists a c > 0 such that for any $\epsilon > 0$ and any positive integer l, there exists a positive integer $n_0(\alpha, \epsilon, l)$ such that any GDH G on $n \ge n_0$ elements that has at least $(\alpha + \epsilon) \frac{r!}{m_T} {n \choose r}$ edges contains a subGDH on l elements with at least $(\alpha + c) \frac{r!}{m_T} {l \choose r}$ edges.

Note that by Theorem 9.4 every $\alpha \in [0, \frac{m_T}{r^r})$ is a jump for any *r*-ary GDH theory *T*. This generalizes the well-known result of Erdős **[18]** that every $\alpha \in [0, \frac{r!}{r^r})$ is a jump for *r*-graphs. The following important theorem on jumps for GDH theories was originally shown by Frankl and Rödl **[27]** for undirected hypergraphs. Their proof works equally well in this setting so the differences here are in name only.
2. JUMPS

THEOREM 9.5. The GDH theory T has a jump α if and only if there exists a finite family \mathcal{F} of GDHs such that $\pi_T(\mathcal{F}) \leq \alpha$ and $b_T(F) > \alpha$ for each $F \in \mathcal{F}$.

PROOF. Let α be a jump and let c be the supremum of all corresponding lengths c to the jump. Fix a positive integer k so that

$$\binom{k}{r}\left(\alpha + \frac{c}{2}\right) > \alpha \frac{k^r}{r!}$$

Let \mathcal{F} be the family of all GDHs on k elements with at least $\left(\alpha + \frac{c}{2}\right) {k \choose r} \frac{r!}{m_T}$ edges. Then $\pi_T(\mathcal{F}) \leq \alpha$ since any slightly larger density implies arbitrarily large subsets with density $\alpha + c$. This in turn would imply the existence of a k-subset with density at least $\alpha + c$. This k-subset would include some member of \mathcal{F} . On the other hand, a given $F \in \mathcal{F}$ will have blowup density

$$b_T(F) \ge m_T p_F\left(\frac{1}{k}, \dots, \frac{1}{k}\right) > \alpha.$$

Conversely, suppose that such a finite family $\mathcal{F} = \{F_1, \ldots, F_k\}$ exists. Let $\epsilon > 0$ and let $\{G_n\}$ be an infinite sequence of GDHs with density that tends to $\alpha + \epsilon$. As in the proof of Theorem 9.2, for any positive integer l, G_n must contain at least $\frac{\epsilon}{2} {n \choose l}$ l-subsets with density at least $\alpha + \frac{\epsilon}{2}$.

Let l be large enough so that any GDH on l vertices with density at least $\alpha + \frac{\epsilon}{2}$ contains some F_i from \mathcal{F} . Therefore, any G_n with n > l contains $\frac{\epsilon}{2} {n \choose l} l$ -sets each with some F_i . Since there are only k members of \mathcal{F} , then this implies that at least $\frac{\epsilon}{2k} {n \choose l} l$ -sets contain the same F_i .

Let $|V(F_i)| = v_i$. By the proof of Theorem 9.2 this implies that there is some positive constant b such that G_n contains at least $b\binom{n}{v_i}$ distinct copies of F_i . By the proof of Theorem 9.3 this shows that if n is large enough, then we get a copy of an arbitrarily large t-blowup of F_i .

Let $c = \min_{F_i \in \mathcal{F}} b_T(F_i)$. For some subset $\mathcal{F}' \subseteq \mathcal{F}$, each $F_i \in \mathcal{F}'$ yields an infinite subsequence of $\{G_n\}$ which contains arbitrarily large *t*-blowups of F_i . The densities of these blowups all tend to at least c. Therefore, for any positive integer m, there exists an m-set of each $\{G_n\}$ for sufficiently large n with density at least $\alpha + c$. Hence, α is a jump.

The following proposition is needed to compare jumps between different GDH theories.

PROPOSITION 9.15. The GDH theory T has a jump α if and only if there exists some c > 0 such that for all families \mathcal{F} of GDHs, either $\pi_T(\mathcal{F}) \leq \alpha$ or $\pi_T(\mathcal{F}) \geq \alpha + c$.

PROOF. Let α be a jump for T and let c > 0 be some corresponding length to the jump. Suppose that \mathcal{F} is a finite family of GDHs of type T for which $\alpha < \pi_T(\mathcal{F}) < \alpha + c$. Let $\{G_n\}$ be a sequence of extremal \mathcal{F} -free GDHs. For each positive integer kthere exists some G_n that contains a k-subset with at least $(\alpha + c)\binom{k}{r}\frac{r!}{m_T}$ edges. Take the sequence of these subsets. They are all \mathcal{F} -free by assumption, and the limit of their densities is at least $\alpha + c$. Therefore, $\pi_T(\mathcal{F}) \geq \alpha + c$, a contradiction.

Conversely, assume that α is not a jump. Let c > 0, then for some $0 < \epsilon < c$ and some positive integer l, there exists an infinite sequence of GDHs, $\{G_n\}$ for which each GDH has density at least $\alpha + \epsilon$ and all l-sets have strictly less than $(\epsilon + c) {l \choose r} \frac{r!}{m_T}$ edges. Hence, $\{G_n\}$ is \mathcal{F} -free where \mathcal{F} is the set of all l-GHDs with at least $(\alpha + c) {l \choose r} \frac{r!}{m_T}$ edges. So $\pi_T(\mathcal{F}) \geq \alpha + \epsilon$. Since any GDH with density at least $\alpha + c$ must have an l-set with density at least $\alpha + c$, then $\pi_T(\mathcal{F}) < \alpha + c$.

We will now look at how jumps are related between two different GDH theories for some fixed edge size r. We will see that in general jumps always "pass up" the subgroup lattice. That is, if $J_{T'} \subseteq J_T$ for GDH theories T' and T, then a jump for T'is a jump for T. The converse is not true in general. In fact, for any GDH theories T' and T with $J_{T'} \subseteq J_T$ such that the order of J_T is at least three times that of $J_{T'}$ we will show that the set of jumps for T' is not equal to the set of jumps for T. The case where $m_T = 2m_{T'}$ is open. **2.1. Jumps pass up the lattice.** First, we will show that for GDH theories T and T' with $J_{T'} \subseteq J_T$ the set of Turán densities of forbidden families of T-graphs is a subset of the set of Turán densities for T'.

THEOREM 9.6. Let T and T' be two GDH theories such that $J_{T'} \subseteq J_T$. Then for any family \mathcal{F} of T-graphs there exists a family \mathcal{F}' of T'-graphs for which $\pi_{T'}(\mathcal{F}') = \pi_T(\mathcal{F})$. Moreover, if \mathcal{F} is a finite family, then \mathcal{F}' is also finite.

PROOF. For each $F \in \mathcal{F}$ let $F_{T'}$ be the set of all T'-graphs that have exactly one edge contained in every edge of F. That is, since $J_{T'} \subseteq J_T$, then there are $\frac{m_T}{m_{T'}}$ possible T' edges contained within one T edge. So $F_{T'}$ is a finite set with at most $\left(\frac{m_T}{m_{T'}}\right)^{e_T(F)}$ members. Let

$$\mathcal{F}' = \bigcup_{F \in \mathcal{F}} F_{T'}.$$

Then \mathcal{F}' is a family of T'-graphs. Moreover, \mathcal{F}' is finite if \mathcal{F} is finite. We want to show that $\pi_{T'}(\mathcal{F}') = \pi_T(\mathcal{F})$.

First, let $\{G'_n\}$ be an extremal \mathcal{F}' -free sequence of T'-graphs. For each G'_n let G_n be the T-graph constructed by replacing each T'-edge of G'_n with its containing T-edge (multiple T'-edges could correspond to the same T-edge but each T-edge can only be added once).

The sequence $\{G_n\}$ is \mathcal{F} -free since otherwise some G_n contains some $F \in \mathcal{F}$ which means that G'_n must have contained at least one member of $F_{T'}$. Therefore,

$$\pi_T(\mathcal{F}) \ge \lim_{n \to \infty} d_T(G_n) \ge \lim_{n \to \infty} \frac{\frac{m_{T'}}{m_T} e_{T'}(G'_n)}{\frac{r!}{m_T} \binom{n}{r}} = \lim_{n \to \infty} \frac{ex_{T'}(n, \mathcal{F}')}{\frac{r!}{m_{T'}} \binom{n}{r}} = \pi_{T'}(\mathcal{F}').$$

Conversely, now let $\{G_n\}$ be an extremal \mathcal{F} -free sequence of T-graphs. For each G_n construct a T'-graph G'_n by replacing each T-edge with all $\frac{m_T}{m_{T'}}$ T'-edges contained in it. The sequence $\{G'_n\}$ is \mathcal{F}' -free with $\frac{m_T}{m_{T'}} \exp_T(n, \mathcal{F})$ edges. Therefore,

$$\pi_{T'}(\mathcal{F}') \geq \lim_{n \to \infty} \frac{\frac{m_T}{m_{T'}} \operatorname{ex}_T(n, \mathcal{F})}{\frac{r!}{m_{T'}} \binom{n}{r}} = \pi_T(\mathcal{F}).$$



FIGURE 40. $\pi(F) = \frac{4}{27}$.

So $\pi_{T'}(\mathcal{F}') = \pi_T(\mathcal{F}).$

The converse of Theorem 9.6 is false in general. For example, the permutation subgroup for the theory T' of $(2 \rightarrow 1)$ -uniform directed hypergraphs is a subgroup of the permutation group for the theory T of undirected 3-graphs, S_3 . The extremal number for the directed hypergraph R_4 (see Chapter 2) is

$$ex_{T'}(n, R_4) = \left\lfloor \frac{n}{3} \right\rfloor \begin{pmatrix} \left\lceil \frac{2n}{3} \right\rceil \\ 2 \end{pmatrix}.$$

Therefore, the Turán density is $\pi_{T'}(R_4) = \frac{4}{27}$. However, it is well-known that no Turán densities exist for 3-graphs in the interval $(0, \frac{6}{27})$.

COROLLARY 9.16. Let T and T' be two GDH theories such that $J_{T'} \subseteq J_T$. If α is a jump for T', then it is also a jump for T.

PROOF. If α is not a jump for T, then for any c > 0 there exists by Proposition 9.15 a family \mathcal{F} such that $\alpha < \pi_T(\mathcal{F}) < \alpha + c$. So by Theorem 9.6 there exists a family \mathcal{F}' of T'-graphs with $\alpha < \pi_{T'}(\mathcal{F}') < \alpha + c$. So α is not a jump for T'.

Corollary 9.16 immediately implies that all nonjumps found for *r*-uniform undirected hypergraphs must also be non-jumps for any GDH with an *r*-ary relation. However, the converse is not true in general.

2.2. Jumps do not pass down the lattice. Roughly speaking, the current best method of demonstrating that a particular α is not a jump for *r*-uniform hypergraphs is to construct a sequence of hypergraphs each with blowup densities that are strictly

larger than α but for which any relatively small subgraph has blowup density at most α . This method originated in [27] and generalizes to GDHs as the following definition and lemma demonstrate.

DEFINITION 9.17. Let $\alpha \in [0,1)$. Call α a demonstrated nonjump for a GDH theory T if there exists an infinite sequence of GDHs, $\{G_n\}$, such that $b_T(G_n) > \alpha$ for each G_n in the sequence and for any positive integer l there exists a positive integer n_0 such that whenever $n \ge n_0$ then any subGDH $H \subseteq G_n$ on l or fewer vertices has blowup density $b_T(H) \le \alpha$.

LEMMA 9.18. Every demonstrated nonjump is a nonjump.

PROOF. Suppose not. Assume that α is a demonstrated nonjump but is a jump. Then there exists a finite family of GDHs \mathcal{F} such that $\pi_T(\mathcal{F}) \leq \alpha$ and $b_T(F) > \alpha$ for each $F \in \mathcal{F}$. Let l be the maximum number of vertices over the members of \mathcal{F} . Let n be large enough so that any subGDH on l or fewer vertices has blowup density at most α . Then some large enough blowup of G_n contains some $F \in \mathcal{F}$ as a subGDH since the blowup density of each G_n tends to something strictly greater than α . Let H be the minimal subGDH of G_n for which the corresponding blowup contains this copy of F. Since H has at most l vertices, then it has a blowup density at most α . Hence,

$$b_T(F) \le b_T(H(t)) \le b_T(H) \le \alpha,$$

a contradiction.

We can now show that a demonstrated nonjump for a GDH theory T yields multiple nonjumps of equal and lesser values down the lattice to GDH theories T' for which $J_{T'} \subseteq J_T$.

THEOREM 9.7. Let T and T' be GDH theories such that $J_{T'} \subseteq J_T$. Let α be a demonstrated nonjump for T. Then $\frac{km_{T'}}{m_T}\alpha$ is a demonstrated nonjump for T' for $k = 1, \ldots, \frac{m_T}{m_{T'}}$.

2. JUMPS

PROOF. Let α be a demonstrated nonjump for T. Let $\{G_n\}$ be the corresponding infinite sequence of GDHs. Fix some $k \in \{1, \ldots, \frac{m_T}{m_{T'}}\}$. For each n let G'_n be a T'-graph constructed from G_n by replacing each T-edge with k T'-edges in any orientation. Then by Proposition 9.13 we know that

$$b_{T'}(G'_n) = \frac{km_{T'}}{m_T} b_T(G_n)$$

and any $H' \subseteq G'_n$ corresponding to $H \subseteq G_n$ also gives:

$$b_{T'}(H') = \frac{km_{T'}}{m_T} b_T(H).$$

Therefore, $b_{T'}(G'_n) > \frac{km_{T'}}{m_T} \alpha$ for each n and for any positive integer l, there exists a n_0 such that $b_{T'}(H) \leq \alpha$ for any subGDH $H \subseteq G_n$ for all $n \geq n_0$.

Constructions of sequences of undirected r-graphs which show that $\frac{5r!}{2r^r}$ is a demonstrated nonjump for each $r \ge 3$ were given in [26]. This gives the following corollary.

COROLLARY 9.19. Let T be an r-ary GDH theory for $r \ge 3$. Then $\frac{5m_Tk}{2r^r}$ is a nonjump for T for $k = 1, \ldots, \frac{r!}{m_T}$.

This in turn shows that the set of jumps for a theory T' is a proper subset of the set of jumps for T for any T such that $J_{T'} \subseteq J_T$ and $m_T \ge 3m_{T'}$.

COROLLARY 9.20. Let T and T' be r-ary GDH theories such that $J_{T'} \subseteq J_T$ and $m_T \geq 3m_{T'}$. Then there exists an α that is a nonjump for T' and a jump for T.

PROOF. Take k = 1, then $\frac{5m_{T'}}{2r^r}$ is a nonjump for T'. Since $m_T \ge 3m_{T'}$, then $m_T > 2.5m_{T'}$. So

$$\frac{5m_{T'}}{2r^r} < \frac{m_T}{r^r}.$$

Therefore, $\frac{5m_{T'}}{2r^r}$ is a jump for T since every $\alpha \in [0, \frac{m_T}{r^r})$ is a jump for T.

3. Continuity and approximation

The following two results are direct adaptations of two theorems from $[\mathbf{Z}]$. They are both general extremal results related to everything discussed in this chapter but did not fit nicely into the other sections. The first result on continuity relates extremal numbers of any infinite family of GDHs to the extremal numbers of its finite subfamilies. The second result on approximation discusses structural aspects of (nearly) extremal sequences for any forbidden family.

THEOREM 9.8 (Continuity). Let \mathcal{F} be an infinite family of T-graphs. For each $\epsilon > 0$ there exists a finite subfamily $\mathcal{F}_{\epsilon} \subset \mathcal{F}$ such that

$$ex_{T}(n,\mathcal{F}) \leq ex_{T}(n,\mathcal{F}_{\epsilon}) < ex_{T}(n,\mathcal{F}) + \epsilon n^{r}$$

for sufficiently large n.

PROOF. Let \mathcal{F} be the infinite family of GDHs. For each positive integer k let \mathcal{F}_k be the subfamily of \mathcal{F} where each member has at most k vertices. Let

$$\gamma_k = \lim_{n \to \infty} \frac{\exp\left(n, \mathcal{F}_k\right)}{\frac{r!}{m_T} \binom{n}{r}}$$

and let

$$\gamma = \lim_{n \to \infty} \frac{\exp\left(n, \mathcal{F}\right)}{\frac{r!}{m_T} \binom{n}{r}}.$$

Since $\mathcal{F}_k \subset \mathcal{F}$, then $\{\gamma_k\}_{k=1}^{\infty}$ is a monotone decreasing sequence and $\gamma_k \geq \gamma$ for all k. Assume for some $\epsilon > 0$ that $\gamma_k > \gamma + \epsilon$ for all k. Note that

$$\frac{\exp\left(n, \mathcal{F}_{k}\right)}{\frac{r!}{m_{T}} \binom{n}{r}} \geq \gamma_{k}$$

is true for all *n*. In particular, when n = k there is an \mathcal{F}_n -free GDH on *n* vertices with strictly more than $(\gamma + \epsilon) \frac{r!}{m_T} {n \choose r}$ edges. Since an \mathcal{F}_n -free GDH on *n* vertices is also necessarily \mathcal{F} -free, then this implies that

$$\exp(n, \mathcal{F}) > (\gamma + \epsilon) \frac{r!}{m_T} {n \choose r},$$

a contradiction.

Theorem 6 in [7] is the Approximation Theorem for totally directed *r*-uniform hypergraphs with bounded multiplicity. We will use the following equivalent statement (in the case of multiplicity one) written in terms of Turán densities as a lemma to prove that this approximation result holds for all GDHs.

LEMMA 9.21. Let \mathcal{F}' be a family of forbidden totally directed r-graphs (r-GDHs under the trivial group), and let $\epsilon > 0$. Then there exists some totally directed rgraph G' such that every blowup of G' is \mathcal{F}' -free and

$$\pi(\mathcal{F}') \ge b(G) > \pi(\mathcal{F}') - \epsilon.$$

THEOREM 9.9 (Approximation). Let \mathcal{F} be a family of forbidden T-graphs, and let $\epsilon > 0$, then there exists some T-graph G for which all blowups of G are \mathcal{F} -free and

$$\pi_T(\mathcal{F}) \ge b_T(G) > \pi_T(\mathcal{F}) - \epsilon.$$

PROOF. Let \mathcal{F}' be the family of totally directed *r*-graphs as defined in the proof of Theorem 9.6. That is, the family of directed hypergraphs for which we know that $\pi(\mathcal{F}') = \pi_T(\mathcal{F})$. We know from the proof of that theorem that any *T*-graph that is the minimal container for an \mathcal{F}' -free graph is \mathcal{F} -free. By Lemma 9.21 there exists some totally directed \mathcal{F}' -free *r*-graph , G', such that

$$\pi(\mathcal{F}') \ge b(G) > \pi(\mathcal{F}') - \epsilon.$$

By Proposition 9.13 we know that if G' is the minimal containing T-graph of G, then $b_T(G) \ge b(G')$. Hence,

$$\pi_T(\mathcal{F}) \ge b_T(G) \ge b(G') > \pi(\mathcal{F}') - \epsilon = \pi_T(\mathcal{F}) - \epsilon.$$

CHAPTER 10

Additional Questions about Directed Hypergraphs

This chapter contains material from a paper published by the Electronic Journal of Combinatorics. [12]

There are many additional questions that we can ask about $2 \rightarrow 1$ directed hypergraphs and about GDHs in general. In this chapter, we will briefly review several open questions that come up naturally in this work.

1. Extremal numbers for tournaments

Brown and Harary [6] started studying extremal problems for directed 2-graphs by determining the extremal numbers for many "small" digraphs and for some more general types of digraphs such as tournaments - a digraph where every pair of vertices has exactly one directed edge. We could follow their plan of attack in studying the $2 \rightarrow 1$ model and look for the extremal numbers of tournaments. Here, a tournament could be defined as a graph with exactly one directed edge on every three vertices. In particular, a transitive tournament might be an interesting place to begin. A transitive tournament is a tournament where the direction of each edge is based on an underlying linear ordering of the vertices as in the oriented lower bound construction of Theorem [4,2]

Denote the $2 \to 1$ transitive tournament on k vertices by TT_k . Since the "winning" vertex of the tournament will have a complete K_{k-1} as its tail link graph, then any H on n vertices for which each T_x is K_{k-1} -free must be TT_k -free. Therefore,

$$n\left(\frac{n-1}{k-2}\right)^2 \binom{k-2}{2} \le \exp(n, TT_k), \exp(n, TT_k).$$

This also immediately shows that the transitive tournament on four vertices with the "bottom" edge removed has this extremal number exactly.

THEOREM 10.1. Let TT_4^- denote the graph with vertex set $V(TT_4^-) = \{a, b, c, d\}$ and edge set

$$E(TT_4^-) = \{ab \to d, bc \to d, ac \to d\}.$$

Then

$$ex(n, TT_4^-) = n \left\lfloor \frac{n-1}{2} \right\rfloor \left\lceil \frac{n-1}{2} \right\rceil.$$

Is it still true if we add an edge to $\{a, b, c\}$?

CONJECTURE 10.1. Let TT_4 denote the graph with vertex set $V(TT_4) = \{a, b, c, d\}$ and edge set

$$E(TT_4) = \{ab \to d, bc \to d, ac \to d, ab \to c\}.$$

Then

$$ex(n, TT_4) = n \left\lfloor \frac{n-1}{2} \right\rfloor \left\lceil \frac{n-1}{2} \right\rceil.$$

2. GDHs with $r \to 1$ edges

The 2 \rightarrow 1 directed hypergraph originally came to the author's attention as a way to model definite Horn clauses in propositional logic. Definite Horn clauses are more generally modeled by $r \rightarrow 1$ edges for any r. Therefore, it seems natural to ask about the extremal numbers for graphs with two $(r \rightarrow 1)$ -edges. If we look at every $(r \rightarrow 1)$ -graph with exactly two edges, then we see that these fall into four main types of graph. Let i be the number of vertices that belong to the tail set of both edges. Then let $I_r(i)$ denote the graph where both edges point to the same head vertex, let $H_r(i)$ denote the graph where the edges point to different head vertices neither of which are in the tail set of the other, let $R_r(i)$ denote the graph where the first edge points to a head vertex in the tail set of the second edge and the second edge points to a head not in the tail set of the first edge, and let $E_r(i)$ denote the graph where both edges point to heads in the tail sets of each other. This extends the notation used in this paper. The degenerate cases are generalized to $I_r(i)$ and $H_r(i)$, and the nondegenerate cases generalize to $R_r(i)$ and $E_r(i)$. For example, the 3-resolvent R_3 is $R_2(1)$. The split between degenerate and nondegenerate is maintained in this way as well as shown in Theorem 9.4

To what extent do the proofs presented in this paper extend to these graphs? Some translate immediately. For example, in the standard version of the problem it can easily be seen that

$$\operatorname{ex}(n, I_r(0)) = n \binom{n-2}{r-1}$$

using Erdős-Ko-Rado [21] for the upper bound and the same basic construction for the lower bound that we used in proving the same result for I_0 . More generally, we can get an upper bound of

$$\exp(n, I_r(i)) \le n \binom{n-1}{r-1}$$

by applying the uniform Ray-Chaudhuri - Wilson Theorem [39] to the tail link graph of each vertex of an $I_r(i)$ -free graph. We can get a general lower bound of

$$n\binom{n-i-2}{r-i-1} \le \exp(n, I_r(i))$$

by constructing an $I_r(i)$ -free graph in the following way: for each vertex x fix a set of i + 1 vertices not including x, C_x , and then add every possible edge with x at the head and C_x in the tail set.

An easy lower bound construction for an $H_r(i)$ -free graph is to fix a vertex x and take all possible edges that point to it giving

$$\binom{n-1}{r} \le \exp(n, H_r(i)).$$

To get an upper bound also on the order of n^r note that we can extend the concept of the directed link graph to apply to more than one vertices. For instance, here let the directed link graph of a set of vertices A of cardinality i be the $(r - i) \rightarrow 1$ directed hypergraph on n-i vertices, $V \setminus A$, for which every edge becomes an edge of the original $(r \to 1)$ -graph when A is added to the tail set. In this case, no directed tail link graph for any set of *i* vertices can contain two independent directed edges.

Therefore,

$$\exp(n, H_r(i)) \le \frac{(r-i+1)\binom{n-i-1}{r-i}\binom{n}{i}}{\binom{r}{i}} = \frac{n(r-i+1)}{n-i}\binom{n-1}{r}.$$

It is easy to see that any $r \to 1$ transitive tournament on n vertices would be $E_r(i)$ free. This immediately solves the oriented version and gives a lower bound for the standard version:

$$ex_o(n, E_r(i)) = \binom{n}{r+1}.$$

As in the first lower bound construction for E we can add r edges to the smallest r+1 vertices in the linear order given by the transitive tournament to get a few more edges in the standard case. Is this the best that we can do?

CONJECTURE 10.2.

$$ex(n, E_r(i)) = \binom{n}{r+1} + r.$$

For the generalized resolvent configurations, the lower bound constructions for R_3 and R_4 both generalize to the $r \to 1$ setting. When $i \ge 1$, then the construction that worked for R_3 gives the better lower bound. Split the vertices into two equal or almost equal parts and take all edges that point from an *r*-set in one to a vertex in the other. This gives

$$n\binom{\frac{n}{2}}{r} \le \exp(n, R_r(i))$$

for $i \ge 1$. When i = 0, the same generalization of the construction for R_4 will produce an $R_r(0)$ -free graph.

3. Differences between oriented and standard extremal numbers

It is interesting to look at the differences between the oriented and standard extremal problems for a given forbidden graph not only in their values but in the difficulty level of their proofs. For instance, the proof of the standard case of I_0 is quite easy while the proof of the oriented case took a lot of effort. For the Escher graph E the situation was reversed. What about the character of these two graphs determines that one version of the problem should be easy and the other difficult, and what is the difference between the two that swaps which version is which?

A more exact request is to ask for a characterization that determines the difference in the value. For instance, H_2 , I_1 , R_3 , R_4 , and the case of two completely overlapping edges each have oriented and standard numbers that are exactly the same while H_1 and I_0 each have differences that are linear in n, the Escher graph E has a constant difference, and the graph made up of two independent edges has a quadratic difference.

Of course, we get an immediate easy bound by observing that every non-oriented F-free graph contains an oriented F-free graph that can be arrived at by removing edges from each triple of vertices until only one remains. So

$$\exp(n, F) \le 3\exp_o(n, F) \le 3\exp(n, F)$$

for any forbidden graph F. The cases in this paper where the difference between the two numbers is zero shows that the upper bound is tight while the case of two independent edges shows that the lower bound is also tight.

But what causes the difference? Perhaps, it would be good to begin answering this question by narrowing the focus to nondegenerate graphs since in this paper almost every nondegenerate case had no difference in the values, and the only one that did had only constant difference. Will the difference always be at most constant or at least $o(n^3)$? No, any graph F that contains a triple with all three possible edges is

certainly not degenerate, and the standard extremal number of F is at least twice as much as the oriented extremal number.

But what if we restrict ourselves further and only consider oriented nondegenerate forbidden graphs, then is

$$ex(n, F) - ex_o(n, F) = o(n^3)$$

for every oriented nondegenerate F? Dániel Gerbner and Balázs Keszegh produced an interesting counterexample to this claim as well. At this point it is unclear to the author what might be an appropriate characterization for graphs with small differences between extremal numbers.

4. General structural results

On a more general level we can ask about the structure of extremal $(2 \rightarrow 1)$ -graphs. For instance, it was already shown in [32] that the 4-resolvent configuration R_4 has a stability result. Roughly speaking, R_4 -free graphs with many edges differ only slightly from the given extremal construction. While we have shown that several of the extremal constructions in this paper are unique, we have not shown that any are stable.

Another avenue of research is to ask for canonical extremal structures. That is, for a forbidden graph F can we fix some constant r such that we can construct an F-free graph on n vertices such that the n vertices are partitioned into r parts and whether $xy \to z$ is an edge or not depends entirely on which parts x, y, and z are in? If we have a general r-part structure like this that is F-free for every n and the limit of the ratio of the number of edges given by the structure over ex(n, F) is one, then we call this a canonical F-free extremal structure. For instance, the Turán graphs are canonical extremal structures with respect to 2-graphs. Applying this idea to hypergraphs is already a major area of research (see [38]) so it seems likely that the question of whether every $(2 \rightarrow 1)$ -graph has such a canonical extremal structure would be even more difficult.

5. GDH questions

It would be nice to show that the set of jumps for some GDH theory T' is a proper subset of the set of jumps of any theory T' up the lattice including those for which $m_T = 2m_{T'}$. Or on the other hand it would be very interesting to learn that this is not true in certain cases for $r \ge 3$!

CONJECTURE 10.3. Let T' and T be r-ary GDH theories for $r \ge 3$ such that $J_{T'} \subseteq J_T$ and $m_T = 2m_{T'}$. Then there exists some $\alpha \in [0, 1)$ for which α is a jump for T but not for T'.

It is known by a result in $[\mathbb{Z}]$ that every $\alpha \in [0, 1)$ is a jump for digraphs. Therefore, the conjecture is not true when r = 2. On a related note, is it always true that when $J_{T'} \subset J_T$, there always exists a family \mathcal{F}' of T'-graphs such that $\pi_{T'}(\mathcal{F}')$ is not contained in the set of Turán densities for T?

CONJECTURE 10.4. Let T' and T be theories such that $J_{T'} \subseteq J_T$. Then there exists some family \mathcal{F}' of T'-graphs such that $\pi_{T'}(\mathcal{F}')$ is not contained in the set of Turán densities for T.

Finally, it would be nice to generalize the definition of a GDH to include other combinatorial structures. For instance we could easily change the current formulation to include multiple relations in order to capture nonuniform GDHs and those with edges that have bounded multiplicity like the structures studied in \square . We could even allow these theories to contain general statements that relate the different relations. An example of this might be the theory of some kind of GDH with an edge-coloring that behaves in a certain way (at least locally). In another direction we could take away the requirement that all vertices of an edge be distinct to allow for kinds of generalized loops or add a condition that the existence of certain edges preclude the existence of others such as in the oriented cases studied here and in [33].

CHAPTER 11

Introduction to the (p,q)-coloring problem.

We will use the standard asymptotic notation in the following chapters. That is, for two functions, f(n) and g(n), we write f = O(g) if there exists some constant c and some integer N such that $f(n) \leq cg(n)$ for all $n \geq N$. We write f = o(g) if $f/g \to 0$ as $n \to \infty$. We write $f = \Omega(g)$ if g = O(f) and $f = \omega(g)$ if g = o(f). Finally, we write $f = \Theta(g)$ if f = O(g) and $f = \Omega(g)$.

Given two integers $s, t \ge 2$, the central question in classical Ramsey theory for graphs asks for the minimum number of vertices N for which any 2-coloring, say red and blue, of the edges of K_N must yield a red K_s or a blue K_t . We say that N = R(s, t), the Ramsey number for s, t. This question generalizes to more than 2 colors in a natural way. That is, we let $R(s_1, \ldots, s_k)$ denote the minimum number of vertices Nfor which a coloring of the edges of K_N with k colors results in either an s_1 -clique in the first color, or an s_2 -clique in the second color, etc.

A variation of the Ramsey problem is given by the following definition.

DEFINITION 11.1. Let n, p, and q be positive integers such that $q \leq {p \choose 2}$. A (p,q)-coloring of the complete graph on n vertices, K_n , is an edge coloring,

$$c: E(K_n) \to [k],$$

for which every subset of p vertices of $V(K_n)$ span at least q distinct edge colors. Let f(n, p, q) denote the minimum number of colors k for which a (p, q)-coloring of K_n exists.

If we let q = 2 in the above definition, then determining an upper bound for the function $f(n, p, 2) \leq k$ is equivalent to giving a lower bound, $n + 1 \leq R(p, \dots, p)$,

for the Ramsey number with k colors. Similarly, giving a lower bound $k \leq f(n, p, q)$ is equivalent to giving the upper bound $R(p, \ldots, p) \leq n$ on the Ramsey number for k-1 colors.

Erdős and Shelah [20] [24] introduced the function f(n, p, q) in 1975, but it was not studied in depth until 1997 when Erdős and Gyárfás [25] looked at the growth rate of f(n, p, q) as $n \to \infty$ for fixed values of p and q. They used the local lemma to give a general upper bound for the function,

$$f(n, p, q) \le cn^{\frac{p-2}{\binom{p}{2}-q+1}}.$$

Other than this, they looked for threshold values for q in terms of p for which f(n, p, q) jumps in order of magnitude. For instance, they showed that when

$$q = \binom{p}{2} - p + 3,$$

 $f(n, p, q) = \Theta(n)$ and f(n, p, q - 1) = o(n). So the function becomes linear in n at this particular value of q. Similarly, they determined that

$$q = \binom{p}{2} - \left\lfloor \frac{p}{2} \right\rfloor + 2$$

is the first value at which f(n, p, q) is quadratic in n and

$$q = \binom{p}{2} - \left\lfloor \frac{p}{4} \right\rfloor + 1$$

is the first value for which

$$\binom{n}{2} - c \le f(n, p, q)$$

where c is some constant depending only on p.

Left as an open question was determining the threshold value of q for which f(n, p, q)first becomes polynomial in n. They showed that

$$n^{\frac{1}{p-2}} - 1 \le f(n, p, p).$$

So therefore, any $q \ge p$ gives a function f(n, p, q) that is polynomial in n. However, it was unclear what the order of magnitude of f(n, p, p-1) is in general. To this end they considered some small cases. When p = 3, they pointed out that since determining f(n, 3, 2) is equivalent to solving the multicolor Ramsey problem for 3-cliques, then

$$c_1 \frac{\log n}{\log \log n} \le f(n, 3, 2) \le c_2 \log n$$

for constants c_1, c_2 . However, for p = 4, they could not beat the probabilistic upper bound

$$f(n, 4, 3) = O(n^{1/2}).$$

For this reason, they called this the "most annoying" case.

In 1998, Mubayi 34 gave an explicit (4,3)-coloring using a subpolynomial number of colors. Specifically, he showed that

$$f(n,4,3) \le e^{O(\sqrt{\log n})}$$

In 2000, Mubayi and Eichhorn [17] demonstrated that for $p \ge 5$, this construction is in general a (p,q)-coloring for $q = 2 \lceil \log_2 p \rceil - 2$. In 2015, Conlon, Fox, Lee, and Sudakov [15] finally proved that f(n, p, p - 1) is subpolynomial for all $p \ge 3$. We will discuss the construction they came up with to demonstrate this in Chapter [12].

In addition to their general results, Erdős and Gyárfás looked at several cases for small values of p. They found that

$$\frac{5}{6}(n-1) \le f(n,4,5) \le n$$

and that

$$f(n, 9, 34) = \binom{n}{2} - o(n^2).$$

Moreover, they singled out the cases of (4, 4) and (5, 9)-colorings as being particularly interesting to look at. In 2000, Axenovich [2] gave a construction showing that $f(n, 5, 9) \leq n^{1+o(1)}$. Since Erdős and Gyárfás showed that $f(n, 5, 8) = \Theta(n)$, then this

118

reduced the difference between the known upper and lower bounds for f(n, 5, 9) to a subpolynomial factor. In 2013, E. Krop and I. Krop [30] improved the lower bound to

$$\frac{7}{4}n - 3 \le f(n, 5, 9)$$

In 2004, Mubayi gave an explicit (4, 4)-coloring which reduced the upper bound to

$$f(n, 4, 4) \le n^{1/2 + o(1)},$$

a subpolynomial factor away from the best known lower bound given by Erdős and Gyárfás of $n^{1/2} - 1$. We will discuss his construction in more detail in Chapter 14 In Chapter 13 we will give an explicit (5,5)-coloring that uses only $n^{1/3+o(1)}$ colors, a subpolynomial factor away from the best known lower bound of $n^{1/3} - 1$. Similarly, in Chapter 14 we will give an explicit (5,5)-coloring that uses only $n^{1/2+o(1)}$ colors, a subpolynomial factor away from the best known lower bound of $\Omega(n^{1/2})$. In both cases, the constructions will be combinations of a modified version of the construction given in 15 which we will define in Chapter 12 and certain "algebraic" colorings that extend the idea behind Mubayi's (4,4)-coloring [35].

CHAPTER 12

The Modified CFLS coloring.

This chapter contains material from a paper to be published in Combinatorics, Probability & Computing. [14]

In this chapter, we will define a particular instance of the general (p, p-1)-coloring of Conlon, Fox, Lee, and Sudakov [15] which we will refer to as the CFLS coloring, and we will show that this coloring avoids certain configurations. These properties will be useful in later chapters. Then we will modify the coloring and give some additional useful properties. We will not define the CFLS coloring in full generality since only a simple case is needed. We borrow part of the notation used in [15], but change it somewhat for clarity in this particular instance.

Let $n = 2^{\beta^2}$ for some positive integer β . Associate each vertex of K_n with a unique binary string of length β^2 . That is, we may assume that our vertex set is

$$V = \{0, 1\}^{\beta^2}.$$

For any vertex $v \in V$, let $v^{(i)}$ denote the *i*th block of bits of length β in v so that

$$v = (v^{(1)}, \dots, v^{(\beta)})$$

where each $v^{(i)} \in \{0, 1\}^{\beta}$.

Between two vertices $x, y \in V$, the CFLS coloring is defined by

$$\varphi_1(x,y) = \left(\left(i, \{x^{(i)}, y^{(i)}\} \right), i_1, \dots, i_\beta \right)$$

where *i* is the first index for which $x^{(i)} \neq y^{(i)}$, and for each $k = 1, ..., \beta$, $i_k = 0$ if $x^{(k)} = y^{(k)}$ and otherwise is the first index at which a bit of $x^{(k)}$ differs from the corresponding bit in $y^{(k)}$.

For convenience, when discussing any edge color α , we will let α_0 denote the first coordinate of the color (of the form $(i, \{x^{(i)}, y^{(i)}\}))$ and let α_k denote the index of the first bit difference of the kth block for $k = 1, \ldots, \beta$. Furthermore, throughout this section, we will say that two vertices x and y agree at i if $x^{(i)} = y^{(i)}$ and that x and y differ at i if $x^{(i)} \neq y^{(i)}$.

1. Avoided configurations

We will show through the following series of lemmas that the CFLS coloring avoids certain specified arrangements of edge colors.

LEMMA 12.1. The CFLS coloring forbids monochromatic odd cycles.

PROOF. Suppose there exists a sequence of distinct vertices, v_1, \ldots, v_k , for which k is odd and

$$\varphi_1(v_1, v_2) = \varphi_1(v_2, v_3) = \cdots = \varphi_1(v_{k-1}, v_k) = \varphi_1(v_k, v_1) = \alpha.$$

Let $\alpha_0 = (i, \{x, y\})$. Without loss of generality we may assume that $v_1^{(i)} = x$ and $v_2^{(i)} = y$. It follows that

$$y = v_2^{(i)} = v_4^{(i)} = \dots = v_{k-1}^{(i)} = v_1^{(i)} = x,$$

a contradiction.

LEMMA 12.2. The CFLS coloring forbids four distinct vertices $a, b, c, d \in V$ for which $\varphi_1(a, b) = \varphi_1(c, d)$ and $\varphi_1(a, c) = \varphi_1(a, d)$ (see Figure 50a).

PROOF. Assume towards a contradiction that $\varphi_1(a, b) = \varphi_1(c, d) = \alpha$ and $\varphi_1(a, c) = \varphi_1(a, d) = \gamma$. Let $\alpha_0 = (i, \{x, y\})$. Without loss of generality, $a^{(i)} = c^{(i)} = x$ and



FIGURE 41. Four configurations avoided by the CFLS coloring.

 $b^{(i)} = d^{(i)} = y$. Then $\gamma_i = 0$ since a and c agree at i, but $\gamma_i \neq 0$ as a and d differ at i, a contradiction.

LEMMA 12.3. The CFLS coloring forbids four distinct vertices $a, b, c, d \in V$ for which $\varphi_1(a, b) = \varphi_1(a, c), \varphi_1(b, d) = \varphi_1(b, c), and \varphi_1(a, d) = \varphi_1(c, d)$ (see Figure 41b).

PROOF. Assume towards a contradiction that we have $\varphi_1(a, b) = \varphi_1(a, c) = \alpha$, $\varphi_1(b, d) = \varphi_1(b, c) = \gamma$, and $\varphi_1(a, d) = \varphi_1(c, d) = \pi$. Let $\alpha_0 = (i, \{x, y\}), \gamma_0 = (j, \{s, t\})$, and $\pi_0 = (k, \{w, v\})$. Without loss of generality we may assume that $a^{(i)} = x$ and $b^{(i)} = c^{(i)} = y$. Since b and c differ at j, then $i \neq j$. Without loss of generality we may assume that $b^{(j)} = s$ and $c^{(j)} = d^{(j)} = t$. So $\pi_j = 0$, and hence, $a^{(j)} = t$ since $\varphi_1(a, d) = \pi$. Therefore, $\alpha_j = 0$, which implies that $b^{(j)} = t$, a contradiction since $s \neq t$.

LEMMA 12.4. The CFLS coloring forbids five distinct vertices $a, b, c, d, e \in V$ that contain two monochromatic paths of three edges each that share endpoints: $\varphi_1(a, b) = \varphi_1(b, c) = \varphi_1(c, d)$ and $\varphi_1(a, c) = \varphi_1(c, e) = \varphi_1(e, d)$ (see Figure 41c).

PROOF. Assume towards a contradiction that

$$\varphi_1(a,b) = \varphi_1(b,c) = \varphi_1(c,d) = \alpha$$

and

$$\varphi_1(a,c) = \varphi_1(c,e) = \varphi_1(e,d) = \gamma.$$

Let $\alpha_0 = (i, \{x, y\})$ and $\gamma_0 = (j, \{s, t\})$. Without loss of generality we may assume that $a^{(i)} = c^{(i)} = x$ and $b^{(i)} = d^{(i)} = y$. Note that $\varphi_1(a, c) = \gamma$ implies $\gamma_i = 0$. Then $e^{(i)} = d^{(i)} = y$ and $e^{(i)} = c^{(i)} = x$. So x = y, a contradiction.

LEMMA 12.5. The CFLS coloring forbids five distinct vertices $a, b, c, d, e \in V$ for which $\varphi_1(a, b) = \varphi_1(a, e) = \varphi_1(e, c)$ and $\varphi_1(a, d) = \varphi_1(d, e) = \varphi_1(b, c)$ (see Figure 41d).

PROOF. Assume towards a contradiction that $\varphi_1(a, b) = \varphi_1(a, e) = \varphi_1(e, c) = \alpha$ and $\varphi_1(a, d) = \varphi_1(d, e) = \varphi_1(b, c) = \gamma$. Let $\alpha_0 = (i, \{x, y\})$. We may assume without a loss of generality that $b^{(i)} = e^{(i)} = x$ and $a^{(i)} = c^{(i)} = y$. We also know that $b^{(k)} = a^{(k)} = e^{(k)} = c^{(k)}$ for all k < i. Since $\varphi_1(b, c) = \gamma$, then $\gamma_0 = (i, \{x, y\})$. So either $d^{(i)} = x$ or $d^{(i)} = y$. Therefore, d must agree with either a or e at i, a contradiction.

2. Modified CFLS

We will now add to the CFLS coloring to avoid the striped K_4 , an edge-coloring of four distinct vertices a, b, c, d such that every pair of non-incident edges have the same color (see Figure 42). The CFLS coloring alone will not avoid such arrangements, but the product of φ_1 with another small edge-coloring, φ_2 , will.

We will define the coloring φ_2 on the same set of vertices as the CFLS coloring, $V = \{0,1\}^{\beta^2}$. However, we will also need to consider the vertices as an ordered set. Consider each vertex to be an integer represented in binary. Then order the vertices by the standard ordering of the integers. That is, x < y if and only if the first bit at which x and y differ is zero in x and one in y. This ordering plays a large role in a recent construction by Mubayi [36] for a small case of the hypergraph version of the (p,q)-coloring problem. Note that each β -block is a binary representation of an integer from 0 to $2^{\beta} - 1$, so these blocks can be considered ordered in the same way.



FIGURE 42. A striped K_4 .

Moreover, note that if x < y and if the first β -block at which x and y differ is i, then it must be the case that $x^{(i)} < y^{(i)}$.

Let $x, y \in V$ such that x < y. We define the second coloring as

$$\varphi_2(x,y) = (\delta_1(x,y),\ldots,\delta_\beta(x,y))$$

where for each i,

$$\delta_i(x,y) = \begin{cases} -1 & x^{(i)} > y^{(i)} \\ +1 & x^{(i)} \le y^{(i)} \end{cases}$$

This construction uses 2^{β} colors. Therefore, the modified CFLS coloring, $\varphi = \varphi_1 \times \varphi_2$, uses

$$\beta^{\beta+1}2^{3\beta} = \sqrt{\log n}^{\sqrt{\log n}+1}2^{3\sqrt{\log n}} = 2^{O\left(\sqrt{\log n}\log\log n\right)}$$

colors.

LEMMA 12.6. The modified CFLS coloring φ forbids four distinct vertices $a, b, c, d \in V$ with $\varphi(a, b) = \varphi(c, d), \ \varphi(a, c) = \varphi(b, d), \ and \ \varphi(a, d) = \varphi(b, c) \ (see \ Figure \ 42).$

PROOF. Assume towards a contradiction that a striped K_4 can occur. Then, $\varphi_1(a,b) = \varphi_1(c,d) = \alpha, \ \varphi_1(a,c) = \varphi_1(b,d) = \gamma, \ \text{and} \ \varphi_1(a,d) = \varphi_1(b,c) = \pi.$ Let $\alpha_0 = (i, \{x,y\}), \ \gamma_0 = (j, \{s,t\}), \ \text{and} \ \pi_0 = (k, \{v,w\}).$ Without loss of generality, assume that $i = \min\{i, j, k\}$. Since $\varphi_1(a,b) = \varphi_1(c,d)$, exactly one of $d^{(i)}$ and $c^{(i)}$ equals $a^{(i)}$. Say $d^{(i)} = a^{(i)}$ without loss of generality. Then, by the minimality of i, it must be the case that j = i and that i < k. Let $a^{(i)} = d^{(i)} = x$, $b^{(i)} = c^{(i)} = y$, $a^{(k)} = b^{(k)} = v$, and $c^{(k)} = d^{(k)} = w$. Without loss of generality we may assume that x < y. This implies that a, d < b, c in the ordering of V as integers represented in binary. If v < w, then $\delta_k(a, c) = +1$ and $\delta_k(d, b) = -1$. Therefore, $\varphi_2(a, c) \neq \varphi_2(b, d)$, a contradiction. So, it must be the case that w < v. But then $\delta_k(a, c) = -1$ and $\delta_k(b, d) = +1$, which yields the same contradiction.

Note that to eliminate the striped K_4 configuration we needed just

$$\beta 2^{3\beta} = \sqrt{\log n} 2^{3\sqrt{\log n}} = 2^{O\left(\sqrt{\log n}\right)}$$

colors since only the first coordinate of the CFLS coloring was needed in the proof. Mubayi used on the order of $n^{1/2}$ colors to eliminate it while defining his (4,4)construction [35].

Before we move on from discussing the modified CFLS coloring, we need to point out one nice fact that will be used in Chapter 13.

LEMMA 12.7. If a < b < c, then $\varphi(a, b) \neq \varphi(b, c)$.

PROOF. Suppose $\varphi_1(a, b) = \varphi_1(b, c) = \alpha$ and that $\alpha_0 = (i, \{x, y\})$ for x < y. Then $a^{(i)} = x$ and $b^{(i)} = y$. But then $c^{(i)} = x$. Therefore, c < b, a contradiction.

CHAPTER 13

A (5,5)-coloring construction.

This chapter contains material from a paper to be published in Combinatorics, Probability & Computing. [14]

As previously stated, we know in general that $f(n, p, p) \ge \Omega\left(n^{1/(p-2)}\right)$. However, the local lemma gives the best general upper bound,

$$f(n, p, p) \le O(n^{2/(p-1)}).$$

Only for p = 3, 4 do we know of a better upper bound.

A (3,3)-coloring is equivalent to a proper edge coloring, one in which no two incident edges can have the same color. Therefore, it is well known that

$$f(n,3,3) = \begin{cases} n & n \text{ is odd} \\ n-1 & n \text{ is even} \end{cases}$$

In 2004, Mubayi [35] provided an explicit (4, 4)-coloring of K_n with only $n^{1/2}e^{O(\sqrt{\log n})}$ colors. This closed the gap for p = 4 to

$$n^{1/2} - 1 \le f(n, 4, 4) \le n^{1/2 + o(1)}.$$

His construction was the product of two colorings. The first was his earlier (4, 3)coloring which used $n^{o(1)}$ colors. The second was an "algebraic" coloring that assigned
to each vertex a vector from a two-dimensional vector space over a finite field, and
then colored each edge with an element from the base field, giving $n^{1/2}$ colors. The
algebraic part of his construction will be detailed further in Chapter 14.

The (5,5)-coloring defined in this chapter extends Mubayi's idea of combining a small (p, p-1)-coloring with an algebraic coloring to obtain the following result.

THEOREM 13.1. As $n \to \infty$,

$$f(n,5,5) \le n^{1/3} 2^{O\left(\sqrt{\log n} \log \log n\right)}$$

Before defining the algebraic part of our construction, we can systematically look at all edge-colorings of a K_5 up to isomorphism that use no more than four colors and do not contain any of the configurations eliminated in Chapter 12 to get a list of possible "bad" colorings of a K_5 that could survive the modified CFLS coloring. A careful mathematician with a free day could work through these cases by hand. A simple computer program like the inelegant one detailed in Appendix A is easier to verify. However this process is executed, we end up with three possible bad colorings of K_5 (see Figure 43). Avoiding these will require both the modified CFLS coloring and the MIP coloring defined in Section 1.

In Section 1, we define the first part of an algebraic coloring which we call the Modified Inner Product (MIP) coloring. Under this construction, each vertex is associated with a vector in a three-dimensional space over a finite field. As in Mubayi's construction [35] each edge is colored with a specific element in the base field. Some slight modifications are needed for special cases, but these will only split each color a constant number of times, ultimately giving $O(n^{1/3})$ colors used in the MIP construction.

In Section 2, we will take the product of the modified CFLS coloring defined in Chapter 12 and the first part of the MIP to get a construction that uses $n^{1/3+o(1)}$ colors and eliminates the first two of the three remaining bad configurations. Finally, in Section 3 we define the rest of the MIP coloring to eliminate the third configuration.



FIGURE 43. Three configurations not avoided by the modified CFLS coloring.

1. The Modified Inner Product coloring

Let q be some odd prime power, and let \mathbb{F}_q^* denote the nonzero elements of the finite field with q elements. The vertices of our graph will be the three-dimensional vectors over this set,

$$V = \left(\mathbb{F}_q^*\right)^3.$$

All algebraic operations used in defining the MIP coloring are the standard ones from the underlying field, and \cdot will denote the standard inner product of two vectors,

$$x \cdot y = x_1 y_1 + x_2 y_2 + x_3 y_3$$

where $x = (x_1, x_2, x_3)$ and $y = (y_1, y_2, y_3)$. Additionally, let < be any linear order on the elements of \mathbb{F}_q , and extend this to a linear order on the vectors so that

$$x < y \iff x_i < y_i$$

where $i \in \{1, 2, 3\}$ is the first position at which $x_i \neq y_i$.

The MIP coloring will be broken up into two parts, $\chi = \chi_1 \times \chi_2$. The first part χ_1 uses at most $12n^{1/3}$ colors. The second part χ_2 uses only four colors and is used to split up colors from χ_1 in order to avoid one particularly difficult configuration. In this section, we will first define χ_1 . Then, after a brief review of the necessary linear



FIGURE 44. The geometric visualization can be misleading since we are working over a finite field, but monochromatic neighborhoods are contained in affine planes.

algebra concepts, we will prove some key properties of χ_1 . The second part χ_2 will be defined in Section 5.

1.1. Motivation. The MIP coloring should be viewed as coloring each edge with the inner product of the two vectors with some adjustments for special cases. As motivation for this coloring, note that each of the three configurations with four colors that survive the modified CFLS coloring (see Figure 43) contains at least one pair of vertices in the intersection of monochromatic neighborhoods of the other three vertices.

For instance, vertices d and e in Figure 50b are in the same monochromatic neighborhood with respect to vertices a, b, and c. Under the MIP coloring, the monochromatic neighborhood of any vertex is contained in an affine plane of \mathbb{F}_q^3 . So, if a, b, and c are linearly independent, then these planes intersect in one point, not two. Therefore, the 50b configuration could only happen under an inner product coloring if the span of a, b, and c has dimension at most 2.

This same idea applies to the other two configurations, and the adjustments to coloring with the inner product are all dedicated to handling the cases for which the five vectors are not in general position. In these cases we frequently end up with three vectors that must all lie on the same affine line, and the offending configurations



FIGURE 45. The intersection of two monochromatic neighborhoods usually lies on an affine line.

would be destroyed if the coloring could be modified to give a proper coloring on every affine line.

1.2. The coloring χ_1 . In Lemma 13.11 we will show that χ_1 induces a proper edge coloring on every line, not just one-dimensional linear spaces but affine lines as well. This will be one of the key lemmas in showing that our construction avoids the remaining configurations. By itself, the inner product almost accomplishes this goal. However, a problem arises when one vector on a given line is orthogonal to the direction of the line. In this case, that particular vector has the same inner product with all other vectors on the line, so we must give these edges new colors. We accomplish this by replacing the inner product with another function.

The first part of χ_1 labels the type of edge-coloring we will have. For two distinct vectors, $x, y \in V$, let T(x, y) be a function defined by

$$T(x,y) = \begin{cases} \text{UP}_1 & x \cdot y = x \cdot x \text{ and } x_1 < y_1 \\ \text{UP}_2 & x \cdot y = x \cdot x, x_1 = y_1, \text{ and } x < y \\ \text{DOWN}_1 & x \cdot y \neq x \cdot x, x \cdot y = y \cdot y, \text{ and } x_1 < y_1 \\ \text{DOWN}_2 & x \cdot y \neq x \cdot x, x \cdot y = y \cdot y, x_1 = y_1, \text{ and } x < y \\ \text{ZERO} & x \cdot y \notin \{x \cdot x, y \cdot y\} \text{ and } x \cdot y = 0 \\ \text{DOT} & \text{otherwise} \end{cases}$$

Here, the categories UP_i and $DOWN_i$ let us know that at least one of the two vectors is orthogonal to the direction of the line between the two, and therefore this edge will need to receive something other than the inner product in the next part of the color. The words UP and DOWN describe the edge from the perspective of the "special" vertex. For instance, if x is orthogonal to the direction of the line it makes with y and x < y, then x looks up the edge to y. The need for different categories when $x_1 = y_1$ is a technical point. The category DOT stands for the inner product (or the "dot" product), and ZERO is the special case where the inner product is zero. The need to split the colors with zero inner product is also a technical point.

Let $f_T(x, y) : \mathbb{F}_q^3 \to \mathbb{F}_q$ be a function defined by

$$f_T(x,y) = \begin{cases} x_1 + y_1 & T \in \{\text{UP}_1, \text{DOWN}_1, \text{ZERO}\} \\ x_2 + y_2 & T \in \{\text{UP}_2, \text{DOWN}_2\} \\ x \cdot y & T = \text{DOT} \end{cases}$$

One final technical point is to differentiate colors based on whether the two vectors are linearly dependent or independent. Let

$$\delta(x,y) = \begin{cases} 0 & \{x,y\} \text{ is linearly dependent} \\ 1 & \{x,y\} \text{ is linearly independent} \end{cases}$$

This is enough to define the coloring. For vertices x < y, let T = T(x, y), and set

$$\chi_1(x,y) = (T, f_T(x,y), \delta(x,y)).$$

1.3. Algebraic definitions and facts. We assume that the reader has some familiarity with basic linear algebra notions such as dimension, linear independence, linear combination, and span. The following definitions and facts are perhaps less familiar. All are reproduced from definitions and propositions in Chapter 2 of the great *Linear Algebra Methods in Combinatorics* book by László Babai and Péter Frankl **3**.

DEFINITION 13.1. Let \mathbb{F}^n be a vector space, and let $S \subseteq \mathbb{F}^n$ be a set of vectors. The rank of S is the dimension of the linear space spanned by S.

FACT 13.2. Let \mathbb{F} be a field, and let A be a $k \times n$ matrix over \mathbb{F} . Then the rank of the set of column vectors as vectors in \mathbb{F}^k is equal to the rank of the set of row vectors as vectors in \mathbb{F}^n . We know this value as the rank of the matrix A, rk(A).

DEFINITION 13.3. Let \mathbb{F}^n be a vector space. An affine combination of vectors $v_1, \ldots, v_k \in \mathbb{F}^n$ is a linear combination $\lambda_1 v_1 + \cdots + \lambda_k v_k$ for $\lambda_1, \ldots, \lambda_k \in \mathbb{F}$ such that $\lambda_1 + \cdots + \lambda_k = 1$. An affine subspace is a subset of vectors that is closed under affine combinations.

FACT 13.4. Any affine subspace U is either empty or the translation of some linear subspace V. That is, each vector $u \in U$ can be written in the form u = v + t where v is some vector in V and t is a fixed translation vector.

DEFINITION 13.5. The dimension dim(U) of an affine subspace U is the dimension of the unique linear subspace of which U is a translate.

DEFINITION 13.6. Let \mathbb{F}^n be a vector space. Let $v_1, \ldots, v_k \in \mathbb{F}^n$. We say that these vectors are affine independent if

$$\lambda_1 v_1 + \dots + \lambda_k v_k = 0$$

implies that

$$\lambda_1 = \dots = \lambda_k = 0$$

for any $\lambda_1, \ldots, \lambda_k \in \mathbb{F}$ for which $\lambda_1 + \cdots + \lambda_k = 0$. Otherwise, these vectors are affine dependent. We say that a set of vectors S is a basis for an affine subspace if they are affine independent and every vector in the subspace is an affine combination of vectors in S.

FACT 13.7. A basis of an affine subspace U contains exactly dim(U) + 1 elements.

FACT 13.8. Let \mathbb{F}^n be some vector space. Let A be a $k \times n$ matrix over \mathbb{F} and $b \in \mathbb{F}^n$. Then the solution set to Ax = b is an affine subspace of dimension n - rk(A).

DEFINITION 13.9. A vector $x \in \mathbb{F}^n$ is isotropic if $x \cdot x = 0$. A linear subspace $U \subseteq \mathbb{F}^n$ is totally isotropic if $x, y \in U$ implies that $x \cdot y = 0$.

FACT 13.10. For any nonzero vector $x \in \mathbb{F}^n$, the set of vectors $\{y : x \cdot y = 0\}$ is a linear subspace of \mathbb{F}^n with dimension n - 1.

1.4. Properties of χ_1 .

LEMMA 13.11. The coloring χ_1 induces a proper edge coloring on every one-dimensional affine subspace.

PROOF. Let $a, b, c \in \mathbb{F}_q^3$ be three distinct vectors in a one-dimensional affine subspace. Then there exists some $\lambda \in \mathbb{F}_q$ such that $c = \lambda a + (1 - \lambda)b$. Suppose towards a contradiction that $\chi_1(a, b) = \chi_1(a, c)$, and let T = T(a, b) = T(a, c). If $T \in \{\text{ZERO, DOT}\}$, then

$$a \cdot b = a \cdot (\lambda a + (1 - \lambda)b).$$

So $\lambda a \cdot (a - b) = 0$. Since $c \neq b$, then $\lambda \neq 0$. Therefore, $a \cdot (a - b) = 0$. But this contradicts the assumption that $T \in \{\text{ZERO}, \text{DOT}\}$.

If $T \in {\text{UP}_1, \text{DOWN}_1}$, then $f_T(a, b) = f_T(a, c)$ gives

$$a_1 + b_1 = a_1 + \lambda a_1 + (1 - \lambda)b_1.$$

So $b_1 = a_1$, a contradiction since $T \in \{\text{UP}_1, \text{DOWN}_1\}$ implies that $a_1 \neq b_1$. Similarly, if $T \in \{\text{UP}_2, \text{DOWN}_2\}$, then $a_2 = b_2$ by the same argument and $a_1 = b_1$ by definition. But then either $a_3(a_3 - b_3) = 0$ or $b_3(b_3 - a_3) = 0$. Both cases imply that $a_3 = b_3$. So a = b, a contradiction. DEFINITION 13.12. Given a vertex $a \in V$ and an edge-color A, let

$$N_A(a) = \{x : \chi_1(a, x) = A\}$$

be the A-neighborhood of a

OBSERVATION 13.13. Given a vector $a \in V$ and a color $A = (T, \alpha, i)$, the vectors in $N_A(a)$ all belong to the two-dimensional affine subspace defined by $\{x : f_T(a, x) = \alpha\}$. In particular, this plane can be defined as the solution space to either $a \cdot x = \alpha$ when T = DOT, $(1,0,0) \cdot x = \alpha - a_1$ when $T \in \{ZERO, UP_1, DOWN_1\}$, and $(0,1,0) \cdot x = \alpha - a_2$ when $T \in \{UP_2, DOWN_2\}$.

In certain cases, we can actually say something a little stronger. First, note that if $T(a, x) \in \{\text{UP}_1, \text{DOWN}_1\}$, then we will have $a_1 < x_1$ if, and only if, a < x. Therefore, if $f_T(a, x) = \alpha$, since $f_T(a, x) = a_1 + x_1$, we will have $a_1 < \alpha = a_1$ if, and only if, a < x.

LEMMA 13.14. Given a vector $a \in V$, and a color $A = (T, \alpha, i)$, the vectors of $N_A(a)$ all belong to a one-dimensional affine subspace if one of the following three cases holds for all $x \in N_A(a)$:

- (1) $T \in \{ZERO, UP_2, DOWN_2\};$
- (2) $T = UP_1$ and a < x;
- (3) $T = DOWN_1$ and a > x.

PROOF. In the first case, if T = ZERO, then every $x \in N_A(a)$ must satisfy the system of linear equations

$$a \cdot x = 0$$
$$(1, 0, 0) \cdot x = \alpha - a_1.$$

Since a contains no zero components, then the rank of $\{a, (1, 0, 0)\}$ is two. Therefore, the solution space must be a one-dimensional affine subspace. If $T \in \{UP_2, DOWN_2\}$,


FIGURE 46. The intersection of two monochromatic neighborhoods.

then every $x \in N_A(a)$ must satisfy the system

$$(1, 0, 0) \cdot x = a_1$$

 $(0, 1, 0) \cdot x = \alpha - a_2.$

Since (1, 0, 0) and (0, 1, 0) are linearly independent, then, as before, the set of solutions is a one-dimensional affine subspace.

In each of the other two cases, we see that every $x \in N_A(a)$ must satisfy the system

$$a \cdot x = a \cdot a$$
$$(1, 0, 0) \cdot x = \alpha - a_1.$$

As before, the solution space must be a one-dimensional affine subspace.

Therefore, we immediately get the following corollary by Lemma 13.11

COROLLARY 13.15. Let $a, b, c, d \in V$ be four distinct vertices such that

$$\chi_1(a,b) = \chi_1(a,c) = \chi_1(a,d) = (T,\alpha,i).$$

The set of vertices $\{b, c, d\}$ span three distinct edge colors under χ_1 if any of the following are true:

- (1) $T \in \{ZERO, UP_2, DOWN_2\};$
- (2) $T = UP_1$ and a < b, c, d;
- (3) $T = DOWN_1$ and a > b, c, d.

LEMMA 13.16. Let $a, b, c, d, e \in V$ be vectors such that $\{a, b\}$ is linearly independent, $\chi_1(a, c) = \chi_1(a, d) = \chi_1(a, e)$, and $\chi_1(b, c) = \chi_1(b, d) = \chi_1(b, e)$ (see Figure 46). Then the set $\{c, d, e\}$ spans three distinct edge colors.

PROOF. Let $\chi_1(a, c) = \chi_1(a, d) = \chi_1(a, e) = A$ and $\chi_1(b, c) = \chi_1(b, d) = \chi_1(b, e) = B$. The result is immediate if either pair (a, A) or (b, B) satisfies the conditions listed in Corollary 13.15 So assume not. If $A = (T_a, \alpha, i)$, then by Observation 13.13 we know that c, d, and e must either satisfy $a \cdot x = \alpha$ or $(1, 0, 0) \cdot x = \alpha - a_1$. Similarly, if $B = (T_b, \beta, j)$, then c, d, and e must either satisfy $b \cdot x = \beta$ or $(1, 0, 0) \cdot x = \beta - b_1$. Since the sets $\{a, b\}$, $\{a, (1, 0, 0)\}$, and $\{(1, 0, 0), b\}$ are all linearly independent, then every case gives us the result immediately except when $T_a, T_b \in \{\text{UP}_1, \text{DOWN}_1\}$. Since we assume that none of the cases from Corollary 13.15 hold, then this can only happen when $x \cdot (x - a) = x \cdot (x - b) = 0$ for x = c, d, e. In this case, c, d, and e all satisfy the two linear equations,

$$(a-b) \cdot x = 0$$
$$(1,0,0) \cdot x = \alpha - a_1$$

Hence, c, d, and e are affine independent, and the result follows from Lemma 13.11 unless

$$a_2 - b_2 = a_3 - b_3 = 0$$

But if this is true, then $c \cdot (c-a) = c \cdot (c-b)$ implies that a = b, a contradiction.

2. Combining the colorings

Let $n = (q-1)^3$ where q is an odd prime power. To each $\alpha \in \mathbb{F}_q$ we associate the unique element $\alpha' \in \{0,1\}^{\lceil \log q \rceil}$ which represents in binary the rank of α under the linear order given to the elements of \mathbb{F}_q in Section 1. Let β be the minimum positive integer for which

$$3\left\lceil \log q \right\rceil \le \beta^2.$$

We associate each of the *n* vertices of K_n with a unique vector in $(\mathbb{F}_q^*)^3$ as in Section 1. To each vertex (x_1, x_2, x_3) , we associate $(x'_1, x'_2, x'_3, 0) \in \{0, 1\}^{\beta^2}$ as well, where for each *i*, x'_i is the binary representation of the rank of x_i , and 0 denotes a string of $\beta^2 - 3 \lceil \log q \rceil$ zeros. Let

$$C = \varphi \times \chi_1.$$

Since

$$\beta = \Theta\left(\sqrt{3\log q}\right) = \Theta\left(\sqrt{\log n}\right),$$

it follows that the number of colors used in this combined coloring is at most

$$12q\beta 2^{3\beta} = n^{1/3} 2^{O\left(\sqrt{\log n}\log\log n\right)}$$

colors. This bound on the number of colors generalizes to all n by the standard density of primes argument [37].

2.1. The first two configurations.

LEMMA 13.17. Any distinct vertices $a, b, c, d, e \in V$ for which $C(a, c) = C(a, d) = C(a, e), C(b, c) = C(b, d) = C(b, e), and <math>C(a, c) \neq C(b, c)$ (see Figure 46) span at least five distinct edge colors.

PROOF. Lemma 12.1 implies that neither color between $\{a, b\}$ and $\{c, d, e\}$ can be repeated on the edges spanned by $\{c, d, e\}$. Therefore, if $\{a, b\}$ is linearly independent it follows from Lemma 14.1 that $\{a, b, c, d, e\}$ span at least 5 colors.

Otherwise, $b = \lambda a$ for some $\lambda \in \mathbb{F}_q$. If C(a, b) repeats one of the colors from the edges spanned by $\{c, d, e\}$, then this gives us the configuration forbidden by Lemma 12.2. If C(a, b) = C(a, c) or C(a, b) = C(b, c), then all five vectors belong to a one-dimensional linear subspace spanned by a which must be properly edge-colored by Lemma 13.11. Therefore, the set of vertices $\{a, b, c, d, e\}$ spans at least 5 colors. This immediately shows that the first configuration will not appear under the combined coloring.

COROLLARY 13.18. Let $a, b, c, d, e \in V$ be five distinct vertices. It cannot be the case that C(a, b) = C(a, c) = C(a, d) = C(a, e), C(b, c) = C(b, d) = C(b, e), and C(c, d) = C(c, e) as in Figure 50b.

The second configuration also will not appear under the combined coloring.

LEMMA 13.19. Let $a, b, c, d, e \in V$ be five distinct vertices. It cannot be the case that

$$C(a,c) = C(a,d) = C(a,e) = C(b,c) = C(b,d) = C(b,e),$$

and C(c,d) = C(c,e) as in Figure 43b.

PROOF. By Lemma 14.1, this can happen only if there exists some $\lambda \in \mathbb{F}_q$ such that $b = \lambda a$. In this case,

$$\chi_1(a,c) = \chi_1(a,d) = \chi_1(a,e) = \chi_1(\lambda a,c) = \chi_1(\lambda a,d) = \chi_1(\lambda a,e).$$

If this color is in DOT, then $c \cdot a = c \cdot \lambda a$. So either $\lambda = 1$, a contradiction, or $c \cdot a = 0$, a contradiction that the color is in DOT. If the color is not in DOT, then it must be the case that $a_1 = \lambda a_1$. Since $a_1 \neq 0$, then this forces $\lambda = 1$, a contradiction.

3. Splitting the coloring

Now we will split the colors of C to make a new coloring $C' = C \times \chi_2$, where χ_2 is the second part of the MIP coloring.

Let $U \subseteq \mathbb{F}_q^3$ be a two-dimensional linear subspace. Let G_U be an auxiliary graph where $V(G_U)$ is the set of non-isotropic vectors in U, and

$$xy \in E(G_U) \iff x \cdot y = 0.$$

We wish to show that G_U is bipartite. Note that $x \cdot y = 0$ implies that $\alpha x \cdot \beta y = 0$ for any $\alpha, \beta \in \mathbb{F}_q$. Suppose that $x \cdot z = 0$ for some $z \in V(G_U)$ such that $z \neq \beta y$ for any $\beta \in \mathbb{F}_q$. Then the intersection between U and the two-dimensional linear subspace orthogonal to x must also be a two-dimensional linear subspace. Therefore, x is contained in its own orthogonal linear subspace. So x is isotropic, a contradiction. Hence, G_U is comprised of disjoint complete bipartite graphs and so is itself bipartite.

For each two-dimensional linear subspace U, we label the vertices of G_U with A_U and B_U depending on their part in the bipartition, and then label all isotropic vectors in U with A_U as well.

For any two-dimensional linear subspace $U\subseteq \mathbb{F}_q^3$ and any $x\in U$ we define

$$S(x,U) = \begin{cases} A & x \in A_U \\ B & x \in B_U \end{cases}$$

For a given vector $a \in V$, and a given color type T, define

$$a_{T} = \begin{cases} a & T = \text{DOT} \\ (1, 0, 0) & T \in \{\text{UP}_{1}, \text{DOWN}_{1}, \text{ZERO}\} \\ (0, 1, 0) & T \in \{\text{UP}_{2}, \text{DOWN}_{2}\} \end{cases}$$

and let

$$U_{a,T} = \{x : a_T \cdot x = 0\}.$$

For convenience, let

$$a_b = \begin{cases} 0 & a_T \cdot a_T = 0\\ (a_T \cdot b)(a_T \cdot a_T)^{-1}a_T & a_T \cdot a_T \neq 0 \end{cases}$$

for any vectors a and b where T = T(a, b).

Now we can define the second part of the MIP coloring. For any two vectors, a < bwith T = T(a, b), let

$$\chi_2(a,b) = (S(a - b_a, U_{b,T}), S(b - a_b, U_{a,T})).$$

3.1. The third configuration. Let $a, b, c, d, e \in V$ be five distinct vertices such that

$$C'(a,b) = C'(a,c) = C'(a,d) = C'(a,e) = \text{Black},$$

and let

$$C'(b,c) = C'(c,d) = C'(d,e) = C'(e,b) = \text{Red}$$

as shown in Figure 43c. By Lemma 12.7 we know that either b, d < c, e or c, e < b, d. Similarly, we know that either a < b, c, d, e or b, c, d, e < a. So without loss of generality, we can say that either a < b, d < c, e or b, d < c, e < a. In either case,

$$S(b - a_b, U_{a,T}) = S(c - a_c, U_{a,T})$$

where T = T(a, b) = T(a, c).

By Corollary 13.15 we know that one of the following three cases must be true:

- (1) Black \in DOT,
- (2) Black \in UP₁ such that b, d < c, e < a, or
- (3) Black \in DOWN₁ such that a < b, d < c, e.

This abuses our notation slightly, but the meaning is hopefully clear. For example, Black \in DOT means that the first component of the χ_1 part of the color Black is DOT.

We will show that none of these cases are possible through the following series of lemmas.

LEMMA 13.20. If $Black \in DOT$ and $RED \in DOT$, then the configuration in Figure 43c is not possible under the coloring C'. PROOF. Let the inner product part of color Black be α and the inner product part of Red be β . Note that if either Black or Red encodes linear independence, then b, c, d, and e would all belong to the same one-dimensional linear subspace, a contradiction of Lemma 13.11. Also, since c - e satisfies the three linear equations, $a \cdot x = 0, b \cdot x = 0$, and $d \cdot x = 0$, then $\{a, b, d\}$ cannot be linearly independent since then c = e, a contradiction. So there exist nonzero $\lambda_1, \lambda_2 \in \mathbb{F}_q$ such that $d = \lambda_1 a + \lambda_2 b$.

Note that $a \cdot a \neq 0$ since otherwise

$$a \cdot d = a \cdot (\lambda_1 a + \lambda_2 b)$$
$$\alpha = \lambda_2 \alpha$$

implies that $\lambda_2 = 1$ since $\alpha \neq 0$. If $\lambda_2 = 1$, then we would reach a contradiction by taking the inner product of both sides of $d = \lambda_1 a + b$ with c to get that $\beta = \lambda_1 \alpha + \beta$, a contradiction since $d \neq b$.

So $a_b = a_c = \alpha (a \cdot a)^{-1} a$, then

$$d = \lambda_1' a_b + \lambda_2 b$$

where $\lambda'_1 = \lambda_1 \alpha^{-1} (a \cdot a)$. Taking the inner product of both side of this with *a* gives that

$$\alpha = (\lambda_1' + \lambda_2)\alpha.$$

So it follows that a_b , b, and d are affine dependent. By the same arguments we can conclude that a_b , c, and e are also affine dependent.

Note that $(b-d) \cdot (c-e) = 0$. Therefore, $(b-a_b) \cdot (c-a_c) = 0$. Since

$$S\left(b-a_{b}, U_{a,\text{DOT}}\right) = S\left(c-a_{c}, U_{a,\text{DOT}}\right),$$

then $b-a_b$ and $c-a_c$ are contained in the same part of the bipartition of the auxiliary graph on $U_{a,\text{DOT}}$. Therefore, either $b-a_b$ or $c-a_c$ must be isotropic since otherwise the fact that they are orthogonal would have made them adjacent in the auxiliary graph. Assume without loss of generality that $b - a_b$ is isotropic. Since

$$(b-a_b)\cdot(b-a_b)=0,$$

then $b \cdot b = \alpha^2 (a \cdot a)^{-1}$. Since $(b - a_b) \cdot (c - a_c) = 0$, then $\beta = \alpha^2 (a \cdot a)^{-1}$. Therefore, $b \cdot b = \beta$. Hence,

$$b \cdot (b - c) = 0.$$

This contradicts our assumption that $\text{Red} \in \text{DOT}$.

Note in what follows that if Red \notin DOT, then $b_1 = d_1$ and $c_1 = e_1$.

LEMMA 13.21. If $Black \in DOT$ and $RED \in ZERO$, then the configuration in Figure 43c is not possible under the coloring C'.

PROOF. If Red \in ZERO, then

$$b \cdot (c - e) = d \cdot (c - e) = 0.$$

Also, recall that

$$a \cdot (c - e) = 0.$$

If a, b, d are linearly independent, then c = e, a contradiction. So we must assume that a, b, d are linearly dependent.

If either b or d depends on a, then $\delta(a, x) = 0$ for x = b, c, d, e which implies that all five vectors belong to a one-dimensional linear subspace spanned by a, contradicting Lemma 13.11. If $d = \lambda b$ for some $\lambda \in \mathbb{F}_q$, then $b_1 = d_1 = \lambda b_1$. So either $b_1 = 0$ or $\lambda = 1$, both contradictions. So we must assume that $d = \lambda_1 a + \lambda_2 b$ for nonzero $\lambda_1, \lambda_2 \in \mathbb{F}_q$. But then

$$d \cdot c = \lambda_1(a \cdot c) + \lambda_2(b \cdot c)$$
$$0 = \lambda_1(a \cdot c)$$

Since $\lambda_1 \neq 0$, then $a \cdot c = 0$ which implies that Black \notin DOT, a contradiction.

LEMMA 13.22. If $Black \in DOT$ and $RED \notin \{DOT, ZERO\}$, then the configuration in Figure 43c is not possible under the coloring C'.

PROOF. If Red \in UP₂ \cup DOWN₂, then $b_1 = c_1 = d_1 = e_1$. So all four vectors b, c, d, e satisfy the linear equations $a \cdot x = \alpha$ and $(1, 0, 0) \cdot x = b_1$. Therefore, b, c, d, and e all belong to a one-dimensional affine subspace, a contradiction of Lemma 13.11. If Red \in UP₁, then

$$b \cdot (b - c) = b \cdot (b - e) = d \cdot (d - c) = d \cdot (d - e) = 0$$

since we assume that b, d < c, e. Therefore,

$$b \cdot (c - e) = b \cdot c - b \cdot e = b \cdot b - b \cdot b = 0.$$

Similarly, $d \cdot (c - e) = 0$. Since $c_1 = e_1$, then it follows that

$$\begin{pmatrix} a_1 & a_2 & a_3 \\ b_1 & b_2 & b_3 \\ b_1 & d_2 & d_3 \end{pmatrix} \begin{pmatrix} 0 \\ c_2 - e_2 \\ c_3 - e_3 \end{pmatrix} = 0.$$

Therefore, if any two of (a_2, a_3) , (b_2, b_3) , and (d_2, d_3) are linearly independent as vectors in \mathbb{F}_q^2 , then c = e, a contradiction. Hence, there must exist $\lambda_1, \lambda_2 \in \mathbb{F}_q$ such that $(a_2, a_3) = \lambda_1(b_2, b_3)$ and $(d_2, d_3) = \lambda_2(b_2, b_3)$.

From the equations of the form $x \cdot (x - c) = 0$ for x = b, d we get

$$b_1(b_1 - c_1) + b_2^2 + b_3^2 - b_2c_2 - b_3c_3 = 0$$
$$b_1(b_1 - c_1) + \lambda_2^2b_2^2 + \lambda_2^2b_3^2 - \lambda_2b_2c_2 - \lambda_2b_3c_3 = 0$$

So it follows that

$$c_3 = b_3^{-1} \left(b_1 (b_1 - c_1) + b_2^2 + b_3^2 - b_2 c_2 \right)$$

which in turn gives that

$$(1 - \lambda_2)b_1(b_1 - c_1) = \lambda_2(1 - \lambda_2)(b_2^2 + b_3^2).$$

Since $\lambda_2 \neq 1$, then

$$b_1(b_1 - c_1) = \lambda_2(b_2^2 + b_3^2).$$

Since $b_1 \neq 0$ and $b_1 \neq c_1$, then $b_2^2 + b_3^2 \neq 0$.

Now, since $Black \in DOT$, we get

$$a \cdot b = a \cdot d$$
$$a_1b_1 + \lambda_1b_2^2 + \lambda_1b_3^2 = a_1b_1 + \lambda_1\lambda_2b_2^2 + \lambda_1\lambda_2b_3^2$$
$$\lambda_1(1 - \lambda_2)(b_2^2 + b_3^2) = 0$$

But this is a contradiction, since none of these three terms are zero.

If $\text{Red} \in \text{DOWN}_1$, then we swap b, d and e, c in the previous argument to obtain the same contradiction.

LEMMA 13.23. If Black \notin DOT, then the configuration in Figure 43c is not possible under the coloring C'.

PROOF. In this case, either b, d < c, e < a and Black \in UP₁, or c, e > b, d > aand Black \in DOWN₁. In both cases,

$$b \cdot (b-a) = c \cdot (c-a) = d \cdot (d-a) = e \cdot (e-a) = 0.$$

Moreover, $b_1 = c_1 = d_1 = e_1$, which implies that

$$\text{Red} \in \text{ZERO} \cup \text{UP}_2 \cup \text{DOWN}_2 \cup \text{DOT}.$$

If $\text{Red} \in \text{ZERO}$, then

$$b \cdot (c - e) = d \cdot (c - e) = 0.$$

Therefore,

$$\begin{pmatrix} b_2 & b_3 \\ d_2 & d_3 \end{pmatrix} \begin{pmatrix} c_2 - e_2 \\ c_3 - e_3 \end{pmatrix} = 0.$$

So $(d_2, d_3) = \gamma(b_2, b_3)$ for some $\gamma \in \mathbb{F}_q$, and $b \cdot c = d \cdot c$ gives

$$b_1^2 + c_2 b_2 + c_3 b_3 = b_1^2 + \gamma c_2 b_2 + \gamma c_3 b_3.$$

Thus, $(1 - \gamma)(c_2b_2 + c_3b_3) = 0$. Therefore, either $\gamma = 1$, a contradiction since $b \neq d$, or $c_2b_2 + c_3b_3 = 0$, also a contradiction since this implies that $b \cdot c = b_1^2 \neq 0$.

If Red \in UP₂ \cup DOWN₂, then we have $b_2 = d_2$, $c_2 = e_2$, and either $b \cdot (b - c) = b \cdot (b - e) = 0$ or $c \cdot (c - b) = c \cdot (c - d) = 0$. In the first case, $b \cdot (e - c) = 0$ so $b_3(e_3 - c_3) = 0$. So either $b_3 = 0$ or $e_3 = c_3$, both contradictions. Similarly, in the second case, $c \cdot (b - d) = 0$ means that $c_3(b_3 - d_3) = 0$, which gives the same contradictions.

Finally, if Red \in DOT, then $b \cdot (c - e) = 0$ and $d \cdot (c - e) = 0$. So

$$\begin{pmatrix} b_2 & b_3 \\ d_2 & d_3 \end{pmatrix} \begin{pmatrix} c_2 - e_2 \\ c_3 - e_3 \end{pmatrix} = 0.$$

Therefore, either c = e, a contradiction, or $(d_2, d_3) = \lambda(b_2, b_3)$ for some nonzero $\lambda \in \mathbb{F}_q$.

If β is the inner product represented by Red, then we get that

$$\beta = b_1^2 + b_2 c_2 + b_3 c_3$$
$$\beta = b_1^2 + \lambda b_2 c_2 + \lambda b_3 c_3$$

So,

$$(1-\lambda)(b_2c_2+b_3c_3) = 0.$$

Therefore, either $\lambda = 1$ or $b_2c_2 + b_3c_3 = 0$. If $\lambda = 1$, then b = d, a contradiction. So we must assume that $b_2c_2 + b_3c_3 = 0$. But then

$$(b - (b_1, 0, 0)) \cdot (c - (b_1, 0, 0)) = 0.$$

Since $a_b = a_c = (b_1, 0, 0)$ we know that

$$(b-a_b)\cdot(c-a_c)=0.$$

Therefore, since

$$S(b - a_b, U_{a,T}) = S(c - a_c, U_{a,T}),$$

it must be that either $(0, b_2, b_3)$ or $(0, c_2, c_3)$ is isotropic.

First, assume that $(0, b_2, b_3)$ and $(0, c_2, c_3)$ are linearly independent. Then they must span the linear subspace $U_{a,T}$. Without loss of generality assume that $(0, b_2, b_3)$ is isotropic. Therefore, it is orthogonal to every vector in the subspace $U_{a,T}$. Since this space is defined to be orthogonal to (1, 0, 0), then this means that $(0, b_2, b_3)$ is linearly dependent on (1, 0, 0), a contradiction.

So we must assume that $(0, b_2, b_3)$ and $(0, c_2, c_3)$ are linearly dependent. Since at least one of them is isotropic, then they belong to a totally isotropic one-dimensional linear subspace. So $b_2^2 + b_3^2 = 0$ and $c = (b_1, \lambda b_2, \lambda b_3)$ for some $\lambda \in \mathbb{F}_q$. But then $b \cdot (b-c) = 0$, a contradiction of the assumption that Red \in DOT.

Since these lemmas show that the third and final configuration does not appear, the coloring C' is a (5,5)-coloring of K_n .

CHAPTER 14

A (5,6)-coloring construction.

In this chapter, we improve the probabilistic upper bound of $f(n, 5, 6) \leq cn^{3/5}$ by giving an explicit (5, 6)-coloring of K_n that uses few colors. The new upper bound comes close to matching the best known lower bound in order of magnitude.

THEOREM 14.1. As $n \to \infty$,

$$\left(\frac{5}{6}n - \frac{95}{144}\right)^{1/2} \le f(n, 5, 6) \le n^{1/2} 2^{O\left(\sqrt{\log n} \log \log n\right)}$$

The lower bound comes from the following lemma, a generalization of an argument used by Erdős and Gyárfás [25], and stated explicitly as equation 11 in [15].

LEMMA 14.1. Let t = f(n, p, q), then

$$f\left(\left\lceil \frac{n-1}{t} \right\rceil, p-1, q-1\right) \le t.$$

PROOF. Suppose we have a (p,q)-coloring of K_n with t colors. Fix some vertex x, then at least $\left\lceil \frac{n-1}{t} \right\rceil$ vertices must appear in a monochromatic neighborhood of x. The number of colors t must be enough to give a (p-1, q-1)-coloring on this set.

Erdős and Gyárfás showed that $\frac{5}{6}(n-1) \leq f(n,4,5)$ [25]. This, combined with the lemma, gives the stated lower bound in Theorem [14.1].

The construction providing the upper bound combines two existing constructions with some modification. The first is the modified CFLS construction given in Chapter 12. The second construction is the "algebraic" part of the (4, 4)-coloring given by Mubayi in [35].

After defining the construction in the next section, we will demonstrate that it avoids many different configurations of colored edges on five or fewer vertices. By ruling these cases out, the algorithm in Appendix $\overline{\mathbf{A}}$ is used to show that no copy of K_5 can span fewer than six distinct colors.

1. The algebraic construction

We will now define the algebraic part of the construction, $\zeta = \zeta_1 \times \zeta_2$. The first part of this construction, ζ_1 , is exactly the algebraic part of the (4, 4)-coloring given by Mubayi [35]. The second part, ζ_2 , is a modification original to this chapter but based on a similar modification used to alter the algebraic portion of the (5, 5)-coloring in Chapter 13.

Let $n = q^2$ where q is some odd prime power. Associate each vertex of K_n with a unique vector in the space \mathbb{F}_q^2 over the finite field with q elements. Between any two vectors $x = (x_1, x_2)$ and $y = (y_1, y_2)$, we define the color ζ_1 of the edge between them as

$$\zeta_1(xy) = (x_1y_1 - x_2 - y_2, \delta(x_1, y_1))$$

where

$$\delta(x_1, y_1) = \begin{cases} 0 & x_1 = y_1 \\ 1 & x_1 \neq y_1 \end{cases}$$

Here, all algebraic operations are taken to be the standard ones defined by the finite field.

The modification to this coloring, ζ_2 , requires that we give the elements of \mathbb{F}_q^2 some linear order. When we combine the algebraic part of the coloring with the modified CFLS coloring, this order will agree with the order put on the binary strings, but for now we just assume that there is some linear order.

For each element $\alpha \in \mathbb{F}_q$ let G_α be the graph with vertex set $\mathbb{F}_q \setminus \{\alpha\}$ such that

$$xy \in E(G) \iff x+y=2\alpha.$$

For two distinct elements, $\alpha, \beta \in \mathbb{F}_q$, define the function

$$f_{\alpha}(\beta) = \begin{cases} S & \beta \in S_{\alpha} \\ T & \beta \in T_{\alpha} \end{cases}$$

Now we can define ζ_2 for two vectors, x < y, as

$$\zeta_2(xy) = (f_{x_1}(y_1), f_{y_1}(x_1)).$$

The coloring, ζ_1 , gives at most 2q colors on q^2 vertices, and the modification, ζ_2 , gives four colors. So overall the modified algebraic coloring ζ uses at most $8q = 8\sqrt{n}$ colors.

2. Combining the constructions

Begin with $n = q^2$ for some odd prime power q, and associate each vertex with a distinct vector of \mathbb{F}_q^2 as in the previous section. Give some linear order for the elements of the base field, \mathbb{F}_q . To each $\alpha \in \mathbb{F}_q$ we associate the unique element $\alpha' \in \{0, 1\}^{\lceil \log q \rceil}$ which represents in binary the rank of α under the this linear order. Let β be the minimum positive integer for which

 $2\left\lceil \log q \right\rceil \le \beta^2.$

To a vertex of K_n associated with vector $(x_1, x_2) \in \mathbb{F}_q^2$, we also associate the binary string $(x'_1, x'_2, 0) \in \{0, 1\}^{\beta^2}$ where for each i, x'_i is the binary representation of the rank of x_i , and 0 denotes a string of $\beta^2 - 2 \lceil \log q \rceil$ zeros.

The edge-coloring $\varphi \times \zeta$ is then given by applying ζ to the vectors and φ to the binary strings. Since

$$\beta = \Theta\left(\sqrt{2\log q}\right) = \Theta\left(\sqrt{\log n}\right),$$



FIGURE 47. Four configurations that each contain a forbidden "color cycle."

it follows that the number of colors used in this combined coloring is at most

$$8a\beta 2^{2\beta} = n^{1/2} 2^{O\left(\sqrt{\log n}\log\log n\right)}$$

colors. This upper bound on the number of colors generalizes to all n by the standard density of primes argument [35, 37].

3. Configurations avoided by CFLS

In Chapter 12, we showed that the modified CFLS coloring, φ , avoids several possible configurations of edge colors on small cliques. Several of these cases, including monochromatic odd cycles, are covered by Lemma 14.2.

3.1. General "color cycle" configurations. Let p and q be positive integers. Assume that we have a copy of K_p under an edge-coloring

$$c: E(K_p) \to \{C_1, \ldots, C_q\}.$$

Define an auxiliary digraph D on the set of edge colors, $V(D) = \{C_1, C_2, \ldots, C_q\}$, such that $C_i \to C_j \in E(D)$ if and only if there exist vertices $v_1, \ldots, v_k \in V(K_p)$ for $k \ge 3$ such that

$$c(v_1v_2) = c(v_2v_3) = \dots = c(v_{k-1}v_k) = C_i$$

and $c(v_k v_1) = C_j$.

Now, color the directed edges of D "Odd" or "Even" depending on the parity of the number of vertices k that gives the directed edge. Note that multiedges with different parities are possible in D.

LEMMA 14.2. The CFLS coloring avoids any edge-colored copy of K_p for which the auxiliary digraph on the colors spanned by that clique contains a directed cycle with at least one Odd edge.

PROOF. Suppose that colors C_1, \ldots, C_m make such a directed cycle:

$$C_1 \to C_2, C_2 \to C_3, \ldots, C_m \to C_1 \in E(D)$$

such that (without loss of generality) $C_1 \to C_2$ is colored Odd. Then there exist vertices $v_1, \ldots, v_k \in V(K_p)$ for some odd integer $k \ge 3$ for which $\varphi_1(v_i v_{i+1}) = C_1$ for $i = 1, \ldots, k - 1$ and $\varphi_1(v_k v_1) = C_2$. Let the zero coordinate of the color C_1 be $(i, \{x, y\})$. Without loss, assume that $v_1^{(i)} = x$. It follows that $v_k^{(i)} = x$ as well. Therefore, the *i*th coordinate of C_2 is zero.

Now, each subsequent directed edge $C_j \to C_{j+1}$ for $j = 2, \ldots, m$, regardless of color, forces the *i*th coordinate of the "head" color to be zero as well. To see this assume that C_j is zero in its *i*th coordinate. A monochromatic path in color C_j , $u_1u_2 \cdots u_l$, implies that u_1 agrees with u_l at *i*. Therefore, the *i*th coordinate of C_{j+1} must also be zero. The same must be true for C_1 since this is a directed cycle, a contradiction.

For the (5, 6)-coloring we use Lemma 14.2 to eliminate the configurations shown in Figure 47 as well as monochromatic odd cycles.

3.2. Configurations containing a monochromatic P_3 . Assume that we have an edge-colored K_5 that contains a monochromatic P_3 on vertices *abcd* in color Black. The edges *ac* and *bd* cannot be Black since color classes are bipartite in CFLS. So we color edge *ac* Red. Let Black₀ = $(i, \{x, y\})$, then Red_i = 0. Therefore, any Red edge from the Black P_3 to vertex *e* fixes the value of $e^{(i)}$ as either *x* or *y*. CFLS would then forbid any third color from having an edge between an *x* and a *y* as well as one between two vertices that agree at *i*.



FIGURE 48. CFLS forbids these 16 configurations, each contain a monochromatic P_3 .

There are 16 possible configurations that fit this description when we do not consider those in which the third color never touches vertex e (these cases are summarized separately). Figure 48 presents these 16 configurations.

3.3. Configurations containing an alternating C_4 . Next, consider configurations that contain a 2-colored C_4 with alternating colors. If the configuration also has two same-colored edges adjacent at the fifth vertex so that the other two endpoints are each incident to either endpoint of an edge of the C_4 (as shown by the edge-colored cliques in Figure 49), then we can say that under CFLS, the color from the fifth vertex must be distinct from any color spanned by the other four vertices. Moreover, none of these spanned colors can be incident with the fifth vertex.

LEMMA 14.3. Let a, b, c, d, e be distinct vertices such that $\varphi(ab) = \varphi(cd) = \alpha$, $\varphi(bc) = \varphi(ad) = \beta$, $\varphi(ae) = \varphi(de) = \gamma$, $\varphi(ac) = \pi_1$, $\varphi(bd) = \pi_2$, $\varphi(be) = \lambda_1$, and



FIGURE 49. Configurations containing an alternating C_4 .

 $\varphi(ce) = \lambda_2$. Then

$$\gamma, \lambda_1, \lambda_2 \not\in \{\alpha, \beta, \pi_1, \pi_2\}.$$

PROOF. Let $\beta_0 = (i, \{x, y\})$. Without loss of generality, we may assume that $a^{(i)} = x$ and $d^{(i)} = y$. If $e^{(i)} = x$ or $e^{(i)} = y$, then we get that $\gamma_i = 0$ and $\gamma_i \neq 0$, a contradiction. Hence,

$$e^{(i)} = z \notin \{x, y\}.$$

This alone shows that $\gamma \neq \beta$.

If $\gamma = \alpha$, then $\alpha_i \neq 0$ so it must be the case that $b^{(i)} = y$ and $c^{(i)} = x$. Therefore, three distinct binary strings, x, y, z, pairwise have the same first index of difference α_i . This is impossible since two must either both be zero or both be one.

If $\gamma = \pi_1$, then $c^{(i)} = y$ and so $b^{(i)} = x$. Hence, the distinct binary strings x, y, zagain pairwise have the same first index of difference, γ_i , a contradiction. The same argument applies if $\gamma = \pi_2$.

Next, assume that $\lambda_1 = \pi_1$. Since $e^{(i)} = z$ and $b^{(i)} \in \{x, y\}$, then λ_1 is nonzero at *i*. Since $\varphi(ac) = \lambda_1$ and $a^{(i)} = x$, then $c^{(i)} = y$ and so $b^{(i)} = x$. Hence, colors γ and λ_1 must agree in coordinate *i*. This again gives us that x, y, z all pairwise differ at the same first index, a contradiction. The same argument applies if $\lambda_2 = \pi_2$.

If $\lambda_1 = \pi_2$, then $b^{(i)} = x$ and so $c^{(i)} = y$. So again γ and λ_1 agree at coordinate *i*. This again forces the contradiction with x, y, z. The same argument applies if $\lambda_2 = \pi_1$.



FIGURE 50. Five additional configurations avoided by φ .

Finally, note that $\lambda_1, \lambda_2 \neq \beta$ since $e^{(i)} = z$. If $\lambda_1 = \alpha$, then $b^{(i)} = y$ and so γ and α agree at coordinate *i* which gives the same contradiction as before. The same reasoning applies if $\lambda_2 = \alpha$.

3.4. Additional configurations. We finish this section by showing that the modified CFLS coloring eliminates the additional cases shown in Figure 50. We already showed in Chapter 12 that this coloring eliminates Figures 50a and 50b.

LEMMA 14.4. Let a, b, c, d, e be distinct vertices. The CFLS coloring forbids $\varphi(ab) = \varphi(bc) = \varphi(de) = \alpha$ and $\varphi(cd) = \varphi(ae) = \beta$ (see Figure 50c).

PROOF. Assume that this can happen and that $\alpha_0 = (i, \{x, y\})$. Without loss of generality, assume that $b^{(i)} = x$ and $a^{(i)} = c^{(i)} = y$. Moreover, without loss we can assume that $d^{(i)} = x$ and $e^{(i)} = y$. Then $a^{(i)} = e^{(i)}$ implies that $\beta_i = 0$. But $c^{(i)} \neq d^{(i)}$ implies that $\beta_i = 0$, a contradiction.

LEMMA 14.5. Let a, b, c, d, e be distinct vertices. The CFLS coloring forbids $\varphi(ab) = \varphi(cd) = \alpha$, $\varphi(ae) = \varphi(bc) = \beta$, and $\varphi(ac) = \varphi(de) = \gamma$ (see Figure 50d).

PROOF. Let $\gamma_0 = (i, \{x, y\})$. Without loss of generality we may assume that $a^{(i)} = x$ and $c^{(i)} = y$. If $d^{(i)} = y$ and $e^{(i)} = x$, then $\alpha_i = \beta_i = 0$ and so $b^{(i)} = x$ and $b^{(i)} = y$, a contradiction. Hence, $d^{(i)} = x$ and $e^{(i)} = y$. So $b^{(i)} = z \notin \{x, y\}$. Moreover, $\alpha_i = \beta_i$ both go between x and y. Therefore, the three distinct binary strings x, y, z are all pairwise different at the same first index, α_i , a contradiction.

LEMMA 14.6. It cannot be the case that a, b, c, d, e are distinct vertices such that $\varphi(ab) = \varphi(cd) = \alpha, \ \varphi(bc) = \varphi(ad) = \beta, \ \varphi(ae) = \varphi(ac) = \gamma, \ and \ \varphi(bd) = \varphi(bc) = \pi$ (see Figure 50e).

PROOF. Let $\pi_0 = (i, \{x, y\})$. Without loss of generality we may assume that $e^{(i)} = x$ and $c^{(i)} = y$. It follows from color γ that $a^{(i)} \notin \{x, y\}$. Since $d^{(i)} \in \{x, y\}$, then it follows that $\beta_i \neq 0$. Hence, $b^{(i)} = x$ and so $d^{(i)} = y$. So $\varphi(cd) = \alpha$ implies that $\alpha_i = 0$, but $\varphi(ab) = \alpha$ implies that $\alpha_i \neq 0$, a contradiction.

4. Configurations avoided by the algebraic coloring

We begin with two basic lemmas about the algebraic construction ζ .

LEMMA 14.7. Let a, b, c be three distinct vertices such that $\zeta(ab) = \zeta(ac)$, then $b_1 \neq c_1$.

PROOF. Let $\zeta(ab) = \zeta(ac)$. Then

$$a_1b_1 - a_2 - b_2 = a_1c_1 - a_2 - c_2$$

 $a_1(b_1 - c_1) = b_2 - c_2.$

If $b_1 = c_1$, then $b_2 = c_2$ as well. Hence, b = c, a contradiction.

LEMMA 14.8. Let a, b, c, d be four distinct vertices such that $\zeta(ab) = \zeta(ac)$ and $\zeta(db) = \zeta(dc)$, then $a_1 = d_1$.

PROOF. Since $b_1 \neq c_1$ by Lemma 14.7, then we know that

$$a_1 = \frac{b_2 - c_2}{b_1 - c_1} = d_1.$$

As shown in [35], the algebraic construction ζ_1 avoids monochromatic C_4 s (see Figure 51a) as well as the configuration shown in Figure 51b. We provide these two results for completeness.

LEMMA 14.9. Let a, b, c, d be distinct vertices. The algebraic coloring ζ forbids $\zeta(ab) = \zeta(bc) = \zeta(cd) = \zeta(da)$ (see Figure 51a).

PROOF. By Lemma 14.7, $b_1 \neq d_1$. But $b_1 = d_1$ by Lemma 14.8, a contradiction.

LEMMA 14.10. Let a, b, c, d be distinct vertices. The algebraic coloring ζ forbids $\zeta(ab) = \zeta(ac) = \zeta(ad)$ and $\zeta(bc) = \zeta(bd)$ (see Figure 51b).

PROOF. By Lemma 14.8, $a_1 = b_1$. Therefore, $\delta(a_1, b_1) = 0$. So $c_1 = a_1 = d_1$. But $c_1 \neq d_1$ by Lemma 14.7.

Now we will take care of a few additional configurations. We will use the following technical lemma.

LEMMA 14.11. Let a, b, c, d be four distinct vertices such that $\zeta(ab) = \zeta(cd)$ and $\zeta(bc) = \zeta(ad)$. Then

$$(a_1 + c_1)(b_1 - d_1) = 2(b_2 - d_2).$$

PROOF. The two colors give us the following relations:

$$a_1b_1 - a_2 - b_2 = c_1d_1 - c_2 - d_2$$

 $a_1d_1 - a_2 - d_2 = c_1b_1 - b_2 - b_2.$

We subtract the second equation from the first to get the desired equation.

LEMMA 14.12. Let a, b, c, d, e be distinct vertices. Then C(ab) = C(cd), C(bc) = C(ad), C(ae) = C(ce), and C(be) = C(de) (see Figure 51c) is forbidden by the coloring $C = \varphi \times \zeta$.



FIGURE 51. Configurations eliminated by the modified algebraic coloring.

PROOF. By Lemma 14.11, we know that

$$(a_1 + c_1)(b_1 - d_1) = 2(b_2 - d_2),$$

and from $\zeta(be) = \zeta(de)$ we get that

$$e_1(b_1 - d_1) = b_2 - d_2.$$

Therefore,

$$(a_1 + c_1)(b_1 - d_1) = 2e_1(b_1 - d_1)$$

 $a_1 + c_1 = 2e_1$

since $b_1 \neq d_1$ by Lemma 14.7. So $f_{e_1}(a_1) \neq f_{e_1}(c_1)$. By Lemma 12.7, $\varphi(ae) = \varphi(ce)$ implies that either a, c < e or e < a, c. In either case, $\zeta_2(ae) \neq \zeta_2(ce)$, a contradiction.

LEMMA 14.13. Let a, b, c, d, e be distinct vertices. Then C(ab) = C(cd), C(bc) = C(ad) = C(be) = C(de), and C(ac) = C(ae) (see Figure 51d) is forbidden by the coloring $C = \varphi \times \zeta$.

PROOF. As in the previous proof we get that $a_1 + c_1 = 2e_1$. By Lemma 14.8 we know that $b_1 = a_1$. Therefore, it follows from the second part of ζ_1 that $c_1 = d_1$. Therefore, $b_1 + d_1 = 2e_1$ and so $f_{e_1}(b_1) \neq f_{e_1}(d_1)$. As before, this fact along with Lemma 12.7 forces $\zeta_2(be) \neq \zeta_2(de)$, a contradiction.

LEMMA 14.14. Let a, b, c, d, e be distinct vertices. Then $\zeta(ab) = \zeta(cd) = \zeta(de)$ and $\zeta(bc) = \zeta(ad) = \zeta(be)$ (see Figure 51e) is forbidden by the algebraic coloring ζ .

PROOF. By Lemma 14.11 we know that

$$(a_1 + c_1)(b_1 - d_1) = 2(b_2 - d_2),$$

and by Lemma 14.8 we know that $b_1 = d_1$. Hence, $b_2 - d_2 = 0$ and so b = d, a contradiction.

LEMMA 14.15. Let a, b, c, d, e be distinct vertices. Then $\zeta(bc) = \zeta(cd) = \zeta(de)$, $\zeta(eb) = \zeta(ba) = \zeta(ad)$, and $\zeta(ac) = \zeta(ae)$ (see Figure 51f) is forbidden by the algebraic coloring ζ .

PROOF. By Lemma 14.7 we get that $b_1 \neq d_1$. By Lemma 14.8 we get that $a_1 = d_1$. Therefore, since the color encodes equality in the first coordinate we see that $b_1 = d_1$, a contradiction.

LEMMA 14.16. Let a, b, c, d, e be distinct vertices. Then $\zeta(bc) = \zeta(cd) = \zeta(de)$, $\zeta(eb) = \zeta(ba) = \zeta(ad)$, and $\zeta(ec) = \zeta(ae)$ (see Figure 51g) is forbidden by the algebraic coloring ζ .

PROOF. By Lemma 14.8 we get that $a_1 = c_1$. By Lemma 14.7, we get that $a_1 \neq c_1$, a contradiction.

CHAPTER 15

Additional questions about (p,q)-colorings.

1. General (p, p)-coloring with $n^{1/(p-2)+o(1)}$ colors

In general, the method of combining a variation of the CFLS coloring with a general algebraic construction using vectors from a space of dimension p-2 has the potential to show that

$$n^{1/(p-2)} \le f(n, p, p) = n^{1/(p-2)+o(1)}$$

for $p \ge 6$. Once the difficulty of case analysis is circumvented, then properties of vector spaces could hopefully be used to eliminate *p*-cliques which span only p - 1 colors.

2. A better bound for f(n, 5, 7)

The (5,5) and (5,6)-colorings have left q = 7 as the only remaining value for which a polynomial gap (in the order) between the known upper and lower bounds exists when p = 5. In this case we know that there are positive constants c_1 and c_2 such that

$$c_1 n^{2/3} \le f(n, 5, 7) \le c_2 n^{3/4}.$$

Is it possible to lower the upper bound to $n^{2/3+o(1)}$ using methods similar to those in Chapters 13 and 14?

3. The hypergraph version

Let $f_k(n, p, q)$ denote the minimum number of colors needed to color the edges of the complete k-uniform hypergraph on n vertices in such a way so that every p vertices span at least q colors. To date, little work has been done on this hypergraph version of the problem. There appear to be only two papers published on the topic, one by Conlon, Fox, Lee, and Sudakov [16] and one by Mubayi [36].

The main problem in the area is to determine for fixed p the threshold values for q at which there are large jumps in the order of the f(n, p, q). For $p > k \ge 3$ and 0 < i < k, Conlon, Fox, Lee, and Sudakov [16] showed that there exists a constant c dependent on k, p, and i for which

$$f_k\left(n, p, \binom{p-i}{k-i} + 1\right) = \Omega\left(\log_{(i-1)}(n)^c\right)$$

where we define $log_0(x) = x$ and $log_i(x) = log(log_{i-1}(x))$. They conjecture that this value of q is such a jump in the order. Is it true that

$$f_k\left(n, p, \binom{p-i}{k-i}\right) = \left(\log_{(i-1)} n\right)^{o(1)}?$$

Perhaps algebraic constructions have a place in this area as well.

APPENDIX A

Algorithm for reducing cases

This chapter contains material from a paper to be published in Combinatorics, Probability & Computing. [14]

The following algorithm is not difficult to verify, so we present it here without proof. The specific implementation we rely on is a Python script that can be found (with comments) at http://homepages.math.uic.edu/~acamer4/EdgeColors56.py. This particular script forbids monochromatic odd cycles and the edge-colorings shown in Figures 47, 48, 49, 50, as well as Figures 51a and 51b. The output is seven edge-colored copies of K_5 that each contain one of the remaining configurations in Figure 51.

Suppose we want to find every edge-coloring, up to isomorphism, of K_n that uses at most m colors and does not contain a copy of any $F \in \mathcal{F}$, a list of edge-colored complete graphs on n or fewer vertices. The algorithm takes \mathcal{F} , n, and m as input and returns a list \mathcal{R} of edge-colorings of K_n satisfying these requirements.

For each k = 3, ..., n, the algorithm creates a list L_k of acceptable edge-colorings of K_k by adding a new vertex to each K_{k-1} listed in L_{k-1} (where L_2 is the list of exactly one K_2 with its single edge given color 1), and then coloring the k-1 new edges in all possible ways from the color set [m]. For each graph in L_{k-1} and each way to color the new edges, we test the resulting graph to see if it contains any of the forbidden edge-colorings. If it does, then we move on. If not, then we test it against the new list L_k to see if it is isomorphic to any of the colorings of K_k already on the list. If it is, then we move on. Otherwise, we add it to the list L_k . The algorithm terminates

when it has tested all colorings of K_n .

```
Algorithm 1: List all edge-colorings with no forbidden subcoloring
Data: number of vertices n; maximum number of colors m; list of forbidden
        colorings \mathcal{F}
initialize L_2 as list containing one K_2 with its edges colored 1;
for k = 3, ..., n do
    initialize empty list L_k;
    for H \in L_{k-1} do
       for each function f : [k-1] \to [m] do
           let G be K_k with edge-colors same as H on the first k-1 vertices
            and color f(i) on edge ki for i = 1, \ldots, k - 1;
           if G contains no element of \mathcal{F} and is isomorphic to no element of
            L_k then
            | add G to the list L_k
           end
        end
    end
end
return L_n
```

CITED LITERATURE

- Dana Angluin, Michael Frazier, and Leonard Pitt. Learning conjunctions of Horn clauses. *Machine Learning*, 9(2-3):147–164, 1992.
- Maria Axenovich. A generalized Ramsey problem. Discrete Mathematics, 222(1-3):247-249, 2000.
- [3] László Babai and Péter Frankl. Linear Algebra Methods in Combinatorics: With Applications to Geometry and Computer Science. Department of Computer Science, Univ. of Chicago, 1992.
- [4] Rahil Baber and John Talbot. Hypergraphs do jump. Combinatorics, Probability and Computing, 20(02):161–171, 2011.
- [5] WG Brown, Paul Erdős, and M Simonovits. Extremal problems for directed graphs. Journal of Combinatorial Theory, Series B, 15(1):77–93, 1973.
- [6] WG Brown and F Harary. Extremal digraphs. Coll. Math. Soc. J. Bolyai, 4:135– 198, 1969.
- [7] WG Brown and Miklós Simonovits. Digraph extremal problems, hypergraph extremal problems, and the densities of graph structures. *Discrete Mathematics*, 48(2):147–162, 1984.
- [8] Alex Cameron. Extremal numbers for $2 \rightarrow 1$ directed hypergraphs with two edges part I: The nondegenerate cases. arXiv preprint arXiv:1607.04930, 2016.
- [9] Alex Cameron. Extremal numbers for $2 \rightarrow 1$ directed hypergraphs with two edges part II: The degenerate cases. arXiv preprint arXiv:1607.04932, 2016.
- [10] Alex Cameron. Extremal problems on generalized directed hypergraphs. arXiv preprint arXiv:1607.04927, 2016.
- [11] Alex Cameron. An explicit edge-coloring of K_n with six colors on every K_5 . arXiv preprint arXiv:1704.01156, 2017.
- [12] Alex Cameron. Extremal numbers for directed hypergraphs with two edges. *The Electronic Journal of Combinatorics*, 25(1):P1.56, 2018.
- [13] Alex Cameron and Emily Heath. A (5,5)-coloring of K_n with few colors. arXiv preprint arXiv:1702.06227, 2017.

- [14] Alex Cameron and Emily Heath. A (5,5)-colouring of K_n with few colors. Combinatorics, Probability & Computing (to appear), 2018.
- [15] David Conlon, Jacob Fox, Choongbum Lee, and Benny Sudakov. The Erdős– Gyárfás problem on generalized Ramsey numbers. Proceedings of the London Mathematical Society, 110(1):1–18, 2015.
- [16] David Conlon, Jacob Fox, Choongbum Lee, and Benny Sudakov. On the grid Ramsey problem and related questions. *International Mathematics Research Notices*, 2015(17):8052–8084, 2015.
- [17] Dennis Eichhorn and Dhruv Mubayi. Edge-coloring cliques with many colors on subcliques. *Combinatorica*, 20(3):441–444, 2000.
- [18] Paul Erdős. On extremal problems of graphs and generalized graphs. Israel Journal of Mathematics, 2(3):183–190, 1964.
- [19] Paul Erdős. On some extremal problems on r-graphs. Discrete Mathematics, 1(1):1-6, 1971.
- [20] Paul Erdős. Problems and results on finite and infinite graphs. Recent Advances in Graph Theory (Proc. Second Czechoslovak Sympos., Prague, 1974), pages 183– 192, 1975.
- [21] Paul Erdős, C. Ko, and R. Rado. Intersection theorems for systems of finite sets. Quart. J. Math. Oxford, 12:313–320, 1961.
- [22] Paul Erdős and M Simoinovits. A limit theorem in graph theory. In Studia Sci. Math. Hung. Citeseer, 1966.
- [23] Paul Erdős, Arthur H Stone, et al. On the structure of linear graphs. Bull. Amer. Math. Soc, 52(1087-1091):3, 1946.
- [24] P Erdős. Solved and unsolved problems in combinatorics and combinatorial number theory. *European Journal of Combinatorics*, 2:1–11, 1981.
- [25] Paul Erdős and András Gyárfás. A variant of the classical Ramsey problem. Combinatorica, 17(4):459–467, 1997.
- [26] Peter Frankl, Yuejian Peng, Vojtech Rödl, and John Talbot. A note on the jumping constant conjecture of Erdős. *Journal of Combinatorial Theory, Series* B, 97(2):204–216, 2007.
- [27] Peter Frankl and Vojtěch Rödl. Hypergraphs do not jump. Combinatorica, 4(2-3):149–159, 1984.
- [28] Giorgio Gallo, Giustino Longo, Stefano Pallottino, and Sang Nguyen. Directed hypergraphs and applications. *Discrete applied mathematics*, 42(2):177–201, 1993.

- [29] Peter Keevash. Hypergraph Turán problems. Surveys in combinatorics, 392:83– 140, 2011.
- [30] Elliot Krop and Irina Krop. Almost-rainbow edge-colorings of some small subgraphs. Discussiones Mathematicae Graph Theory, 33(4):771–784, 2013.
- [31] Marina Langlois. *Knowledge representation and related problems*. PhD thesis, University of Illinois at Chicago, 2010.
- [32] Marina Langlois, Dhruv Mubayi, Robert H Sloan, and György Turán. Combinatorial problems for Horn clauses. In *Graph Theory, Computational Intelligence* and *Thought*, pages 54–65. Springer, 2009.
- [33] Imre Leader and Ta Sheng Tan. Directed simplices in higher order tournaments. Mathematika, 56(01):173–181, 2010.
- [34] Dhruv Mubayi. Edge-coloring cliques with three colors on all 4-cliques. Combinatorica, 18(2):293–296, 1998.
- [35] Dhruv Mubayi. An explicit construction for a Ramsey problem. *Combinatorica*, 24(2):313–324, 2004.
- [36] Dhruv Mubayi. Coloring triple systems with local conditions. Journal of Graph Theory, 81(3):307–311, 2016.
- [37] Alberto Perelli, János Pintz, and Saverio Salerno. Bombieri's theorem in short intervals. Annali della Scuola Normale Superiore di Pisa-Classe di Scienze, 11(4):529–539, 1984.
- [38] Oleg Pikhurko. On possible turán densities. Israel Journal of Mathematics, 201(1):415–454, 2014.
- [39] Dijen K Ray-Chaudhuri, Richard M Wilson, et al. On t-designs. Osaka J. Math, 12(3):737–744, 1975.
- [40] Alexander A Razborov. Flag algebras. The Journal of Symbolic Logic, 72(04):1239–1282, 2007.
- [41] SJ Russell and P Norvig. Artificial Intelligence: A Modern Approach. Prentice Hall, 2002.
- [42] Paul Turán. On an extremal problem in graph theory. Mat. Fiz. Lapok, 48(137):436-452, 1941.
- [43] Paul Turán. On the theory of graphs. Colloq. Math., 3:19–30, 1954.
- [44] Paul Turán. Research problems. Magyar Tud. Akad. Kutató Int. Közl., 6:417– 423, 1961.

VITA

NAME:	Alex Cameron
EDUCATION:	B.A., Mathematics and Physics, Hendrix College, Conway, Arkansas, 2006
	J.D., Law, Duke University, Durham, North Carolina, 2011
	M.S., Mathematics, University of Illinois, Chicago, Illinois, 2017
	Ph.D., Mathematics, University of Illinois, Chicago, Illinois, 2018
TEACHING:	University of Illinois at Chicago, 2011-2018
PUBLICATIONS:	Alex Cameron. Extremal Numbers for Directed Hyper- graphs with Two Edges. The Electronic Journal of Com- binatorics 25(1), P1.56 (2018).
	Alex Cameron and Emily Heath. A $(5,5)$ -coloring of K_n with few colors. To appear in Combinatorics, Probability and Computing, Preprint posted February 2017.
	Alex Cameron. Counting Ring Homomorphisms. In PME Journal, Vol. 13, pages 65-67, 2010.
POSTERS & TALKS	AMS Special Session on Research from GRWC Joint Mathematics Meetings, San Diego, CA Talk: A $(5,5)$ -coloring of K_n with few colors, 2018.
	29th Cumberland Conference on Combinatorics, Graph Theory and Computing Vanderbilt University, Nashville, TN Talk: A $(5,5)$ -coloring of K_n with few colors, 2017.
	Combinatorics Seminar Rényi Institute, Budapest, Hungary Talk: Extremal Problems on Directed Hypergraphs, 2016.

VITA

Combinatorics Seminar University of Szeged, Szeged, Hungary Talk: Extremal Problems on Directed Hypergraphs, 2016.

EXCILL III: Extremal Combinatorics at Illinois IIT, Chicago, IL Poster: Extremal Problems on Generalized Directed Hypergraphs, 2016.

Graduate Research Workshop in Combinatorics University of Wyoming, Laramie, WY Talk: (p, q)-colorings, 2016.

Graduate Students Combinatorics Conference Clemson University, Clemson, SC Talk: Extremal Problems on Directed Hypergraphs: Forbidden subgraphs with two edges, 2016.

2006 Oklahoma-Arkansas MAA meeting University of Arkansas, Fayetteville, AR Talk: Counting Ring Homomorphisms, 2006.

Copyrights to Published Papers

Chapters 2-8 as well as Chapter 10 reproduce, with some minor changes, a paper published by the Electronic Journal of Combinatorics 12 which gives the following copyright notice at http://www.combinatorics.org/ojs/index.php/eljc/about/submissions#copyrightNotice

"Copyright Notice The copyright of published papers remains with the authors. We only require your agreement that we publish it, as described in the following publication release agreement:

1. This is an agreement between the Electronic Journal of Combinatorics (the "Journal"), and the copyright owner (the "Owner") of a work (the "Work") to be published in the Journal. 2. The Owner warrants that s/he has the full power and authority to enter into this Agreement and to grant the rights granted in this Agreement. 3. The Owner hereby grants to the Journal a worldwide, irrevocable, royalty free license to publish or distribute the Work, to enter into arrangements with others to publish or distribute the Work, and to archive the Work. 4. The Owner agrees to ensure, to the extent reasonably possible, that further publication of the Work with the same or substantially the same content will include an acknowledgement of prior publication in the Journal.

Published papers will contain a copyright notice indicating that copyright is held by the authors. In addition, we strongly encourage authors to release their paper under one of the Creative Commons licenses.

Most papers published before March 31, 2018 did not contain explicit copyright or license statements. In those cases, authors agreed to items 1-3 above and to the following version of item 4: "The Owner agrees that further publication of the Work, with the same or substantially the same content as appears in the Journal, will include an acknowledgement of prior publication in the Journal.""

Chapters 12 and 13 reproduce, with some minor changes, joint work with Emily Heath that has been reviewed and accepted for publication in Combinatorics, Probability & Computing 14. A copy of the signed transfer of copyright follows. It includes a detailed list of what is and is not allowed to be republished. The version of the paper included in this thesis is an "author original" so it is my understanding that its submission to the UIC library is allowed.

Journal - Copyright Transfer Form

Please complete both Sections A and B, sign, and return this page by post to Journals Production, Cambridge University Press, University Printing House, Shaftesbury Road, Cambridge, CB2 8BS, UK, or by email to journalscopyright@cambridge.org as a scanned, signed (but not electronically signed) document, as soon as possible. By completing, signing and returning this form you hereby agree to the Terms and Conditions enclosed (Doc.JCT.T&C.STM.14.1).

Combinatorics, Probability & Computing	CPC
In consideration of the publication in Combinatorics, Probability & Computing	11 - 11
of the contribution entitled: A (575) - COLOUTING OF KIN WI	th tew colours
by (all authors' names): Atex (anetron Faulty then the	
Section A - Assignment of Convright (fill in either part 1 or 2 or 3)	

To be filled in if copyright belongs to you

Transfer of copyright

I/we hereby assign to Cambridge University Press, full copyright in all forms and media in the said contribution, including in any supplementary materials that I/we may author in support of the online version.

I/we hereby assert my/our moral rights in accordance with the UK Copyright Designs and Patents Act (1988).

 the sole author(s) Signed (tick one)

Name (block letters) ALEX CAMERON Institution Company Department of Mathematics, Statistics, and Computer Science, OIC Address \$515, forgas St, Chicago, IC 60607 USA Signature: Date: 4/10/18 (Additional authors' names, affiliations and nationalites should be provided on the second sheet of this form and all should be aware of, and accept, the terms of this form and accompanying form Doc.JCT.T&C.STM.14.1.)

To be filled in if copyright does not belong to you 2 Name and address of copyright holder ... a

The copyright holder hereby grants to Cambridge University Press, the exclusive right to publish the contribution in the Journal including any supplementary materials that support the online version and to deal with requests from third parties.

(Signature of copyright holder or authorised agent)

US Government exemption

I've certify that the paper above was written in the course of employment by the United States Government so that no copyright exists. Signature: ... Name (Block letters): sector and a sector sec

Section B - Warranty and disclosure of conflict of interest

I/we warrant that I am/we are the sole owner or co-owners of the contribution and have full power to make this agreement, and that the contribution has not been previously published, contains nothing that is in any way an infringement of any existing copyright or licence, or duty of confidentiality, or duty to respect privacy, or any other right of any person or party whatsoever and contains nothing libellous or unlawful; and that all statements purporting to be facts are true and that any recipe, formula, instruction or equivalent published in the Journal will not, if followed accurately, cause any injury or damage to the user.

I/we further warrant that permission for all appropriate uses has been obtained from the copyright holder for any material not in my/our copyright including any audio and video material, that the appropriate acknowledgement has been made to the original source, and that in the case of audio or video material appropriate releases have been obtained from persons whose voices or likenesses are represented therein. I/we attach copies of all permission and release correspondence.

I indemnify and keep Cambridge University Press, indemnified against any loss, injury or damage (including any legal costs and disbursements paid by them to compromise or settle any claim) occasioned to them in consequence of any breach of these warranties,

Name (block letters) ALEX CAMERON
Signature Date 4/10/18
(one author authorised to execute this warranty statement above and conflict of interest statement below on behalf of all the authors of the above
article)

Please disclose any potential conflict of interest pertaining to your contribution or the Journal; or write 'NONE' to indicate you declare no such conflict of interest exists. A conflict of interest might exist if you have a competing interest (real or apparent) that could be considered or viewed as exerting an undue influence on you or your contribution. Examples could include financial, institutional or collaborative relationships, The Journal's editor(s) shall contact you if any disclosed conflict of interest may affect publication of your contribution in the Journal.

Potential conflict of interest NONE

The information provided on this form will be held in perpetuity for record purposes. The name(x) and address(es) of the author(s) of the contribution may be reproduced in the journal and provided to print and online indexing and abstracting services and bibliographic databases.



Please provide additional authors' names, addresses, affiliations, and nationalities on this page. All authors should be aware of, and accept, the terms of this form and accompanying form 'Doc.JCT.T&C.STM.14.1'

EMILY HEATH Department of Mathematics University of Illinois at Urbana-Champaign Urbana, IL 61801 USA


(Doc.JCT.T&C.STM.14.1) Terms and Conditions for authors to Scientific, Technological and Medical journals published by Cambridge University Press.

By completing, signing and returning the Copyright Transfer form (Form JCT.14.1) you have agreed to abide by the following Terms and Conditions. Please retain this document for future reference.

Definitions²¹

Author's Original (AO)	Any version of the article that is considered by you to be of sufficient quality to be submitted for formal peer review by a second party. Content and layout as set out by you.		
Submitted Manuscript Under Review (SMUR)	Any version of the article that is under formal review for inclusion in the journal. Content and layout follow the journal's stated submission requirements.		
Accepted Manuscript (AM)	The version of the article that has been accepted for publication in the journal. This version may have been revised following peer review but may be subject to further editorial input by Cambridge University Press.		
Version of Record (VoR)	The fixed version of the article that has been made available. This includes an "early release" article as long as it is citable via some permanent identifier(s). This does not include any "early release" article that has not yet been "fixed" by processes that are still to be applied, such as copy-editing or proof corrections. The VoR includes any corrected or enhanced VoR.		
Personal Website	A non-commercial website maintained solely or in part by you.		
Departmental / Institutional Repository	The online archive of intellectual output of the institution at which you were based when the article was written as well the repository of any future institution at which you are based.		
Non-commercial Subject Repository	A repository relating to the subject area of your article which does not allow the content to be used commercially.		
Commercial Repository	A repository which permits the content to be used for commercial gain.		
Social Media Sites	Websites and computer programs that allow people to communicate and share information on the internet using a computer or mobile phone ²²		

Re-use of your article.

All postings shall include a prominent bibliographical reference and statement of copyright ownership. Where possible all postings should include a link to the published article on Cambridge Journals Online. In relation to the posting of the AO or SMUR, a statement that such version has been accepted for publication and will appear in a revised form subsequent to peer

²¹ Adapted from NISO RP-8-2008, Copyright © 2008 by the National Information Standards Organization

¹⁸ http://dictionary.cambridge.org/dictionary/british/social-

media?q=social+media&utm_medium=widget_searchbox&utm_source=widget_searchbox_source&utm_campaign=widget_btacking Form [CT 14.1 and Dec.JCT.T&C.STM.14.1

review and / or editorial input by Cambridge University Press and / or the journal's proprietor , as well as a link to the journal's site on Cambridge Journals Online, should be included.

Cambridge does not permit full articles in AM or VOR form to be posted on Commercial Repositories or Social Media Sites including, but not limited to, Social Science Research Network ('SSRN'), ResearchGate, Academia.edu, Mendeley or LinkedIn.

Notwithstanding the assignment of copyright or grant of licence in your article, you retain the following non-transferable rights to deposit versions of your article, (subject to appropriate permission having been cleared for any third-party material):

	Personal Website	Departmental / Institutional Repository	Non-commercial Subject Repository	Commercial Repository and Social Media Sites
AO	At any time	At any time	At any time	At any time
SMUR	At any time	At any time	At any time	At any time
AM	On acceptance of publication.	Six months after first publication.	Six months after first publication	Abstract only in PDF or HTML format no sooner than first publication of the full article.
VOR	Abstract only in PDF or HTML format no sooner than first publication of the full article.	Abstract only in PDF or HTML format no sooner than first publication of the full article.	Abstract only in PDF or HTML format no sooner than first publication of the full article.	Abstract only in PDF or HTML format no sooner than first publication of the full article.

Other

You may make hard copies of the article or an adapted version for your own purposes, including the right to make multiple copies for course use by your students, provided no sale is involved.

You may reproduce the article or an adapted version of it in any volume of which you are editor or author subject to normal acknowledgement.

If your reuse is not covered by the above please follow the 'Rights and Permissions' quick link on the Cambridge website for your region via www.cambridge.org.

Cambridge University Press co-operates in various licensing schemes that allow material to be photocopied within agreed restraints (e.g. the CCC in the USA and the CLA in the UK). Any proceeds received from such licences, together with any proceeds from sales of subsidiary rights in the Journal, directly support its continuing publication.

Cambridge University Press acts in accordance with the UK Bribery Act 2010 and the Data Protection Act 1998. Please refer to the Press's relevant policies, (http://www.cambridge.org/policy/privacy/),

(http://www.cambridge.org/policy/abc_policy/) which may be revised from time to time.